Working Paper Series 768 (ISSN 2788-0443)

Hostility, Population Sorting, and Backwardness: Quasi-Experimental Evidence from the Red Army after WWII

Christian Ochsner

CERGE-EI Prague, November 2023

ISBN 978-80-7343-575-2 (Univerzita Karlova, Centrum pro ekonomický výzkum a doktorské studium) ISBN 978-80-7344-701-4 (Národohospodářský ústav AV ČR, v. v. i.)

Hostility, population sorting, and backwardness: Quasi-experimental evidence from the Red Army after WWII*

Christian Ochsner[†]

October 30, 2023

Abstract

Does a short episode of conflict or exposure to hostile troops cause regional economic backwardness, and if so, why and how does it persist? I answer these questions by exploiting economic differences across the idiosyncratic and short-lived line of contact between the Red Army and the Western Allies in South Austria at the end of WWII. Spatial regression discontinuity estimates show that hostile presence of the Red Army for 74 days caused an immediate relative population decline of around 12%, amplified to 25% by today. Age-specific migration patterns and subsequent fertility differences explain the multiplying effects. Sector development and measures of local labor productivity in 2011 also lag behind in regions briefly seized by the Red Army, likely driven by skill-specific migration and hampered investment patterns after WWII. The findings provide novel insights into the long-run effects of wars and conflicts, and point to the isolated role of the Red Army's hostile actions after WWII to understand the European economic East-West divide.

JEL-Classification: D74, J13, N44, O14

Keywords: Conflict, Hostility, Population shock, Regional development, Red Army

^{*}I thank Dan Bogart, Stephen Broadberry, Lorenzo Casaburi, Nicholas Crafts, Nadja Dwenger, Sergei Guriev, Walker Hanlon, Roland Hodler, Erik Hornung, Alexander Klein, Stephan Maurer, Luigi Pascali, Daniele Paserman, Felix Rösel, Marcel Thum, Hans-Joachim Voth and Nikolaus Wolf for helpful comments. I also thank participants at seminars, conferences and workshops at the Barcelona GSE Summer Forum, CAGE-EHES-IAS Summer School, CERGE–EI Prague, Charles University (UNCE Workshop), DGO Berlin, EHES Tübingen, Humboldt-University Berlin, NOeG Vienna, RES Meeting London, ifo Dresden Workshop on Regional Economics, SITE—Stockholm Institute of Transition Economics, SMYE Halle, SSES Lausanne, Swiss Workshop on Political Economy and Development in Weggis, TU Dresden, University of Hohenheim, University of Warwick and VfS Freiburg/Br. A special thank you goes to Ernst Reinhold Lasnik and Michael Pammer for comments on Austrian (Styrian) history. I acknowledge financial support from the UNCE project (UNCE/HUM/035) and the Czech Science Foundation (GACR 21–26353S). A preliminary version of this paper circulated as "Dismantled once, diverged forever? A quasi-natural experiment of Red Army's misdeeds in post-WWII Europe" (Ochsner, 2017). All remaining errors are mine.

[†]CERGE-EI-A joint workplace of Charles University and the Economics Institute of the Czech Academy of Sciences, Politickych veznu 7, 111 21 Prague, Czech Republic, Phone: +420 224 005 173, e-mail: christian.ochsner@cerge-ei.cz.

1 Introduction

In recent years, the number of globally reported conflicts has reached an all-time peak since World War II (WWII). At the same time, the UNHCR estimates the number of internally and externally displaced people to exceed 100 million in 2022—figures that exceed even the period of mass-expulsions and displacement after WWII.¹ Inevitably, such figures impact the long-run economic trajectory of affected regions. A growing literature analyzes the economic consequences of wars, conflict and hostile invasions of regular armies or armed groups (for reviews see Blattman and Miguel (2010), Rohner and Thoenig (2021) or Munroe et al. (2023)). On the one hand, there is somewhat of a consensus in the literature that the tangible and measurable costs of conflicts—destruction of physical capital—has limited long-run effects on economic figures (Davis and Weinstein, 2002; Brakman et al., 2004; Miguel and Roland, 2011; Waldinger, 2016; Yamada and Yamada, 2021). On the other hand, conflicts also lead to migration, affect intangible assets such as health, behavior and educational attainment, and disrupt social capital and institutions.² Studies on these matters often focus on short-run outcomes on individuals, while their long-run impacts on the economic trajectory of regions are hard to disentangle from each other since they are intervoven and affect the social fabric of societies (Rohner and Thoenig, 2021).

In this paper, I focus on a violent period with high stakes: The Red Army's hostility in Europe at the end of WWII.³ I study the isolated effects of the fear of and the hostility by the Red Army during the liberation of Europe from Nazi Germany on subsequent regional economic development. In the course of the liberation of Europe from Nazi Germany in Spring 1945, the Red Army caused mass exoduses in regions all over Central and Eastern Europe that they liberated, and also dismantled plants and infrastructure. I aim to isolate the presence of the Red Army at the end of the war from other East-West differences across the Iron Curtain such as war damage, Soviet domination for several decades, different institutional and economic legacies or changed behavior and preferences of people (Alesina and Fuchs-Schündeln, 2007). I focus on the Austrian state of Styria in South Austria, where some regions that were initially seized by the Red Army in May 1945 were handed over to the Western Allies after 2.5 months. In fact, these regions in Styria comprised the largest coherent area in Europe that was initially seized by the Red Army, but was not under long-term Soviet control. Post-WWII population growth already points to the long-run effects of the short-term presence of the Red Army. Figure 1 shows

¹Data on conflicts are from the UCDP Armed Conflict Database (https://ucdp.uu.se/) and on displacements from UNHCR (https://www.unhcr.org/global-trends); retrieved in September 2023.

²The effects on social capital (Bellows and Miguel, 2009; Rohner *et al.*, 2013a,b; Besley and Reynal-Querol, 2014; Bauer *et al.*, 2016; Dell and Querubin, 2017) and education (Bedard and Deschênes, 2006; Akresh *et al.*, 2012) are ambiguous, while health outcomes are negative (León, 2012; Aghion *et al.*, 2018).

³The term "Red Army" is used throughout the paper to indicate troops that were under the command of the Soviet Union, including Bulgarian and Yugoslavian troops that arrived in southern Austria. Most Soviet troops engaged in the seizing of Austria belonged to the 3^{rd} Ukrainian Front (Stelzl-Marx, 2012).



Figure 1: Population growth along the Iron Curtain (Germany) and in occupied Austria

Notes: The figure shows the long-run population growth from 1939 (last data before division) to 2011 in percentage points in divided Germany and liberated and occupied Austria. German figures include divided Berlin in the respective parts. The Western-Allies occupied regions in Austria are Upper Austria (South of the River Danube), Salzburg, Styria, Carinthia, Tyrol and Vorarlberg. The Soviet occupation zone in Austria consists of Lower Austria, Burgenland and nothern Upper Austria (Mühlviertel). The divided cities of Linz and Vienna are excluded in all figures. Styria was initially liberated by the Red Army, US and UK troops and was then fully allocated to the Western (UK) occupation zone. Source: See text and Online Appendix B.

the post-WWII population growth in Austrian regions according to the occupation zones instituted from July 1945 to October 1955 and compares them to the figures in divided Germany across the Iron Curtain. In both countries, regions liberated and occupied by the Western Allies show substantial population growth in the last eighty years. In contrast, the population in East Germany, the former German Democratic Republic—largely seized by the Red Army after WWII and under Soviet control until 1989—had a lower population in 2011 than in 1939. In Austria, population figures for regions occupied by the Soviets for more than ten years also lag behind those of regions seized and occupied by the Western troops. Styria, the region of contact between the Red Army and the Western Allies, has also performed poorly, despite the fact that only parts of Styria were dominated for 74 days by the Red Army. These figures already provide a first impression that the Red Army's legacy right after WWII seems to have long-lasing impacts on regional economic trajectories, and this without long-run Soviet domination or changed institutions.

I investigate the isolated and causal long-term effects of the Red Army's hostility at the end of WWII on subsequent population dynamics, sector development and labor productivity up to the present. I focus on the arbitrarily-drawn line of contact within Styria between the Red Army and the Western Allies (US and UK troops). On the day of the Nazi-German surrender on May 8, 1945, the Red Army and the Western Allies overran these last parts of Europe under Nazi-German control from different directions, and in less than one day. The places where the respective Allies met became the line of contact. Whether a certain municipality was the last unlucky one to be seized by the Red Army, or the last fortunate one to be captured by Western troops was—broadly speaking—fully exogenous.⁴ In Styria, this line of contact lasted for only 74 days. In July 1945, the Red Army had entirely withdrawn from Styria towards its officially assigned occupation zone in East Austria. This withdrawal of the Red Army was unique in all of Europe: it only happened at a large scale in Styria due to zoning negotiations (Erickson, 1950; Slapnicka, 1986). Styria as a whole was thereafter assigned to the UK occupation zone. Austria remain occupied by the Allies for more than 10 years until the fall of 1955.

The presence of the Red Army for 74 days in some parts of Styria led to municipalities being exposed to hostile troops. The Austrian civic population feared the arrival of the Red Army mainly due to Nazi propaganda, poor reputation and reports of crimes committed by Red-Army soldiers against civilians in Eastern Europe (Mascher-Pichler, 2009; Stelzl-Marx, 2012). During the last weeks of WWII until the final zoning of Austria in July 1945, a mass exodus from eastern to western Austria took place by people fleeing from the Red Army. Living conditions in places that were seized by the Red Army were hostile: Beer (2004) reports 30,000 estimated rapes in the parts of Styria initially seized by the Red Army. There was also official dismantling of industrial plants and infrastructure, and thousands of informal lootings by the Red Army and its soldiers (Bischof, 1999; Eberhart, 1995; Pickl, 1995). Thus, regions initially seized by the Red Army were less favorable places to live and to initiate economic activity following the war.

I use this spatial discontinuity in hostility along the 74-day-long line of contact between the Red Army and the Western Allies in spatial regression discontinuity (RD) setups. I combine a battery of self-compiled archival data with data from official statistics. I first show that the line of contact divided formerly identical regions in terms of geographical characteristics, and pre-treatment population dynamics and sector shares. I find that regions seized by the Red Army—but without any long-term Soviet dominance or institutional differences in place—suffered in the long-run, even until the present day. Spatial RD estimates show that the liberation caused an immediate relative population decline of around 12%, amplified to 25% by today. I find that younger cohorts, and young families with children in particular, were more likely to escape the Red Army and settled in places liberated by the Western Allies. Age-specific migration patterns and the resulting fertility- and death-rates differentials across the line of contact cause the multiplying effects over the decades. Sector development and measures of local labor productivity in 2011 also lagged behind in regions initially seized by the Red Army. The lag in economic and sector development is likely driven by a skill-specific migration pattern after WWII: Regions liberated by the

⁴The seizing of Germany and Czechia was less exogenous than of Styria. In Germany, US troops strategically waited at the River Elbe until the Red Army had seized the rest of Germany. In Czechia, US troops waited in West Bohemia to let the Red Army proceed. Studies using the line of contact in Germany are Martinez *et al.* (2023) and Eder *et al.* (2023), and Grossmann *et al.* (forthcoming) in Czechia.

Western Allies received a boost in the share of semi-skilled laborers (industrial workers, craftsmen), whereas adjacent municipalities seized by the Red Army immediately became more agricultural, and remained trapped in agriculture until 2011. I also find strong evidence for hampered investments in places seized by the Red Army by analyzing the investment patterns into agricultural machinery after WWII. In contrast, I do not find any long-term effects of local aerial bombing or officially reported dismantling activities by the Red Army. This indicates that people seem to matter more than the loss of tangible assets for long-run economic development of regions. The results are robust to various types of RD specifications, including estimates with single- and multidimensional forcing variables and pairwise regressions with line of contact municipalities, robust bias-corrected RD estimates with a data-driven bandwidth selection, different regional sub-samples, pseudo treatments and difference-in-differences specifications. I also show that the effects are not just locally pronounced by applying the synthetic control method to compare the Styrian capital city of Graz with other cities in eastern and western Austria. The results confirm the economic backwardness of places initially seized by the Red Army. Multiple hypothesis testing shows that population figures and differences in measures of labor productivity are the main long-run outcomes of Red-Army hostility at the end of WWII.

I contribute to several strands of the literature. First, my findings provide novel understanding of the long-run economic trajectory and amplifying effects due to selective migration in the course of conflicts or hostility. Surprisingly, the literature is very scarce here. Only two empirical studies with somewhat extreme settings have explored the long-run effects on sending regions directly. Testa (2020) studies the mass expulsion of ethnic Germans from Czechia after WWII and Chaney and Hornbeck (2016) look at the expulsion of Moriscos from Spain in the 17th century. They find that expulsions caused regional economic backwardness in the long run. Other studies focus on a specific group of people, mainly ethnic or religious minorities, to test how prosecution and displacement affects regional outcomes (Acemoglu et al., 2011; Akbulut-Yuksel and Yuksel, 2015; Pascali, 2016; Arbath and Gokmen, 2022). In the same vein, Waldinger (2010, 2016) and Huber et al. (2021) investigate the loss of business and scientific elites on subsequent decline in performance.⁵ My study contributes to these findings by showing how selective settlement patterns with respect to age and occupational characteristics cause initial differences that amplify over time. These new insights may also help to assess the long-run regional economic costs of the Russian-Ukrainian war today, with millions of displaced people, mainly women and children.

⁵In contrast, a large body of literature analyze how migrants affect the receiving regions. Examples are Hornung (2014) on Huguenot immigrants in Prussia, Semrad (2015), Braun and Kvasnicka (2014), Braun and Omar Mahmoud (2014), Schumann (2014) and Peters (2022) on German expellees after WWII on regional economic activity, industrial growth and schooling outcomes in Germany or the boost of scientific output in US regions that received German Jewish émigrés escaping the Holocaust (Moser *et al.*, 2014). See also Becker and Ferrara (2019), Becker (2022) or Munroe *et al.* (2023) for recent reviews.

Second, my study points to the isolated effects of the hostility by the Red Army after WWII in their seized parts of Europe to understand the ongoing European economic east-west divide along the former Iron Curtain. Despite the conditional convergence of planning economies during the first postwar decades (Ofer, 1987; Allen, 1998; Harrison, 2003), studies document a rather inefficient use of labor and capital in the Soviet economy (Cheremukhin et al., 2013, 2016) or use the inefficiencies in the planning process to explain rent seeking and corruption (Harrison and Kim, 2006; Harrison, 2011). Another strand of literature explains the backwardness in the Soviet hemisphere with reference to war destruction and low investments (Vonyó, 2017, 2020) and reparations mainly paid by satellite states of the Soviet Union.⁶ Regarding differences within re-unified Germany, Fuchs-Schündeln and Masella (2016) and Lichter et al. (2020) show the long-run negative effects of socialist education and government surveillance, respectively. Generally, social norms and trust in East and West Germany are still different even decades after the fall of the Iron Curtain and might thus affect the social fabric of society (Ockenfels and Weimann, 1999; Alesina and Fuchs-Schündeln, 2007; Brosig-Koch et al., 2011; Bondar and Fuchs-Schündeln, 2023). In contrast to these explanations, I show that regions that were initially seized by the Red Army would lag behind Western Europe in terms of population dynamics, sector development or labor productivity even in the absence of long-term Soviet domination or a planning economy. I explain these effects with ageand occupation-specific migration caused by the advance of the Red Army and hampered investment patterns in the direct aftermath of WWII.

Third, my study adds to the literature on regional economic inequality and regional convergence. This literature is roughly divided into two competing explanations. On the one hand, scholars refer to (time-invariant) natural endowments such as resources or geography to explain regional economic differences (Ellison and Glaeser, 1999; Davis and Weinstein, 2002, 2008; Brakman *et al.*, 2004; Redding and Sturm, 2008; Ellison *et al.*, 2010; Nagy, 2022). On the other hand, spatial differences in economic activity can result endogenously due to the local interaction of economic agents and scale economies. Seminal theoretical contributions regarding the so-called New Economic Geography literature are made by Henderson (1974), Krugman (1991), and more recently by Davis and Dingel (2019).⁷ These models predict that short-term shocks can persist and even amplify mainly due to the sorting of people and increasing returns to scale. This also links to the big push theory established by Rosenstein-Rodan (1943) and formalized by Murphy *et al.*

⁶Especially former Allies of Nazi-Germany (Hungary, Romania, Slovakia) and East Germany, but also Poland, Bulgaria and Czechoslovakia faced a decline in capital stock due to the Red-Army dismantling after WWII (see, for example, Stolper (1960); Köhler (1965); Liberman (1996); Taylor (2008) and Bekes *et al.* (2015)). Liberman (1996) also shows that East-German "reparations payments" (incl. dismantling by the Soviets) in the aftermath of WWII reached around 20% of pre-WWII GNP in East Germany.

⁷Random growth models (Simon, 1955; Gabaix, 1999) are a special group of theoretical models that would also predict persistence, but without amplifying effects.

(1989) to explain how an initial reallocation of resources leads to increasing growth rates thereafter (see Kline and Moretti (2014) or Peters (2022) for empirical studies). My findings corroborate these predictions and show a divergence of regions over time rather than a (re-)convergence after the initial shock.⁸

Lastly, my paper also adds to the fast-growing literature on the persistence of historical events on present-day economic outcomes. Examples of historic dependencies of economic variation across space are documented for militarily insecure frontiers (Oto-Peralías and Romero-Ávila, 2016), ancient Roman road infrastructure (Wahl, 2017; Dalgaard *et al.*, 2022), loss of credit collateral due to wartime destruction (Feigenbaum *et al.*, 2022), place-based policies (Kline and Moretti, 2014; Ehrlich and Seidel, 2018), long-obsolete portage cities (Bleakley and Lin, 2012), high sunk costs (Redding *et al.*, 2011) or long-gone exploitative institutions (Dell, 2010). One study on path dependence of a population shock in post-WII Austria is closely related to mine. Eder and Halla (2016) look at population shocks across the 1945 to 1955 occupation zone border in Austria and document a similar population shift as in my setup. However, my paper corroborates this paper based on a very short treatment period and adds a detailed channel discussion of the amplifying population figures and ongoing economic differences.

The remainder of this paper is as follows. Section 2 provides a historical overview. Section 3 introduces the data. Section 4 introduces the pairwise and spatial regression discontinuity (RD) approaches for causal inference. Section 5 shows the results on population and local tax figures. Section 6 adds robustness exercises including a synthetic control method with the city of Graz. Section 7 discusses the mechanism of the amplifying effects and its persistence. Concluding remarks are offered in Section 8.

2 Historical background

Austria was historically the heartland of the Habsburg Empire that collapsed in 1918, and Styria was one of its crown lands. The Austrian economy was dominated by agriculture until WWII, with some industrial core areas around Vienna (Industrieviertel), in Styria (Leoben-Graz) and in Upper Austria (Wels-Linz-Steyr). During the post-WWII occupation from 1945 to 1955, markets and elections were free in all occupation zones and there were—despite dismantling activities by the Soviets—no attempts to introduce different economic legacies by the respective occupation powers. However, Hofbauer (1992) has described a westward shift in economic activity due to the post-WWII order in Austria. After WWII, and mainly after Austria's restoration in 1955, the country developed fast

⁸The standard neoclassical growth theory would predict conditional convergence of regions independent of the starting condition (Solow, 1956). Studies that support regional convergence are Barro and Sala-i Martin (1991, 1992) and Gennaioli *et al.* (2014). My findings are more in line with endogenous growth models that highlight the role of initial endowment to explain regional differences (Gennaioli *et al.*, 2012).

and is among the richest countries of the European Union today. In the following, I outline the seizing and zoning of Austria and Styria after WWII, the hostile actions of the Red Army, and the shift in population in all of Austria and across the line of contact in Styria.

2.1 The seizing of Styria and the zoning of Austria

Austria became part of Nazi Germany with the *Anschluss* in 1938 and was as such part of the liberation campaign by the Allies to liberate Europe from Nazism. In the final stage of WWII, Allied troops from the US, UK, France and the Soviet Union reached the Austrian borders from different directions. The Red Army arrived in the easternmost parts of Austria in March 1945 whereas the Western Allies did not reach Austria until the final days of WWII. Figure 2 shows the Allies' positions on the day of the Nazi-German surrender on May 8, 1945 (bold red lines). The later line of contact in Styria between the Red Army and the Western Allies (dotted line) was at that time far away from the front line and from any combat operations. Indeed, the South-Austrian region was the largest remaining coherent area under Nazi-German control in all of Europe on May 8, 1945.





Notes: The map depicts the location of Styria (bold black borders) within Austria. Bold red lines show the location of the Western Allies (UK, US and French troops) and the Red Army on the day of the Nazi-German surrender on May 8, 1945. Thin red lines with arrows show how Styria was overrun by the Allies until May 9, 1945. The locations where the Allies met became the line of contact until July 22, 1945 (dotted line). Sources: Own illustration based on Karner (2000), Iber *et al.* (2008) and Stelzl-Marx (2012).

The seizing of Styria was rather chaotic and mainly driven by the individual Allies' aiming to capture as much land as possible (Iber *et al.*, 2008; Stelzl-Marx, 2012). The Red Army in particular pushed forward until it met troops from the Western Allies (Karner, 2000). Styria was mainly seized within one day from May 8–9, 1945. The red lines with arrows in Figure 2 depict the course of the Allied liberation of South Austria and Styria. Styria's

capital city, Graz, was reached by the Red Army during the night, US troops entered Styria over the Ausseerland and Pyhrnpass and encountered the Red Army in the Enns Valley in the city of Liezen, and the UK troops ran through Carinthia towards the Mur Valley and over the Packsattel towards Graz, where advance parties of the British and Red Army met on the country road between Köflach and Voitsberg on May 9, 1945 (Stelzl-Marx, 2012).⁹ The places where the respective Allies met became the line of contact within Styria (dotted line in Figure 2), which was in place until the start of the withdrawal of the Red Army on July 22, 1945.¹⁰

Styria was seized without any land battles, and evidence on local resistance near the line of contact is absent in the respective literature. Additionally, the demarcation line was neither foreseeable nor a result of negotiation by the Allies.¹¹ The Online Appendix A offers more details on the last land battles, resistance and the seizing of Styria.

The postwar order of Austria was designed by the Allies. The foreign ministers of the Allies had agreed in the 1943 Moscow Declaration to liberate Austria from Nazi-Germany and to restore an Austrian state within the pre-Anschluss borders of 1937 (Erickson, 1950). Over the following 1.5 years, different zoning proposals were discussed on how to initially occupy Austria after the war. The aim was to partition Austria along state or—in one Soviet proposal—along district borders (Karner, 2000; Stelzl-Marx, 2012). All zoning proposals during the war did not follow the realized line of contact. The final proposal was negotiated secretly and finally agreed on July 9, 1945. Graphic representation of the different proposals, the liberation and the final zoning agreement are provided in Figure C.3 in the Online Appendix. The Soviet Union claimed and finally received, additional regions in northern Austria (Mühlviertel) for military reasons and was thus willing to give up Styria (Erickson, 1950; Slapnicka, 1986).¹² Just 13 days after the zoning agreement, the Red Army began its withdrawal from Styria. All of Styria—with the exception of three municipalities in the Ausseerland—together with the state of Carinthia and East-Tirol became part of the UK occupation zone in postwar Austria. Allied troops stayed in Austria for more than 10 years until the full restoration of Austrian sovereignty with the Austrian State Treaty in 1955.

⁹Stelzl-Marx (2012) reports that it took another two days until the line of contact was reached by larger troop contingents. Moreover, some southern regions of Styria were initially seized by Bulgarian and Yugoslavian (Tito partisans) troops under the Red Army's full or partial command.

¹⁰Throughout the paper, I refer to the 74-day-long line of contact (from May 9 to July 22, 1945) despite the fact that in some places, such as Judenburg, the Red Army withdrew on July 24, 1945, and the last of its troops did not leave Styria before early August 1945 (Stelzl-Marx, 2012).

¹¹Even in areas where the River Enns and Mur formed the line of contact, the Allies arrived simultaneously on the opposite riverbanks (Karner, 2000). There was, however, a potentially arbitrary decision by the Red Army after they reached the River Enns in Landl to follow rather than crossing the river (see Online Appendix A). I address the potential concerns of rivers on subsequent development in Section 6.5.

¹²Styria is thus the largest regional entity in all Europe that was initially seized but not permanently occupied by the Red Army/Soviet Union. However, West Berlin, parts of Vienna and the Danish island of Bornholm were also initially captured by the Red Army and handed over to the Western forces thereafter.

2.2 The Red Army in Styria

The Red Army occupied their seized parts of Styria for around 2.5 months. Figure 3 shows the assignment of Styrian municipalities to the respective liberation forces, indicates divided places and the location of Styria's capital city Graz, which is Styria's main industrial area. The Red Army treated its liberated regions much more harshly than the Western Allies. Several historical sources report (sexual) violence mainly against women, official and unofficial dismantling activities by the Red Army, and lootings and pillages by its soldiers. The official dismantling targeted so-called "German assets", i.e., mainly production facilities installed by Nazi-Germany after the Anschluss of Austria to Nazi-Germany. However, the Red Army made no special effort to distinguish between equipment installed by Nazi-Germany after 1938 and machinery already in operation before (Bischof, 1999). Iber *et al.* (2008) collected dismantling resolutions—signed by Joseph Stalin *after* the dismantling activities in Styria took place—aimed to legalize the removal of entire production plants in the steel and machinery industries. An agent of the US Office of Strategic Services (OSS) reported that:

"Russia's major motivation in evacuating Austrian equipment is obvious: To replace destroyed Soviet producing assets to the maximum extent possible." ¹³

The geography of officially dismantled plants, however, was not exogenous. Figure C.12 in the Online Appendix shows the location of 14 Styrian municipalities where production facilities were officially shipped away. Most dismantled plants—with the exception of the partitioned city of Judenburg—were located in Styria's industrial heartland along the railway line from Vienna to Graz, and thus far away from the line of contact.¹⁴

A wide body of literature also reports the informal dismantling and pillaging of assets by both the Red-Army authorities and its individual soldiers. The Red Army confiscated railroad tracks and locomotives and also dismantled the electricity infrastructure, including transmission lines, electrical overhead cables and transformers, which in turn led to severe electricity shortages in the direct aftermath of WWII (Iber *et al.*, 2008). Additionally, raw materials and semi-finished goods were largely removed from Styria (Pickl, 1995; Iber *et al.*, 2008). Moreover, Red-Army soldiers were allowed to send bundles of items back home for free, which was interpreted as an informal request to pillage (Stelzl-Marx, 2012). Thus, everything was subject to pillage: small production facilities, furniture in private apartments, farming tools and even herds of cattle were driven towards Hungary (Eberhart, 1995; Pickl, 1995; Bischof, 1999). Beer (2004) reports 1,484 notified lootings

¹³Cited after Bischof (1999). Especially former Allies of Nazi-Germany (Hungary, Romania, and, in particular, East Germany), but also Poland, Bulgaria and Czechoslovakia faced a massive decline in their capital stock due to Soviet' dismantling activities after WWII (Liberman, 1996; Bekes *et al.*, 2015).

 $^{^{14}}$ I test for potential effects of official dismantling in Section 6.3.



Figure 3: Tripartite Styria from May 9 to July 22, 1945

Notes: The map shows Styrian municipalities according to their liberation forces of the Red Army, US and UK troops. Tripartite Styria lasted for 74 days (from May 9 until July 22, 1945) before the whole Styria became part of the UK occupation zone (until 1955). Black lines within Styria show municipality borders based on the territorial status of 2011. Municipalities in white are Landl, Judenburg and St. Georgen ob Judenburg, which were partitioned among the liberation forces.

in Graz alone—while Eberhart (1995) and Pickl (1995) report an increasing incidence of pillaging during the withdrawal of the Red Army, indicating a "devil-may-care" mentality.

Lastly, there was also mass violence, mainly against women. There were 9,493 reported rapes and approximately 30,000 estimated rapes, which led to hundreds of abortions (Beer, 2004). Sexual violence in Styria by Red-Army soldiers also caused syphilis and gonorrhea epidemics (Stelzl-Marx, 2012). To sum up, Styrian regions liberated by the Red Army suffered, whereas neighboring regions did not. Dismantling, pillaging and violence made regions seized by the Red Army a less desirable place to live and to initiate economic activities. This matters in particular in the direct aftermath of WWII, when migration was still unobstructed and large waves of refugees fled from the Red Army. Refugees might thus have avoided settling in areas previously seized by the Red Army.

2.3 Population dynamics

This section describes the population dynamics during WWII and immediately after the war in Austria and shows that the short-term presence of the Red Army in Styria immediately affected population figures in areas seized by the Red Army. The crucial period of migration and refugee waves occurred in Spring and early Summer 1945. Oral history, written sources and history books report a large scale movement of people from East to West Austria after the arrival of the Red Army at the eastern border in late March 1945 until the finalization of the permanent occupation zones in July 1945 (Rathkolb, 1985; Eder and Halla, 2016). The Austrian civic population feared the arrival of the Red Army mainly due to Nazi propaganda, poor reputation and reports of crimes by Red-Army soldiers against civilians in Eastern Europe (Mascher-Pichler, 2009; Stelzl-Marx, 2012). By contrast, regions liberated and occupied by the Western Allies were the main target of both internal and external refugees due to better living conditions and food supply right after the war (Beer, 1991; Iber *et al.*, 2008).¹⁵ The period of mass migration ended with the zoning of Austria and associated travel restrictions across the occupation zone borders (*Control Agreements* among the Allies). The free movement of people across the western-Allies occupation-zone borders was reinstalled by 1947, and across the Soviet-occupation zone border by 1953. For Styria, which became part of the UK occupation zone in July 1945, there were no formal travel restriction across the 74-day-long line of contact. However, there are reports that the Red Army stopped people at checkpoints and tried to capture Nazis that fled to the west (Iber *et al.*, 2008).

The above-mentioned shifts in population from eastern to western Austria are also visible in population counts: Neither internal migration nor any westward movement of people took place until January 1945 according to the last available district data before the end of WWII (see Figure C.4 in the Online Appendix). By Fall 1946, however, the first postwar population counts based on food voucher data show a massive increase of population in the Western occupation zones, while the Soviet occupation zone had shrunk. In particular, the Soviet occupation zone and quatripartite Vienna lost more than 326,000 inhabitants from 1939 to 1946, while the overall population of Austria grew by around 2.5% during this period—despite more than 200,000 direct and indirect victims of the war—mainly due to the arrival of external refugees and higher fertility rates in the early 1940s. Figure C.5 in the Online Appendix shows population figures by occupation zone and for Vienna from 1869 to 2011 and confirms the reports of a mass exodus of people from eastern to western Austria; quadripartite Vienna and the Soviet occupation zone saw post-war population decline while the population in the Western occupation zones increased.¹⁶

The population dynamics within Styria are illustrated on the left-hand side in Figure 4 for the municipalities within 20 kilometers of the line of contact (RD sample) and in Figure C.6 in the Online Appendix for the whole of Styria. The figures on the right-hand side in Figure 4 report a rather stable number of population in Red-Army-liberated line of contact municipalities, while the adjacent places, which were liberated by US and UK troops

¹⁵Apart from large-scale internal migration, however, Austria and especially the US zone were also a favored place for ethnic German expellees from Eastern and Southeastern Europe. Radspieler (1955) and Slapnicka (1986) estimate 500,000 to one million temporary foreign refugees after the war, of which the majority left Austria for Germany in 1945 (Stieber, 1995). Nevertheless, there were more than 300,000 naturalizations (a measure for settlement patterns of foreign refugees after WWII) in Austria from 1946 to 1951; this accounts for 4.4% (Styria: 4.2%) of the total population by 1951 (StatistikAustria, 2014). According to StatistikAustria (2014), naturalization during the late 1940s and early 1950s was uniform across Austria. Therefore, external migration cannot explain the westward shift in population.

¹⁶The Austrian capital city of Vienna started to shrink before WWII because the Jewish population of Austria, mostly residing in Vienna, had already started to leave Austria before 1939.





Notes: The figures show population growth at the municipality level across the line of contact in Styria. The map on the left-hand side shows the population growth for 191 municipalities within 20 kilometers of the line of contact (RD sample). The bold black line depicts the line of contact. The right-hand figures depict the average municipality population growth for line of contact municipalities (LoCM). The upper graph shows population figures by liberation force from 1869 to 2011 (1939=100, pairwise sample) and the bottom figure from 1939 (last census before WWII) to 1946 (food vouchers data).

experienced a population boost by almost 14% from 1939 to 1946. This is remarkable given that migration across the line of contact was free, and that the Red Army has withdrawn from the region for more than one year before the population count in 1946. The line of contact, and the short-term presence of the Red Army in Styria, thus shocked regional population density. It is therefore likely that the initial settlement of refugees during Spring and early Summer 1945 account for these figures, i.e., that relocation of refugees did not occur on a large scale after the Red Army withdrew from Styria.¹⁷

An empirical test on the determinants of population growth from 1939 to 1946 in Styria reveals that—besides the relative shrinkage of places seized by the Red Army—regions close to Styria's capital city of Graz and the flat to hilly landscape relatively grew, while more rural and mountainous places relatively shrunk.¹⁸ Overall, however, the population in Styria increased until 1946 by around 8% mainly due to the arrival of refugees. To sum up, the reported facts strongly suggest that mainly Austrian internal refugees relatively avoided settling in regions initially seized by the Red Army.

¹⁷The entire period right after the war is under-studied, in particular at the subregional level. Given the data for Styria, it is possible that internal Styrian refugees returned home after the withdrawal of the Red Army, while refugees from eastern Austria, i.e., from the permanent Soviet occupation zone, remained.

¹⁸Table C.2 in the Online Appendix shows these results of regressing population changes from 1939 to 1946 on the liberation treatment and various time-invariant controls in different subsamples with Styrian municipalities. In all subsamples and independent of time-invariant controls, the Red-Army-liberated places shrunk relatively from 1939 to 1946. The shrinkage was stronger close to the line of contact.

3 Data and summary statistics

3.1 The line of contact

My dataset comprises municipality-level data on the liberation treatment after WWII. The liberation status of Styrian municipalities mainly stems from Red-Army reports of local checkpoints (Iber *et al.*, 2008). Checkpoints were often located between localities, which allows me to assign the liberation treatment at the municipality level (see Online Appendix A for more details). I cross-checked the unclear liberation treatment of Kleinlobming, Lassing, Modriach, Salla and St. Georgen ob Judenburg with the respective municipality chronicle.¹⁹ Nevertheless, the line of contact did not necessarily follow (2011) municipality borders. Some farms in the municipalities of Edelschrott and Kleinlobming and the hamlet of Piber in the city of Köflach were divided from their main municipality. I assign the liberation treatment in these cases according to liberation of the respective main locality.²⁰ The upper panel of the summary statistics in Table C.3 in the Online Appendix shows that 446 (82%) Styrian municipalities were seized by the Red Army, while 37 (7%) and 56 (10%) municipalities were liberated by US and UK troops respectively. Three municipalities (Hieflau, St. Georgen, Judenburg) were divided between the Red Army and Western forces and are excluded in all empirical analyses.

3.2 Variables

The dataset uses hand-collected archival data and data retrieved from the Statistical Office of Austria. I collect data on population figures, socio-demographic variables, sector shares, communal tax revenues and birth and death statistics for Styrian and Carinthian municipalities. Population data base on (mostly decennial) censuses and register data from 1869 to 1939 and for 1951 to 2011. I add self-compiled population data for the direct aftermath of WWII in 1946 based on food voucher statistics, and district measures for population figures during WWII. I further digitize the hard-copy municipality data on demographic variables, sector composition of residents (according to the head of the family) and birth and death statistics for 1934 to 1971 (all data whenever available). More recent data on socio-demographic variables, sector composition, local work places per sector, local numbers of firms/establishments, education level, and communal tax revenues are retrieved from the statistical database *StatCube* by Statistik Austria. I further digitize the machinery in agriculture for the 1950s and 1960s at the court district level. The main unit of analysis is the municipality's territorial status as of 2011.²¹ I follow mergers of

¹⁹I would like to thank Dr. Ernst Reinhold Lasnik for information on the municipality of Salla.

²⁰The empirical results do not change when excluding the municipalities of Kleinlobming, Edelschrott and Köflach from the main analysis. Results are available upon request.

²¹The number of Styrian municipalities decreased from 1,030 in 1934 to 542 in 2011. Mergers of municipalities during this time did not take place across the intra-Styrian line of contact. However, a

municipalities from the 1920s until 2011. Online Appendix B provides more details on data, list all relevant data sources and describes the construction of my dataset.

Table C.3 in the Online Appendix shows the summary statistics for Styria, the RD sample and for the line-of-contact municipalities. Census data for 1939 and 1951 show that Styria was an agricultural society around WWII—approximately 60–70% of the population (according to the head of the family) belonged to the agricultural sector. Data for 2011 show the sector change towards industry and services and that the average firm is small: Average workplaces per firm are around 4, and average workplaces per industrial firm are around 10. Communal tax revenue per local employee is around 400 Euro, implying an average wage sum per employee (full and part-time employees) of around 14,000 Euro. Finally, the bottom panel shows geographic characteristics of Styrian municipalities.

4 Empirical identification strategy

This section introduces the empirical identification strategies and the respective identifying assumptions for causal inference to test for the long-lasting economic impacts of the Red-Army presence after WWII. I employ pairwise estimates with contiguous line-of-contact municipalities and single- and multi-dimensional regression discontinuity designs.

4.1 Pairwise estimates with line of contact municipalities

The first empirical specification limits the analysis to contiguous municipality pairs on both sides of the line of contact. This approach exploits the liberation treatment among neighboring municipalities on subsequent socio-economic outcomes. Each Red-Armyliberated municipality is matched to one or many direct geographical neighbours liberated by the Western Allies. A pair thus consists of two municipalities that share a common land border but experienced a different liberation treatment (Red Army vs. Western Allies).²² These pairwise estimates base on Holmes (1998) and Dube *et al.* (2010) or more recently Lichter *et al.* (2020) or Galofré-Vilà *et al.* (2021). Pairwise estimates with contiguous municipalities is a form of regression discontinuity design to control for unobservable heterogeneity. The cross-sectional pairwise estimate takes the following form:²³

$$Y_{i,p} = \alpha + \beta RedArmy_{i,p} + \mu_p + \epsilon_p \tag{1}$$

 $Y_{i,p}$ is the outcome in municipality *i* in the contiguous pair *p*. The main outcome variables are post-WWII population and sector growth and communal tax revenue per employee.

territorial reform in 2015 reduced the number of municipalities further to 287, and mergers occurred across the former line of contact. Recent data will thus not allow a sharp RD design anymore.

²²In the case of multiple contiguous municipalities on the other side of the line of contact, this municipality is then repeatedly part of the respective contiguous municipality-pair.

²³Section 6.6 and Section 7.2 making use of the time dimension of some data and employing differencein-differences pairwise estimates.

However, Y_i are also pre-treatment covariates to control for smoothness before treatment. α is a constant. RedArmy_{i,p} is a dummy variable that equals one (zero) for municipality *i* in pair *p* that was liberated by the Red Army in May 1945 (liberated by the Western Allies). μ_p are contiguous line-of-contact municipality-pair dummies. The coefficient β thus captures differences between contiguous municipalities with a different liberation history. The error term is denoted by ϵ_p . Standard errors are clustered at the contiguous municipality-pair level. This pairwise estimate is a powerful method to estimate local average treatment effects in a very homogeneous sample. Equation 1 does not include any further municipality controls. Instead of including controls at the municipality level, I show the smoothness of pre-treatment covariates and time-invariant geographic characteristics in the section below. Moreover, potential contemporaneous covariates are bad controls since they are likely to also be affected by the treatment (Angrist and Pischke, 2009).

4.2 (Spatial) regression discontinuity design and sample selection

The second empirical specification employs the standard regression discontinuity (RD) design, which includes the respective hinterland of the line-of-contact municipalities. RD controls for unobservable heterogeneity across treated and non-treated units that are arbitrarily close to each other (Imbens and Lemieux, 2008; Lee and Lemieux, 2010). In the context of Styria, this means that neighboring municipalities on each side of the 74-day-long line of contact are the most comparable units of observation to estimate a causal (local average treatment) effect of unequal liberation treatment between otherwise comparable units. The baseline regression in the RD approach takes the following form:

$$Y_{i,j} = \alpha + \beta RedArmy_i + f(geolocation_i) + \phi_j + X'_i \gamma + \epsilon_i$$
⁽²⁾

 Y_{ij} denotes the dependent variable of interest (post-WWII population growth, communal tax revenue per employee, sector shares, etc.) in municipality *i* along border segment *j*. *RedArmy_i* is a dummy variable that equals one (zero) for municipalities in the east of the line of contact that were seized by the Red Army (west of the line of contact, liberated by the Western Allies). The three partitioned municipalities (Landl, St. Georgen and Judenburg), which are directly located on the line of contact, are excluded in all analyses. The coefficient β captures spatial discontinuities in the outcome Y_{ij} when crossing the line of contact from the west to the east, i.e., from the municipalities liberated by the Western Allies to those seized by the Red Army. The RD polynomial, $f(geolocation_i)$, controls for smooth functions of geographic location. I employ distance in kilometers of the nearest line of contact municipality as a single-dimensional forcing variable²⁴ and longitude and

 $^{^{24}}$ I use the closest distance in kilometers to the respective municipalities' centroids as in, e.g., Eugster *et al.* (2011). I do not use the distance to municipality borders because (i) present-day borders do not necessarily coincide with the line of contact, (ii) I can only assign localities, not the entire territory of a municipality, to a liberation force (Iber *et al.*, 2008) and (iii) the use of centroids give similar distances to the line of contact independent of a municipality's area (see the discussion in Schumann (2014)).

latitude as multi-dimensional forcing variables (Dell, 2010). The main functional forms of $f(geolocation_i)$ are a cubic polynomial fit to test for discontinuity in population figures and a quadratic polynomial fit to test for discontinuity in communal tax revenues. The polynomials for the respective outcome are derived from optimal bandwidth selection criteria (Calonico *et al.*, 2014a,b) and are discussed in more detail below. Table C.4 in the Online Appendix reports the main functional forms of $f(geolocation_i)$ and also reports all other RD polynomials that are employed in the paper. ϕ_j is a set of border segment fixed effects that captures heterogeneous geographic treatment effects *along* the line of contact.²⁵ X_i represents a vector of time-invariant geographic control variables that predict post-WWII settlement patterns independent of the liberation treatment.²⁶ α is a constant. ϵ_{it} is the error term. Standard errors are corrected for spatial dependence (Conley, 1999, 2010).²⁷ Spatial clustering addresses inference issues raised by Kelly (2019).

I employ both single-dimensional and multi-dimensional RD estimates with a common RD sample. I use the optimal bandwidth selection criteria developed by Calonico *et al.* (2014a,b) for a single-dimensional forcing variable.²⁸ Table C.5 in the Online Appendix tests the main outcome variables (population growth from 1939 until 2011 and communal tax revenue per local employee in 2011) with different polynomial fits. Average bandwidths are close to 20 kilometers. I thus define the RD sample as municipalities located within 20 kilometers of the nearest line-of-contact municipality. A cubic polynomial fit for population figures and a quadratic polynomial fit for tax figures have bandwidths closest to the overall averages. These functional forms for $f(geolocation_i)$ will thus be used in the main specifications. However, I will also employ alternative bandwidths, polynomial fits and different specifications of robust bias-corrected estimators with a data-driven bandwidth selection in a sensitivity check in Section 6.4.

4.3 Identifying assumptions and balance

The identifying assumption behind both empirical strategies (pairwise regression and RD) is that any potential outcome variable will run smoothly at the line of contact in the absence of treatment. This assumption is violated if the location of the line of contact was endogenous to municipality characteristics, which may affect post-WWII outcomes. I address potential endogeneity issues with the the line of contact as follows: (i) I restrict

 $^{^{25}}$ I divide the line of contact into three segments according to geographic consideration: The Enns Valley in the north (region of contact between the Red Army and the US troops), the central Mur Valley and the Graz region in the south (regions of contact between the Red Army and the UK troops).

 $^{^{26}}$ Geographic controls are the distance to Styria's capital Graz and distance to Graz squared, share of settlement area, elevation range, and roughness (see the discussion in Section 2.3).

 $^{^{27}}$ I use the Stata command provided by Hsiang (2010) in cross-sectional estimates and the command by Colella *et al.* (2019) for panel estimates to correct for spatial dependence of the standard errors. I will provide estimates with different spatial cutoffs in a sensitivity analysis.

²⁸Optimal bandwidth selection with multiple forcing variables (Cattaneo *et al.*, 2020) require straight cut-offs in both dimensions and are thus not feasible for the strongly curved line of contact in Styria.

the main analysis to the quasi-exogenous part of the line of contact within Styria only, (ii) I show the smoothness of economic variables across the line of contact prior to WWII, (iii) I qualitatively show that the line of contact was neither a historical border nor any other administrative border until today, and (iv) I argue that sorting into the liberation treatment did not occur. I provide evidence of these assumptions below.

Line of contact: To alleviate the identification concerns, I focus on the line of contact within the Austrian federal state of Styria alone. The extension of the line of contact toward the south follows the mountain range of the Koralpe and forms the federal state border between Styria and Carinthia, and is thus not part of the main analysis. Focusing on within-federal-state variation rules out that the results might be driven by idiosyncratic differences between federal states. Furthermore, the line of contact on May 8, 1945, was not a region of any ground battle and was not a location for any defensive wall by the Nazis (Brettner, 2013). The capture of Styria did not follow any liberation plan or any coordination among the Allies (Pickl, 1995; Iber et al., 2008; Stelzl-Marx, 2012). The liberation of Styria can be described as rather chaotic. This is also illustrated by the fact that there were many competing occupation plans by the Allies, and none of them planned to seize Styria close to the line of contact (Erickson, 1950). See more details on the exact timing of the liberation in Section 2.1 and in the Online Appendix A. I also do not find any pre-liberation sorting of people towards the West based on population counts in January 1945 (see Figure C.4 in the Online Appendix). This confirms that any division of Austria was not anticipated until the end of WWII. Moreover, the majority of the line of contact within Styria does not follow natural borders. Mountain ranges might isolate economic regions, i.e., if one valley had been liberated by the Red Army, and another by the Western Allies. However, the line of contact does not follow mountain ranges. The Allies met each other within valleys (Enns and Mur) or met in the flat to hilly landscape in the region of Graz. The sole exception of natural borders forming the line of contact are the River Enns and River Mur for some kilometers. Rivers in Styria, however, should not isolate economic regions since bridges are frequent, and both river banks have similar characteristics. Nevertheless, I will control for the potential effects of rivers in Section 6.5. Overall, the line of contact within Styria was exogenous and does not divide economic regions. Whether a certain municipality was just the last one to be seized by the Red Army was random and not foreseeable.

Balance of pre-WWII covariates: I explore the validity of my design by checking for smoothness of time-invariant geographic covariates and pre-WWII socio-economic characteristics based on the population census in 1934 and 1939. Figure 4 has already shown similar population trends across the line of contact before WWII. Table 1 shows the results of a balance test for the RD sample (mean comparison in Panel A, and single and multi-dimensional RD estimates in Panels C and D respectively) and for contiguous

-		a	,			1		Junicipalit	, covariat	05		a	20			
_	Geography			Census in 1934				Census in 1939								
	Area	Usable area	Sea level	Distance to Graz	Agrar	Industry	Unempl.	Female	Agrar	Industry	$\begin{array}{c} {\rm Farms} \\ {<}10 {\rm ha} \end{array}$	Farms/area	Self- employed	Pop/HH	Age < 18	Age >6
-	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)
Panel A: OLS du	mmy spe	cification														
Red Army	$\begin{array}{c} 0.30 \\ (0.61) \end{array}$	$5.53 \\ (5.36)$	-196.37^{***} (68.95)	-19.98^{***} (3.88)	-0.1 (6.83)	-1.37 (5.57)	$1.15 \\ (1.41)$	$\begin{array}{c} 0.15 \\ (0.40) \end{array}$	$\begin{array}{c} 0.07 \\ (6.37) \end{array}$	-2.29 (5.27)	10.37^{**} (4.44)	1.48 (1.43)	-1.09 (5.36)	-0.44^{*} (0.24)	-1.32 (0.87)	$\begin{array}{c} 0.41 \\ (0.32) \end{array}$
No. of obs. Mean of Dep. Var.	191 9.04	$191 \\ 36.97$	$191 \\ 925.75$	$191 \\ 57.66$	$191 \\ 64.97$	$191 \\ 22.35$	$191 \\ 11.96$	191 49.48	$\begin{array}{c} 191 \\ 60.69 \end{array}$	$191 \\ 25.08$	$191 \\ 54.21$	191 13.74	$\begin{array}{c} 191 \\ 49.79 \end{array}$	$ 191 \\ 4.70 $	$191 \\ 26.90$	191 7.73
Segment FE Geography FE	Yes No	Yes No	Yes No	Yes No	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes
Panel B: Pairwise	e regressi															
Red Army	-1.12 (1.09)	-0.79 (2.44)	-23.88 (29.04)	-2.80^{***} (0.72)	-2.31 (4.49)	-0.15 (3.61)	$\begin{array}{c} 0.42 \\ (0.81) \end{array}$	-0.01 (0.31)	-1.14 (4.29)	-0.93 (3.49)	1.24 (3.1)	1.27 (1.03)	-2.19 (3.17)	0.27 (0.24)	$0.46 \\ (0.65)$	0.01 (0.220)
No. of obs.	90	90	90	90	90	90	90	90	90	90	90	90	90	90	90	90
No. municipality	48	48	48	48	48	48	48	48	48	48	48	48	48	48	48	48
Mean of Dep. Var. Pair FE	9.78 Yes	29.18 Yes	1020.11 Yes	63.18 Yes	59.45 Yes	26.56 Yes	11.77 Yes	48.57 Yes	55.27 Yes	29.30 Yes	46.00 Yes	11.68 Yes	43.03 Yes	4.60 Yes	27.58 Yes	6.94 Yes
Panel C: Single-d	imension	al RDD														
Red Army	$\begin{array}{c} 0.01 \\ (0.98) \end{array}$	5.63 (7.29)	-141.15^{*} (84.48)	-7.97^{**} (3.88)	-0.34 (6.96)	-1.56 (5.56)	0.67 (1.28)	-0.03 (0.48)	$1.42 \\ (6.53)$	-2.81 (5.16)	$3.08 \\ (5.04)$	-0.06 (2.07)	-2.25 (5.26)	-0.07 (0.24)	-0.12 (1.08)	$\begin{array}{c} 0.08 \\ (0.37) \end{array}$
No. of obs.	191	191	191	191	191	191	191	191	191	191	191	191	191	191	191	191
Mean of Dep. Var.	9.04	36.97	925.75 V	57.66 V	64.97	22.35 V	11.96 V	49.48	60.69	25.08	54.21 V	13.74 V	49.79 V	4.70	26.90	7.73 X
Segment FE Geography FE	Yes No	Yes No	Yes No	Yes No	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes
Panel D: Multi-di	mension	al RDD														
Red Army	-0.02 (0.89)	0.89 (5.51)	-7.58 (64.6)	-0.81^{**} (0.37)	4.65 (6.38)	-4.13 (5.54)	-0.97 (1.28)	-0.02 (0.51)	4.76 (6.14)	-4.82 (5.37)	$1.65 \\ (4.61)$	-0.26 (1.78)	0.08 (4.96)	-0.10 (0.26)	-0.43 (1.19)	0.04 (0.38)
No. of obs.	191	191	191	191	191	191	191	191	191	191	191	191	191	191	191	191
Mean of Dep. Var.	9.04	36.97	925.75	57.66	64.97	22.35	11.96	49.48	60.69	25.08	54.21	13.74	49.79	4.70	26.90	7.73
Segment FE Geography FE	Yes No	Yes No	Yes No	Yes No	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes

Table 1: Smoothness of pre-treatment (pre-WWII) covariates across the line of contact

Notes: The table tests for smoothness across the line of contact in various pre-treatment and time-invariant covariates. Columns (1) to (4) show geographic covariates, Columns (5) to (9) show covariates based on the census in 1934, and Columns (10) to (16) based on the census in 1939. Coefficients represent shares in percentage points except for area (in km^2), average sea level (in m), distance to Graz (in km), farms per area (farms per km²) and population per household. The shares of the industrial sectors, of unemployed and of self-employed are according to the head of the family. Panels A, C and D consist of municipalities within 20 kilometers of the nearest line-of-contact municipality (RDD sample). Panel B consists of contiguous line-of-contact municipalities. Panel A employs an OLS dummy specification (Dummy=1 for Red-Army liberated municipalities, 0 otherwise). RDD estimates in Panels C and D employ a cubic polynomial fit. The estimates include segment and geography fixed effects. Smoothness of geographic and pre-treatment covariates without segment and geography fixed effects are shown in Table C.6 in the Online Appendix. Standard errors in parentheses in Panel B are clustered at the contiguous pair level and corrected for spatial dependence in Panels A, C and D. Significance levels: *** 0.01, ** 0.05, * 0.10.

line-of-contact municipalities (Panel B); it also reports the mean of the dependent variable. Despite some differences in the sample means (Panel A), I find no statistical significant discontinuities of covariates in all RD setups. The sole exception is the distance to Graz, which, given its location within the Red-Army-liberated part of Styria, is not surprising. Geographic characteristics such as area, share of settlement area and altitude (Columns (1) to (3)), sector shares of local residents, share of unemployed, farm and employment structure, gender ratios, people per household or shares of age cohorts vary smoothly across the line of contact prior to the treatment. The validity of this identifying assumption does not change if I repeat the balance test without any pair, segment and geographic fixed effects in the respective panels.²⁹ The line of contact thus divides a homogeneous region in terms of geography and pre-treatment socio-economic characteristics.

Historical or administrative borders: In addition to federal state borders, sub-regional borders (Keele and Titiunik, 2015) or long-gone historical borders might affect economic trajectories (Grosfeld and Zhuravskaya, 2015; Becker *et al.*, 2015; Bukowski, 2019). In Austria, districts form the administrative units between federal states and municipalities. Districts in Austria do not have any legislative or executive power; they have to follow the instructions from the federal or state government (Rihs, 2021).³⁰ However, the line of contact has never and still does not coincide with any district border. Furthermore, the demarcation line does not coincide with any historical border. The municipalities along the line of contact belonged to the same jurisdictional and historical entity for centuries prior to May 1945 (Duchy of Styria, Crown Land of Styria, federal state of Styria after 1918, and Reichsgau Styria during the period of Nazi Germany) and after July 1945 until today (UK occupation zone from July 1945 to 1955 and federal state of Styria) (Pirchegger, 1996). The sole exception of a different historical exposure across the line of contact was during the 74-day-long treatment period. I can thus rule out that sub-regional administrative or historical borders can shape economic outcomes across the line of contact.

Sorting into treatment: Another identification concern is whether the units of observation (municipalities) have manipulated their liberation treatment via, e.g., local resistance or negotiation. Negotiation on the location of the line of contact can be ruled out given the chaotic capture of Styria (see above) and the absence of evidence in the related literature. There was no local resistance or any land battles close to the line of contact.³¹ The closest land battle took place during the withdrawal of the German Army at Schanzsattel, around 60 kilometers east of the later line of contact (Pickl, 1995). There is, however, an anecdote

²⁹Table C.6 in the Online Appendix repeats the balance tests without fixed effects and confirms the smoothness of almost all covariates in RD estimates (Panels B to D). Distance to Graz does not show statistical discontinuities anymore, while altitude slightly differs in the single-dimensional RD.

³⁰As such, local politicians cannot directly influence the economic trajectory of a district. But as an administrative unit, regional policies might differ across district borders, i.e, for EU structural funds.

³¹The relevant literature does not report on local resistance or land battles within 60 kilometers of the demarcation line (Pickl, 1995; Pirchegger, 1996; Iber *et al.*, 2008; Brettner, 2013; Stelzl-Marx, 2012).

of the municipality of Murau. Local citizens flagged thier town with the Union Jack in the hope of avoiding capture by the Red Army (Stelzl-Marx, 2012). In the end, Murau was located around 30 kilometers to the west of the line of contact (outside the RD sample) and was liberated by UK troops anyway. To sum up, local resistance and negotiations did not affect the drawing of the line of contact.

5 Main results

5.1 Population effects

I first test effects of the liberation treatment on population dynamics, a widely used proxy in regional economics. Figure 5 shows RD plots on population dynamics in respect to the last pre-treatment census in 1939. The upper two panels confirm that the later line of contact is also exogenous to population growth before treatment without any pre-treatment discontinuities. By contrast, the liberation matters after WWII. The lower panels show a sharp negative decline in population figures for places liberated by the Red Army. The initial relative decline in population across the line of contact after the war (lower left graph) more than doubled until 2011 (lower right graph). The RD plots also uncover that the effects are rather local; there is a general decline in population from the east to the west. I explain this result with the agglomeration effects of Styria's capital city of Graz, which is located in the east of the RD sample. Section 6.1 zooms into Graz and analyzes how the short period of Red-Army presence affects larger cities and agglomerations.

Empirical estimates confirm the graphical inspection. Table 2 shows the pairwise estimates with contiguous line-of-contact municipalities (Panel A), and RD estimates with single-(Panel B) and multi-dimensional forcing variables (Panel C). Columns (1) and (2) show the smoothness of population dynamics for the entire pre-WWII period and test for a potential shock due to WWI, respectively. Columns (3) and (4) report discontinuities from 1939 to 1946 (data based on food vouchers) and from 1939 to 1951 (first post-WWII census), respectively. These estimates confirm that the liberation had an immediate effect on population figures, even though the Red Army had completely withdrawn from Styria for more than one year (1946) and more than six years (1951). Column (5) shows that the initial relative decline of around 10–14% amplifies to 22% (Panel C) to almost 26% (Panel B) by 2011. Lastly, column (6) shows these amplifying population dynamics across the line of contact from 1951 to 2011. Table C.7 reports population dynamics across the line of contact from census year to census year, i.e., in approximately ten-year steps. While the initial post-WWII decline in population is large, the relative shrinkage of places liberated by the Red Army was modest but ongoing in the decades after, but accelerated somewhat in the 1990s. The legacy of the Red Army is long-lasting and shapes population figures even decades after the Red Army had withdrawn from Styria.



Figure 5: RD plots on population dynamics across the line of contact

Notes: The figures show RD plots on population growth in percentage points across the line of contact. All plots are based on a linear fit and include segment and geography fixed effects. The upper graphs show pre-treatment (pre-WWII) population figures; the bottom graphs show population figures after the liberation treatment. The bins represent local averages of municipalities within a 5-kilometer interval. Negative (positive) distances in all RD plots represent municipalities with respect to the nearest line of contact municipality (LoCM) that were liberated by the Western Allies (Red Army). The vertical dashed lines represent the line of contact. Thin grey lines represent 95% confidence intervals.

5.2 Labor productivity: Communal tax revenues

Next, I examine whether the line of contact is visible in present-day economic variables other than population figures. One of the most compelling indicators of regional economic activity is labor productivity. Local labor productivity combines different aspects of production conditions, i.e., the capital stock (technology), skill levels of workers and laborers, and variables including economies of scale and scope (Krugman, 1991; Davis and Dingel, 2019). To address the long-lasting impact of the liberation by the Red Army on local labor productivity, I rely on the communal tax in Austria (*Kommunalsteuer*). This tax, introduced in 1994, is a payroll-based payment that employers pay for locally

	Dependent variable: Municipality population growth (in %)							
	Pre-V	WWII	Pre-W	Post-WWII				
	1869-1939	1910-1923	1939–1946	1939–1951	1939–2011	1951-2011		
	(1)	(2)	(3)	(4)	(5)	(6)		
Panel A: Pairwise	regression							
Red Army	-1.19 (6.14)	$1.41 \\ (2.26)$	-17.60^{***} (4.65)	-13.80^{***} (3.05)	-23.71^{***} (7.66)	-11.42^{*} (5.78)		
No. of obs.	90	90	90	90	90	90		
No. of munc	48	48	48	48	48	48		
Mean of Dep. Var. ^a	90.26	101.88	109.60	106.54	99.96	91.88		
Pair FE	Yes	Yes	Yes	Yes	Yes	Yes		
\mathbb{R}^2 adjusted	0.56	0.70	0.64	0.74	0.69	0.68		
Panel B: Single-di	mensional F	RDD						
Red Army	-0.24 (7.49)	1.08 (2.72)	-11.18^{***} (4.10)	-11.60^{***} (2.44)	-25.70^{***} (9.52)	-14.23^{**} (7.11)		
No. of obs.	191	191	191	191	191	191		
Mean of Dep. Var. ^a	89.84	101.40	109.26	107.12	115.44	105.97		
Seg. & Geo. FE	Yes	Yes	Yes	Yes	Yes	Yes		
\mathbb{R}^2 adjusted ^b	0.21	0.00	0.15	0.13	0.53	0.60		
Panel C: Multi-dir	nensional R	RDD						
Red Army	1.57	-0.68	-6.43*	-9.74***	-22.14***	-11.87*		
	(7.11)	(2.44)	(3.47)	(2.53)	(7.91)	(6.40)		
No. of obs.	191	191	191	191	191	191		
Mean of Dep. Var. ^a	89.84	101.40	109.26	107.12	115.44	105.97		
Seg. & Geo. FE	Yes	Yes	Yes	Yes	Yes	Yes		
R ² adjusted ^b	0.21	0.07	0.14	0.14	0.57	0.64		

Table 2: Population dynamics across the line of contact

Notes: The table tests for spatial discontinuities in municipal population growth across line-of-contact municipalities (in percentage points). Columns (1) and (2) test for discontinuities in pre-treatment (pre-WWII) population figures; Columns (3) to (5) for pre-WWII versus post-WWII population growth and Column (6) for post-WWII population growth from 1951 to 2011. Panel A consists of contiguous line-of-contact municipalities and include pair fixed effects. Panels B and C consist of municipalities within 20 kilometers to the nearest line-of-contact municipality (RD sample). Panel B uses distance in kilometers of the nearest line-of-contact municipality, and Panel C uses longitude and latitude as the forcing variables. RD estimates employ a cubic polynomial fit and include segment and geography fixed effects. Point estimates on spatial discontinuities from census to census in the Online Appendix. a) Mean of the dependent variable is in respect to 1939=100 (to 1923 in Column (2)). b) Adj. R² measures stem from Stata's reg command. Standard errors in parentheses in Panel A are clustered at the contiguous pair level and corrected for spatial dependence in Panel B and C. Significance levels: *** 0.01, ** 0.05, * 0.10.

employed people (Knirsch and Niemann, 2005). Private firms and establishments have to pay 3% of their total wage sum.³² I divide the total communal tax revenues averaged for 2010 to 2012 (to account for year-to-year volatility) for each municipality by the number

³²See the assessment of the taxable base and potential exceptions under https://www.usp.gv.at/en/ steuern-finanzen/kommunalsteuer.html. For example, firms with a monthly wage sum below €1,460 get a tax allowance. Wages in public administration, schools or hospitals are not taxed. Self-employed, among them family farmers, are excluded, too. Note that certain occupations might be industrial but belong to a public company (electricity, waste industry, etc.). Unfortunately, I cannot distinguish my employee data by type of establishment. I thus add non-taxable occupations as controls to my estimates.

of employees in 2011 within the same municipality to get a proxy of local average wages, which is in turn a proxy for municipal labor productivity.

Table 3 tests for spatial discontinuities in communal tax revenues per local employee. Column (1) shows the baseline results for pairwise and different RD estimates without any economic control variables. Columns (2) to (5) sequentially control for local economic characteristics that might differ across the former line of contact and affect the taxable base directly. These are sector shares (i.e., to control for the share of agriculture as a non-taxable sector), firm size distribution, occupation characteristics (i.e., to account for differences in blue collar workers), and non-taxable occupations such as public servants or self-employed. Lastly, column (6) includes all these controls simultaneously.³³

The estimates in Panel A and B shows a large and statistically significant decline in communal tax revenue per local employee when crossing the former line of contact from the Western Allies' to the Red Army's liberated parts. Effects are larger and more precisely estimated in the pairwise estimates with contiguous line-of-contact municipalities (Panel A) compared to single-dimensional RD estimates (Panel B). Estimates based on multi-dimensional forcing variables (Panel C), however, are smaller and barely statistical different from zero.³⁴ The results are substantial. By comparing the coefficients with the mean of the dependent variable in column (6), Red-Army-liberated municipalities face a lower communal tax base per local employee of 14% (57/410 in Panel C) up to 32% (139/431 in Panel A). These values are likely to be upward biased. For example, the wages of public servants are not part of the regressions, but this occupation group is paid equally across the former line of contact, which would level the wage differentials somewhat. Once again, the RD plots in Figure C.8 in the online Appendix depict the local nature of the decline of communal tax revenues; places closer Graz have much higher revenue figures.

These results are not sensitive to the base years employed in the regression. Table C.8 in the Online Appendix reports the same estimates with the tax revenues in 2011 only (instead of the three-year averages). As noted above, the communal tax was introduced in 1994, and data availability is only given for more recent years. However, Table C.9 in the Online Appendix tests for spatial discontinuities of other firm-related tax revenues in 1987 (business tax and an earlier type of wage sum tax) and compares them to the communal tax in 2011. I divide these tax figures by proxies of local employees (occupation data are missing for the 1980s). The results confirm that municipalities liberated by the Red Army also had less revenues in previous decades, but the effects are—given the used proxies for local employees—estimated with less precision.

 $^{^{33}}$ RD estimates on sector shares and other 2011 covariates are discussed in Section 7.2 and reported in Table 5 and Table C.22 in the Online Appendix.

 $^{^{34}}$ Tax revenues are more unequally distributed across space than population figures and as such, precise estimation in a relatively small RD sample is less feasible. In Section 6.2, I will show that in more homogeneous subsamples, tax figures also differ significantly in the multi-dimensional RD setup.

	Dependent variable: Communal tax revenues per local employee (in \in)								
		Tax base controls							
	Baseline	Sector	Firm size	Occupation	Non-tax occupation	All controls			
	(1)	(2)	(3)	(4)	(5)	(6)			
Panel A: Pairwise	e regression								
Red Army	-153.29***	-119.60***	-203.91***	-129.58***	-127.13***	-149.31***			
	(45.96)	(43.59)	(36.65)	(45.05)	(42.23)	(42.21)			
No. of obs.	90	90	90	90	90	90			
No. of munc.	48	48	48	48	48	48			
Mean of Dep. Var.	431.39	431.39	431.39	431.39	431.39	431.39			
Pair FE	Yes	Yes	Yes	Yes	Yes	Yes			
\mathbb{R}^2 adjusted	0.61	0.74	0.75	0.68	0.72	0.80			
Panel B: Single-d	imensional 1	RDD							
Red Army	-106.71*	-95.88**	-126.32***	-87.98*	-98.72**	-99.61***			
	(59.30)	(46.65)	(45.05)	(45.08)	(42.35)	(37.50)			
No. of obs.	191	191	191	191	191	191			
Mean of Dep. Var.	410.00	410.00	410.00	410.00	410.00	410.00			
Seg. & Geo. FE	Yes	Yes	Yes	Yes	Yes	Yes			
R ² adjusted ^a	0.21	0.48	0.44	0.46	0.45	0.54			
Panel C: Multi-di	imensional H	RDD							
Red Army	-74.22	-53.54	-87.91*	-47.62	-61.22	-57.35			
*	(55.60)	(50.00)	(48.64)	(44.48)	(43.56)	(43.72)			
No. of obs.	191	191	191	191	191	191			
Mean of Dep. Var.	410.00	410.00	410.00	410.00	410.00	410.00			
Seg. & Geo. FE	Yes	Yes	Yes	Yes	Yes	Yes			
R ² adjusted ^a	0.16	0.45	0.40	0.44	0.43	0.52			

Table 3: Communal tax revenues per local employee

Notes: The table tests for spatial discontinuities in municipal tax revenues per local employee (average of 2010-2012) across the intra-Styrian line of contact. Column (1) shows the baseline specification without any tax-base controls. Columns (2) to (5) include different economic variables that determine the taxable base (Sector controls: Share of workplaces in industry, share of workplaces in services (agriculture as residual); Firm size controls: Workplaces divided by the number of firms, industrial workplaces divided by the number of industrial firms; Occupation controls: Share of blue-collar workers; Non-tax occupation controls: Share of self-employed, share of workplaces in public administration). Column (6) includes all tax base controls simultaneously. Panel A consists of contiguous line-of-contact municipalities and includes pair fixed effects. Panels B and C consist of municipalities within 20 kilometers of the nearest line-of-contact municipality (RD sample). RD estimates employ a quadratic polynomial fit and include segment and geography fixed effects. Spatial discontinuities of variables that affect the taxable base are shown in Table C.22 in the Online Appendix. a) Adj. R² measures stem from Stata's reg command. Standard errors in parentheses in Panel A are clustered at the contiguous pair level and corrected for spatial dependence in Panel B and C. Significance levels: *** 0.01, ** 0.05, * 0.10.

6 Robustness checks

I test alternative specifications to show the robustness of the long-lasting legacy of the Red Army. I investigate the effects on big cities and agglomerations (Section 6.1), document heterogeneous effects along the line of contact (Section 6.2), and show that regional shocks and regional policies cannot explain persistence (Section 6.3). I also test alternative RD specifications and inference (Section 6.4), look at pseudo-lines of contact (Section 6.5),

and employ a differences-in-difference setup (Section 6.6). These exercises confirm that the Red Army's hostility after WWII causes economic backwardness in the long-run.

6.1 Going urban: SCM with the city of Graz

Estimates so far have shown that there is a decline in post-WWII economic activity across the line of contact in areas that were seized by the Red Army. However, RD plots on population dynamics (Figure 5) and communal tax revenue per employee (Figure C.8 in the Online Appendix) have also uncovered that the treatment effects are rather local. There seems to be a general increase in economic activity towards the east, i.e., towards regions that were fully seized by the Red Army. In fact, the economic heartland of Styria is the region in the east of the main RD sample around the Styrian capital city of Graz.

This section tests whether the Red-Army's legacy also matters for regions far away from the line of contact. I focus on the city and agglomeration area of Graz and compare the population dynamic after WWII with other major Austrian metropolitan areas. I employ the synthetic control method (SCM), a technique often applied in comparative studies when one or few aggregate units were exposed to an intervention (Abadie and Gardeazabal, 2003; Abadie et al., 2015). The main advantage of building a regional synthetic control is that the pre-intervention characteristics of the treated region is accurately approximated by a combination of untreated regions in the donor pool rather than by any unaffected unit alone (Abadie, 2021). In the case of Styria, the city of Graz is by far the largest city in the area, and the same applies to the metropolitan area of Graz. Graz also shares similar de-urbanization patterns with other Austrian cities since the 1960s when mainly the surrounding areas of cities experienced rapid population growth. As such, I compare the city Graz and its metropolitan area to other main cities and their metropolitan areas in the rest of Austria. I repeat this exercise twice, once with cities/metropolitan areas in West Austria that were liberated and occupied by the Western Allies after WWII, and once with cities/metropolitan areas in East Austria that were seized by the Red Army and were part of the 10-year-long Soviet occupation zone. Table C.10 in the Online Appendix shows the respective donor pools to Graz and its metropolitan area.³⁵ I standardize city/metroplitan population to 100 for the last pre-intervention census in 1939; this scaling is suggested by Abadie (2021) to correct for differences in absolute size between the units.

Figure 6 shows the main outcome of the SCM. The left-hand side graphs plot the population dynamics of Graz and its metropolitan area versus the respective synthetic control unit. The upper and middle graphs show that Graz and its metropolitan area grew moderately after WWII. The population of Graz and its metropolitan area grew by around 25–40%

³⁵The donor pool consists of all 30 cities with more than 10,000 inhabitants in 1939. Vienna is excluded because of its quadripartite occupation status. Metropolitan areas consist of all municipalities within a 25 kilometer great-circle distance of the respective city with the same occupation treatment from 1945–1955.



Figure 6: Synthetic control method with Graz

City of Graz versus cities in the Western occupation zone

Notes: The figures compare population growth in the Graz region to population growth of its synthetic control units (SCU). The upper four graphs compare Graz with other Austrian cities and metropolitan areas (municipalities within 25 kilometers of the respective city) that were seized and occupied by the Western Allies. The bottom graphs compare Graz with other Austrian cities that were seized by the Red Army in 1945 and occupied by the Soviet Union from 1945 to 1955. The respective donor pools consist of Austrian cities with more than 10,000 inhabitants in 1939 (see Table C.10 in the Online Appendix). The graphs on the left compare the city of Graz (metropolitan area of Graz) with its SCU; the graphs on the right show the relative change in population of all major cities (all major metropolitan areas) compared to their respective SCU. SCUs are matched over population trends before WWII and normalized to 100 for 1939. The SCU for Graz consists of Feldkirch (40.5%), Bregenz (33.7%) and Bad Ischl (25.8%). The SCU for the metropolitan area of Graz consists of Feldkirch (57.1%), Bad Ischl (32.3%) and Innsbruck (10.6%). The SCU for Graz with eastern cities consists of Schwechat (28.4%), Waidhofen an der Ybbs (23.3%), Ternitz (19.7%), Mödling (14.5%), St. Pölten (9.0%), Hollabrunn (4.1%) and Amstetten (1.0%).

Graz vs Soviet SCU

---- Soviet cities vs SCU

----- Synthetic Graz

Graz

by 2011, while the population of the respective synthetic control units consisting of cities/metropolitan areas in West Austria doubled. The right-hand graphs in Figure 6 plot differences between all units of the donor pool, including Graz/metropolitan area of Graz, to their respective synthetic control units. This is a way of inference of the SCM suggested by Abadie et al. (2015). Graz and its metropolitan area are clear outliers when compared to other cities/metropolitan areas in West Austria. By contrast, the graphs on the bottom in Figure 6 compares the city of Graz with cities in East Austria, i.e., cities that were seized by the Red Army and occupied by the Soviets for more than 10 years. The population dynamics of Graz seem to follow the eastern legacy with long-run Soviet occupation. I also show that the SCM is not sensitive to the inclusion of other matching variables in the pre-treatment period. Figure C.9 in the Online Appendix shows SCM plots with matching over population dynamics and socio-economic variables (occupational shares, density and gender). Matching with alternative variables confirms the rather weak growth performance of Graz. Reducing the donor pool to other state capital cities show that Graz has by far the weakest population dynamics of all state capitals in West Austria (bottom graphs in Figure C.9). To sum up, focusing on Graz and its metropolitan area shows that urban places are also affected by the Red Army's legacy. The analysis with East and West Austrian cities shows that the short liberation treatment of 74 days has similar effects as a 10-year lasting occupation by the Soviet Union.

6.2 Heterogeneous effects along the line of contact

In this section, I examine differences in long-run economic outcomes along the line of contact. I first examine whether the liberation of Styria led to different economic trajectories along the line of contact between the Red Army on the one hand and US and UK troops separately on the other hand and whether these initial differences persist to the present day. While the Red Army was advancing from the east, US troops liberated parts of the northwest of Styria (Enns Valley) and the UK troops liberated parts of central- and south-west Styria (Mur Valley and Graz region). Figure C.10 in the Online Appendix shows these regional subsamples of heterogeneous liberation treatment, and Figure C.11 depicts the respective population diagrams from 1869 to 2011. Generally, the relative economic decline of Red-Army-liberated places was strongest compared to neighboring US liberated regions (see Columns (1) to (3) in Table C.11 in the Online Appendix). Right after the war, streams of refugees mainly targeted regions liberated by the US because of better food provision there (Slapnicka, 1986). Depending on the model, this initial relative decline in population in northwestern Styria doubled to 33% (single-dimensional RD) to 56% (pairwise estimates) by 2011 in Red-Army-liberated places. In the Mur Valley and Graz region, where the Red Army met the UK troops on May 8, 1945, the initial decline in population was less severe but is still substantial today. Moreover, communal tax revenues

per local employee are estimated more precisely in this subsample; even the coefficient of the two-dimensional RD is highly statistical significant (Panel C in column (6) in Table C.11). Lastly, Columns (7) to (12) include the extension of the line of contact along the Styrian-Carinthian state border, and also examine the effects for the state-border sample alone. This segment of the line of contact is not exogenous and follows a mountain range. The results are comparable with the main results.³⁶ I conclude that initial differences, mainly of refugee flows, persist and amplify until 2011, that state borders do not affect the outcome, and that the effects on tax figures as a measure of local labor productivity are often estimated with more precision in regional and more homogeneous subsamples.

6.3 Regional shocks and policies

Regions on either side of the line of contact might have been affected by different regionspecific shocks or policies during or after WWII, which in turn may (partially) explain present-day differences. I thus examine whether local shocks and policies such as aerial bombing by the Allies during WWII, officially reported dismantling activities by the Red Army during their 74-day-long presence in Styria, the construction of local infrastructure after WWII and eligibility for regional structural funds from the European Union matters in explaining spatial discontinuity in economic activity across the line of contact.³⁷ Localities that are affected by the respective regional shocks or policies are illustrated in Figure C.12 in the Online Appendix. Graphical inspection reveals that these shocks or policies do not differ across the line of contact.³⁸ However, I also examine their effects empirically. The results in Table C.12 (population figures) and Table C.13 (tax figures) in the Online Appendix repeat the main empirical specifications (pairwise estimates, single- and multidimensional RDs) and include dummy variables that equal one for municipalities that were affected by the respective shocks or policies (zero otherwise). Spatial discontinuities are hardly affected by the step-wise and simultaneous inclusion of these dummies (compared to the baseline results in Tables 2 and 3). If anything, the precision of the estimates—mainly for the tax figures—increases. The respective shocks or policies themselves have ambiguous and almost no statistically significant effects on population and tax figures.³⁹ Thus.

³⁶Estimates in some specifications are not always precise because of the rather small sample size.

³⁷Data on air strikes during WWII are from Ulrich (1978) and the localities of officially dismantled plants from Iber *et al.* (2008). The respective dummy variables include targeted municipalities and their direct neighbours. Municipalities exposed to infrastructure investments since WWII include all municipalities within a 10 kilometer distance of the nearest highway slip road in 2011. Local eligibility for the European Regional Development Fund (ERDF) for 2000 to 2006 period stem from the European Commission (https://ec.europa.eu/regional_policy/en/funding/erdf/) and include 'Objective 2' and 'Phasing-out' regions. There was no EU ERDF funding before Austria joined the EU in 1995.

 $^{^{38}}$ I am not aware of any other shocks or polices that might differ across the line of contact. For example, there were no combat operations near the line of contact (see Section 4.3) and the Marshall Plan from 1948 to 1952/3 only marginally targeted Styria (Hofbauer, 1992) (no data at the local level available).

³⁹Tax figures are positively correlated with areal bombings during WWII, and negatively correlated with ERDF eligibility. This is, however, explained by the nature of the respective policy: Bombing targeted economic centers that are also richer today, and the ERDF has targeted present-day poor regions.

region-specific shocks and policies, including the loss of tangible assets due to bombing and dismantling, cannot explain spatial discontinuity in present-day economic activity.

6.4 Sensitivity analyses

This section tests whether the main results are sensitive to alternative RD specifications, different bandwidth choices or alternative ways to calculate the standard errors, and examines a multiple-hypothesis test. All tables and figures of the respective sensitivity analyses are reported in the Online Appendix C. First, I use different functional forms of the RD polynomial, $f(geolocation_i)$. Table C.14 examines spatial discontinuities for linear, quadratic, cubic and quartic polynomial fits in both single- and multi-dimensional RDs, and Figure C.7 shows coefficient plots of spatial discontinuities in population figures with respect to 1939 for all census years. Results are robust with somewhat smaller and less-precise estimates in the multi-dimensional RD for population figures only. Second, I vary the RD sample and employ bandwidths of 10, 15, 25 and 30 kilometers and robust bias-corrected estimators based on various RD polynomials with a data-driven equal and unequal bandwidth selection (Calonico et al., 2014a,b). Tables C.15 and C.16, respectively, report that the baseline estimates are robust to these alternative specifications; the precision of the estimates even increases when allowing for unequal bandwidths, while estimates in very small samples (i.e., 10km) somewhat loose precision.⁴⁰ Third, Table C.17 reports statistical inference with varying cutoffs for spatially clustered standard errors and for alternative specifications of the standard errors. Results are robust, but inference becomes ambiguous when the spatial cutoff is increased. Generally, larger cutoffs also yield higher statistical significance for communal tax revenues in the multi-dimensional RDs. Lastly, Table C.18 shows that the effects on the main outcomes (population and tax figures) remain statistically significant when adjusting the p-values for multiple hypothesis testing based on List *et al.* (2019) and adapted for regressions by Barsbai *et al.* (2020).

6.5 Soviet proposal and pseudo-treatments

The results might be driven by the respective Allies capturing specific regions, by segments of the line of contact that consist of natural borders, and by a general economic east-west gradient in economic activity. First, I account for strategic liberation of Styria by looking at the only occupation proposal by the Soviet Union from 1944 that aimed to divide Styria. This proposal suggested dividing Styria along mountain ranges and district borders. Second, I propose the River Mur in Central Styria as a natural east-west divide. This pseudo-line of contact accounts for the fact that parts of the realized line of contact follows the River Enns in northwestern Styria, and for some kilometers the River Mur in the Mur Valley. Third, I apply two rather technical pseudo-lines of contact by shifting the line of

⁴⁰Figure C.13 shows the respective regional samples for bandwidths of 10, 15, 20, 25 and 30 kilometers.

contact 20 and 40 kilometers eastward (within the Red-Army-liberated parts). Table C.19 in the Online Appendix examines pairwise and RD estimates across all these pseudo-lines of contact in Styria, and Figure C.14 shows the location of the respective pseudo-samples. The results confirm that the initial Soviet occupation proposal, natural borders and a general east-west gradient in economic activity are not the driver of the main results.⁴¹

6.6 Differences-in-difference with population figures

I also compare the RD estimates of the population figures with a differences-in-difference (DiD) approach. Table C.20 in the Online Appendix shows various DiD specifications with annualized population growth rates. I run a pairwise and conventional DiD with line of contact municipalities and also examine the treatment effects from census year to census year. The results are comparable with the RD estimates: The relative annualized decline in population in Red-Army-liberated places is 0.46% (Column (3) in Table C.20); this implies a relative shrinkage of around 33% from 1939 until 2011 (0.46×72 years), which is slightly above the pairwise and RD estimates in Section 5.1. Concerning subperiods, the population shrinks mainly in the initial period after WWII. With an enlargement of the bandwidth of included municipalities in the DiD to 5, 10 and 20 kilometers, the effects start to fade out, pointing to the local structure of the effect. I conclude that a differences-in-difference approach yields similar or even larger effects than the RD approach.

7 Mechanism of long-run regional inequality

The analysis has shown that the 74-day-long line of contact after WWII between the Red Army and the Western Allies has long-run effects on regional economic activity. This section examines mechanisms to explain the persistent and amplifying effects of population figures (Section 7.1), and provides evidence that occupation-specific migration (Section 7.2) and hampered investments (Section 7.3) can explain lower communal tax figures.

7.1 Age-specific migration and subsequent fertility differences

I examine the shift in demographic composition of local residents from 1939 to 1951 across the line of contact in terms of age and gender and show how these initial differences explain the amplifying effects over time. I focus on the line of contact municipalities and run pairwise estimates (see equation 1). Panel A in Table 4 reports the general decline of population in Red-Army-liberated places (as shown in Section 5.1) and compares this by gender (Columns (2) and (3)) and by age cohorts (Columns (4) to (6)). The relative decline of the male population is somewhat larger than for females in Red-Army-liberated

 $^{^{41}}$ Table C.19 reports 36 estimates of which 2 show weak significant spatial discontinuities. However, the signs of the coefficients are ambiguous (one is negative, one positive) and can also be explained by chance.

places.⁴² However, I find large differences with respect to the change in the age cohort structure. The relative shrinkage of young people below the age of 20 is large, while there is no significant relative shrinkage of elderly people from 1939 to 1951 (aged above 65). This implies that most likely families with children and young adults settled in regions liberated by the Western Allies, and that Red-Army seized places become relatively older.

Panel A	Dependent variable: Municipal population growth from 1939 to 1951									
		By g	ender	By age cohorts						
	Baseline	Female	Male	Young (below 20)	Working age (20–65)	Old (above 65)				
	(1)	(2)	(3)	(4)	(5)	(6)				
Red Army	-13.80^{***} (3.05)	-10.34^{**} (4.51)	-15.89^{***} (4.01)	-22.43^{***} (6.97)	-10.98^{***} (3.70)	-6.47 (5.13)				
No. of obs. Mean D.V. ^a Pair FE R^2 adj.	90 106.54 Yes 0.74	90 110.97 Yes 0.49	90 102.18 Yes 0.62	90 134.20 Yes 0.54	90 93.81 Yes 0.58	90 121.35 Yes 0.60				
Panel B	Annua	Mechanism								
	$\frac{1^{\rm st}\ {\rm generation}}{(1951/1961)}$	$\begin{array}{c} 2^{\mathrm{nd}} \text{ generation} \\ (1971 - 1991) \end{array}$	$3^{ m rd}$ generation $(2001/2011)$	Pooled (1951-2011)	Total effect (60 years)	$\frac{\text{Marriages of}}{1^{\text{st}} \text{ generation}}$				
	(1)	(2)	(3)	(4)	$(5) = (4) \times 60 \mathrm{y}$	(6)				
Red Army	-0.30^{***} (0.09)	-0.07 (0.08)	-0.25^{***} (0.09)	-0.19^{***} (0.06)	-11.17%	-0.22^{***} (0.05)				
No. of obs. Mean D.V. ^b Pair FE R^2 adj.	180 1.27 Yes 0.44	270 0.30 Yes 0.21	180 -0.25 Yes 0.41	630 0.43 Yes 0.48		180 0.92 Yes 0.35				

Table 4: Selective migration and subsequent natural population growth

Notes: The table tests for spatial discontinuities across the line of contact in various demographic characteristics in percentage points. The estimates consist of contiguous line-of-contact municipalities and include pair fixed effects. Panel A divides the shift in population from 1939 to 1951 (Column (1)), by gender (columns (2) and (3)), and age cohorts (Columns (4) to (6)). Panel B tests for differences in annualized natural population growth (birth rate minus death rate) for three time periods based on decennial censuses from 1951 to 2011 (Columns (1) to (3)) and the pooled post-WWII effects (Column (4)). Column (5) calculates the total population effects across the line of contact for the entire post-WWII period. Column (6) tests for differences in marriage patterns based on the censuses in 1951 and 1961. Estimates per census year, and estimates separated by birth and death rates are shown in Table C.21 in the Online Appendix. a) Mean of the dependent variable (Mean D.V) is in respect to 1939=100. b) Mean of the dependent variable (Mean D.V) is in respect to 1939=100. b) Mean of the dependent variable (Mean D.V) is in percentage points and in numbers of marriages per 100 people in Column (6)). Standard errors in parentheses are clustered at the contiguous pair level. Significance levels: *** 0.01, ** 0.05, * 0.10.

This shift in age structure has huge implications for the long-run trajectory of population figures. Panel B in Table 4 reports differences in annualized natural population growth rates (birth rates minus death rates) across the line of contact for different post-WWII

 $^{^{42}}$ Generally, the male population in the sample stays relatively stable from 1939 to 1951, while the female population grows (see the mean of the dependent variables (Mean D.V.) in Panel A (1939=100)). This is an indirect evidence for a lack of men after WWII due to casualties and prisoners of war.

subperiods (Columns (1) to (3)) and the pooled effect for the entire post-WWII period (Column (4)). Table C.21 repeats these estimates based on individual census years and shows the differences in birth rates and death rates separately. Birth rates (death rates) in Red-Army seized places are mostly below (above) the numbers in Western-Allies liberated places. This is likely the result of the shift in the population structure by age: The loss of children and young adults in the direct aftermath of WWII results in lower birth rates and higher death rates in the following decades. The pooled effect of around 0.19% per year implies an amplifying effect of the relative shrinkage of population of around 11% until 2011 (0.19 × 60 years from 1951 to 2011, see Column (5)). This number matches one to one to the estimated amplifying effect in population figures in Column (5) in Table 2. The decline in young people until 1951 is also visible in differences in the marriage rates in the two decades after WWII (Column (6)), which corroborates the natural population figures. I conclude that conflict-related migration has a distinct age pattern that leads to amplifying effects of population figures even in the decades after.

7.2 Shift in employment structure

I test whether the migration right after WWII also affects the composition of the workforce in terms of sector development across the line of contact. I focus on municipalities at the line of contact and use detailed population census data from the pre-war period (1934 and 1939 censuses) and the censuses from 1951 onward. For the shift in the occupation structure across the line of contact, I run the following pairwise fixed effects model:

$$Y_i = \beta (RedArmy_{i,p} \times PostWWII_t) + \gamma (\mu_p \times PostWWII_t) + \delta_t + \epsilon_p \tag{3}$$

 Y_i represents the outcomes of interest. $RedArmy_{i,p}$ is a dummy variable that equals one for municipality *i* in pair *p* that was seized by the Red Army (zero otherwise). μ_p are pair fixed effects for contiguous line-of-contact municipalities. $PostWWII_t$ is a dummy variables that equals one for the periods after WWII (zero otherwise). δ_t are year fixed effects. ϵ_p is the error term with standard errors clustered at the pair level. Variation stems from within contiguous pair differences across the line of contact over time.

I first inspect coefficient plots based on an event-study design in Figure 7 before turning to the regression outputs. I run different event-study setups based on the census years 1934 to 2011 for occupational shares of agriculture, industry and services separately. Data are based on the head of the family and differences are standardized to zero for the last pre-WWII census in 1939. First, the event study setup confirms the exogeneity of the line of contact since pre-treatment differences are not distinguishable from zero (confirming a parallel pre-treatment trend). Second, the event-study setup reveals a shift in the occupation shares of local residents until 1951: Places seized by the Red Army became more agricultural, while places liberated by the Western Allies became more industrial until 1951. However, these initial post-WWII differences somewhat fade out in the subsequent decades and even reverse for the industrial shares. Third, the service sector—very underdeveloped around WWII with an employment share of roughly 14% in the pairwise sample—does not seem to be affected initially by the liberation treatment, but the graph reveals evidence for a delayed structural change away from agriculture and industry towards the service sector in places seized by the Red Army.



Figure 7: Shift in the sector shares of residents' occupation

Notes: The figures depict coefficient plots of an event-study design for sector shares (in percentage points) of residents at the municipality level (according to the head of the family). The coefficients stem from fixed-effects estimates with contiguous line-of-contact municipalities and include pair and year fixed effects. All sector shares are standardized to the last pre-treatment census in 1939. Dashed vertical lines represent the period of WWII, the grey solid lines show the period of the Red Army's presence in Styria for 74 days. Thin grey lines around the coefficients represent 95% confidence intervals, which are based on standard errors clustered at the contiguous pair level.

The main insights from the event-study setup are confirmed in the regressions in Panel A in Table 5. The share of residents belonging to agriculture increases by around 4.20% (Column (1)) and the industrial share declines by the same amount in places seized by the Red Army (Column (2)), while the shares of the service sector and public administration are not affected (Columns (3) and (5)). Surprisingly, there are more self-employed people in Red-Army-liberated places (Column (6)), which could reflect worse labor market conditions finding an ordinary job. The relative decline in population figures in Red-Army-liberated places is also visible in local living conditions. Column (7) shows that people live less densely in places initially seized by the Red Army. However, all these numbers on sector shares are based on population censuses according to the head of the family, and thus do not reflect one to one differences in local workplaces since people might commute or might have different dependence ratios. Nevertheless, the relative increase of agriculture in Red-Army seized places is reasonable given that farmers—and in particular family farmers with their farmsteads—were less likely to escape the arrival of the Red Army or Soviet occupation. Farmers remain, while other professions are more likely to move.
Panel A	Dependent variable: Change of economic variables from $1934/39$ to 1951								
-		Sectoral shares (in %)					Pop. characteristics		
-	Agrar	Industry	Service	Public admin ^a	Liberal professions ^a	Self- employed ^a	Pop/HH		
-	(1)	(2)	(3)	(4)	(5)	(6)	(7)		
Red Army \times PostWWII	4.20***	-4.19***	-0.02	-0.78**	0.20	2.73***	-0.60**		
	(1.36)	(1.27)	(0.75)	(0.37)	(0.21)	(0.88)	(0.27)		
No. of obs.	270	270	270	180	180	180	270		
Mean of Dep. Var.	54.79	30.88	14.33	1.64	1.69	20.43	4.64		
Pair $FE \times Year FE$	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
\mathbb{R}^2 adj.	0.59	0.65	0.34	0.52	0.60	0.87	0.43		
Panel B Dependent variable: Economic variables in 2011									
-	Workp	laces by sector	(in %)	No. of workplaces		Pop. characteristics (in %)			
-	Agrar	Industry	Services	Workplaces per pop	Workplaces per firm	Compulsory schooling	Unemployed		
-	(1)	(2)	(3)	(4)	(5)	(6)	(7)		
Red Army	7.68**	1.26	-8.94**	-0.01	0.44	1.96***	-0.59		
	(3.35)	(2.88)	(3.76)	(0.03)	(0.43)	(0.74)	(0.38)		
No. of obs.	90	90	90	90	90	90	90		
Mean of Dep. Var.	24.66	22.02	53.31	0.33	3.73	29.24	4.24		
Pair FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes		
\mathbb{R}^2 adj.	0.68	0.69	0.62	0.40	0.55	0.61	0.41		

	1 . C	. • 1	1011	1	•	1	•	0011
Table 5: Sector	shifts	until	1951	and	economic	conditions	1n	2011

Notes: The table tests for spatial discontinuities across the intra-Styrian line of contact in the shift of economic variables until 1951 (Panel A) and differences in workplaces and residents' characteristics in 2011 (Panel B). The estimates consist of contiguous line-of-contact municipalities. Panel A employs pairwise difference-in-differences estimates with pair and year fixed effects and include the censuses of 1934, 1939 and 1951. Sector shares and self-employment in columns (1) to (6) are according to the head of the family and divided by total population. Columns (7) divides total population by the number of local households. The bottom panel employs pairwise regressions with pair fixed effects. Columns (1) to (3) divide the number of local workplaces by local population and the number of local firms, respectively. Columns (6) and (7) divide the respective characteristics of local residents by the overall number of municipality population. a) Data only for 1939 and 1951. Standard errors in parentheses are clustered at the contiguous pair level. Significance levels: *** 0.01, ** 0.05, * 0.10.

I also test for spatial discontinuities across the line of contact in 2011 based on local workplace statistics, and education and unemployment data from local residents.⁴³ The results are based on pairwise estimates (see equation 1) and are reported in Panel B in Table 5. Spatial differences in sector shares across the line of contact are more pronounced in terms of local workplaces than occupation data according to the head of the family. Places seized by the Red Army have a workplace share in 2011 in agriculture that is around 8% larger than in neighboring places, and fewer workplaces in the service sector of around the same magnitude, while workplaces in the industrial sector are almost identical (see Columns (1) to (3)). These figures may well explain present-day differences in communal tax revenues per local employee as a proxy for differences in labor productivity. However, places do not differ in terms of local workplaces per population or firm (Columns (4) and (5)), indicating that tax figures do not necessarily reflect increasing returns to scale

⁴³There are no workplace statistics at the municipality level available until the late 20th century.

due to larger firm entities. Finally, the population residing in Red-Army-liberated places have on average a lower educational level—the share of residents with only compulsory education is about 2% higher, while unemployment figures do not differ across the line of contact.⁴⁴ Further variables of the economic and socio-demographic structure in 2011 and the respective single- and multidimensional RD estimates are shown in Table C.22 in the Online Appendix. Generally, RD estimates with these variables yield similar coefficients but are less precisely estimated. To sum up, places seized by the Red Army are still more agricultural and lack workplaces in the services sector today. These differences likely originate from the occupation composition of migrants in the direct aftermath of WWII.

7.3 Early investments

So far, the mechanisms to explain long-run regional economic inequality highlight the crucial role of sorting of migrants by age and occupation. However, the Red Army during their 74-day-long presence also officially dismantled industrial plants, and its soldiers looted in the region. This might have affected the capital stock in Red-Army-liberated places, too, which hinders fast economic development.

Ideally, I should analyse the shift in the local capital stock in pre- and post-WWII periods to account for the long-run effects of dismantling and looting. However, such data are not available at the municipality or firm level (only at the federal state level). The results in Section 6.3 show that neither controlling for dismantling nor dismantling itself can explain the economic trajectory, while informal looting cannot be addressed empirically. Nevertheless, this section aims to find some evidence on the differences in the capital stock accumulation across the line of contact. I rely on a unique measure in this regard: machinery in agriculture. Austria published detailed agricultural machinery data at the court district level after WWII. Court districts are the administrative unit between municipalities and districts. I assign court districts in South Austria to their liberation treatment and run cross-sectional OLS regressions to test whether Red-Army-liberated places had less agricultural machinery after WWII.⁴⁵

Columns (1) and (2) in Table 6 show the differences in agricultural machinery between Red Army and Western-Allies seized court districts. Each coefficient with the respective

⁴⁴Around 80% of school expenses at the municipality level are remunerated by the federal and state level, which rules out local finance differences as a source of different school attainment (Mitterer and Seisenbacher, 2020). The lower educational level in Red-Army-seized places might just reflect the higher share of agriculture, mainly family farmers, which have a lower educational attainment in general.

⁴⁵In particular, I run the following OLS regression: $Y_c = \alpha + RedArmy_c + \epsilon_c$, where Y_c is agricultural machinery, α is a constant and ϵ_c is the error term. $RedArmy_c$ is a dummy variables equal one if the court district c was mainly seized by the Red Army (zero otherwise). Note that the line of contact does not always follow court district borders. The assignment of the treatment at the court district level (by a dummy or by the municipality share) does not affect the estimates. I include court districts along the within-Styrian and Styrian-Carinthian line of contact (line of contact court districts and their respective neighbors). Results with different treatment assignments and within Styria are available upon request.

	Dep	endent variable: N	fachinery per 1,00	00 workers
	OL	S differences	Propensi	ty score matching
			<i>match over:</i> for cupation share	farm structure and oc- res in 1939
	1953	1962	1953	1962
	(1)	(2)	(3)	(4)
Panel A: Tractor versus ho	orse-based pro	oduction		
Tractors all	-9.98** (3.99)	-34.70^{**} (15.07)	-7.12^{*} (3.56)	-31.11^{**} (14.68)
Trailer all	-14.43^{**} (5.98)	-41.41^{*} (22.19)	-18.17^{**} (7.96)	-54.77^{*} (28.26)
Trailer for tractors	-11.21^{**} (4.15)		-10.65^{**} (4.81)	
Trailer for horses	-3.23 (2.53)		-7.52 (5.23)	
Tillage machines	-4.82^{*} (2.60)	-11.54^{**} (4.85)	-4.33^{*} (2.51)	-11.84^{**} (5.18)
Tillage machines for tractors	-0.48 (0.44)	-7.85^{***} (2.78)	-0.82^{*} (0.48)	-9.03^{***} (3.18)
Tillage machines for horses	-4.33^{*} (2.29)	-3.69 (2.97)	-3.51 (2.33)	-2.81 (3.20)
Panel B: Cropland farming	5			
Cultivation machines ^a	-42.11^{***} (13.28)	-120.37^{***} (32.46)	-40.23^{***} (11.13)	-140.84^{***} (38.68)
Harvesters ^b	-18.96^{***} (5.93)	-54.13^{***} (12.82)	-23.44^{***} (4.14)	-66.55^{***} (11.97)
Panel C: Dairy farming				
Hay machines ^c	-33.40^{*} (17.31)	-114.90^{***} (31.7)	-29.04 (23.77)	-118.81^{***} (37.29)
Milking machines	-2.97^{**} (1.12)	-24.70^{*} (12.31)	-2.84^{**} (1.22)	-20.77 (14.37)
No. of units	38	38	27	27

Notes: The table shows differences in agricultural machinery per 1,000 employees in agriculture among Red-Army- and Western-Allies-liberated areas in South Austria. Each coefficient and its respective standard error stem from a single regression. The sample consists of court districts along the within-Styrian and Styrian-Carinthian line of contact (court districts along the line of contact and their respective neighboring court districts). Census data from 1951 (1961) are merged to the agricultural census in 1953 (1962). Matching variables for columns (3) and (4) include average agricultural residents per farm in 1939, share of agriculture and share of industry in 1939. a) Crop cultivation machines include: plows, harrows, sugar beet hoes, tillage machines (for horses and tractors), network harrows, sewing machines and fertilizer spreaders. b) Harvesters include: sheaf-binding harvesters, combine harvesters and potato harvesters. c) Hay machines include: hay tedders, hay rakes (various types), pasture and hay loaders. Significance levels (robust standard errors): *** 0.01, ** 0.05, * 0.10.

standard error stem from a single regression. Generally, places seized by the Red Army lacked agricultural machinery in 1953 and also in 1962. There were almost 10 tractors less per 1,000 employees in agriculture (according to the head of the family) in 1953 in

Red-Army-liberated parts (Column (1) in Panel A). This is a large number given that tractors in agriculture were almost absent before WWII and in 1953, only 17.9 tractors were in operation per 1,000 employees in agriculture. The delay in mechanization is also confirmed when comparing machinery that can be used with the tractor (the newly arriving technology after WWII) or with horses (the dominant but old technology). I test potential differences for trailer and tillage machinery according to the means of the respective drive unit (tractors vs horses) in Panel A. While machinery that is used with the tractor—and tractors themselves—is less common in Red-Army-liberated places, differences for horsedriven machinery are not statistically different from zero indicating that farmers in places seized by the Red Army were somewhat restricted to the old technology, while their neighbors steadily started to invest in the new one.

Panel B and C in Table 6 test for differences in machinery used in cropland and dairy farming. Once again, Red-Army-liberated places lack behind adjacent places. Moreover, the effects in Panels B and C also show that the resulting differences across the line of contact are not driven by potential differences in agricultural practices in different subregions, i.e., cropland vs. dairy farming. I also employ a propensity score matching approach to account for potential pre-treatment differences in the agriculture sector across the line of contact. The matching results remain stable (Columns (3) and (4)). Moreover, Table C.23 in the Online Appendix employs an alternative matching technique, and also shows differences in machinery for other sub-categories. In conclusion, my analysis of agricultural machinery after WWII gives some indication of a delayed investment pattern into the capital stock in regions initially seized by the Red Army. If these findings can be transferred to other economic sectors, it is likely that these figures are a proxy for local labor productivity.

8 Conclusion

Short-run exposure to the Red Army after WWII has affected regional economic activity until today. My study disentangles the regional exposure to hostility from other conflictrelated shocks that often confound long-run outcomes. I document that an initial population shock that resulted from people escaping the arrival of the Red Army amplifies as time goes by. Regions that were initially seized by the Red Army became relatively older, resulting in lower fertility and higher death rates in the decades after. Selective migration patterns also likely account for ongoing economic backwardness in terms of structural change and local wage-sum tax revenues as a proxy of local labor productivity. While farmers tend to stay, workers in non-agricultural occupations fled the arrival of the Red Army and they shifted industrial employment and boosted the service sector in their destination. I also find evidence for delayed investments in places initially seized by the Red Army.

My findings provide novel insights that aid better understand of the long-run economic effects of conflicts. While studies mainly focus on the effects of destroying tangible and intangible assets, the crucial rule of selective migration due to conflict is so far understudied. This is surprising given the fact that conflicts cause migration. Currently, around 12 million Ukrainians—mainly women and children—are internally or externally displaced persons. Any development plan for regional development after conflicts should certainly address how to bring these people back. Anthologies and blueprints on how to reconstruct Ukraine, for example, mainly focus on the rebuilding of tangible assets and institutions, while strategies for return migration are a side note only (Becker *et al.*, 2022; EconPol-Forum, 2023). Eichengreen (2023) is one of the few advocates for fostering re-migration after the war, with housing stock and voucher programs to motivate people to return. My study shows that such initiatives need a much higher emphasis to foster recovery.

The short-lived line of contact between the Red Army and the Western Allies also provides a unique setting to understand the ongoing European economic east-west divide. More than 30 years after the fall of the Iron Curtain, countries in Central and Eastern Europe still lag behind Western Europe in various economic figures.⁴⁶ Economists, politicians and the public refer to the long-term Soviet legacy to explain backwardness. My findings suggest that the short intervention of the Red Army at the end of WWII matters for long-run harm to regions. Regarding the east-west divide in re-unified Germany, a simple back-of-the-envelope calculation suggests that about one third of the shocked demographic trend could be attributed to the period directly after the war.⁴⁷ Despite population effects, the short period of the presence of the Red Army also shifts proxies for local labor productivity by at least 15%—a number that also somewhat matches the figures of the east-west divide within Germany.⁴⁸ Countries exposed to long-run Soviet dominance after WWII would thus also lag behind Western Europe even if the Red Army and the Soviets had completely withdrawn from the region in summer 1945.

References

ABADIE, A. (2021). Using Synthetic Controls: Feasibility, Data Requirements, and Methodological Aspects. *Journal of Economic Literature*, **59** (2), 391–425.

 $^{^{46}}$ For example, Czechoslovakia (now Czechia and Slovakia) was among the richest countries in Europe around 1930. Today, GDP per capita is approximately 50% of the levels in Western Europe and the population is only around 14% larger today than before WWII, while France (+58%) or Italy (+49%) experienced higher population growth. Sources: Maddison-Project (2013), Bolt and van Zanden (2014).

 $^{^{47}}$ My RD estimates suggest a relative population decline in regions seized by the Red Army of about 22 to 25%. The relative shrinkage of East Germany is about 75% (see Figure 1); thus: 25/75 is about 1/3.

 $^{^{48}}$ GDP per capita (per worker) in East Germany in 2015 was around 70% (80%) of the West German level (VGR, 2016). All these figures do not show any further convergence during the last two decades.

- -, DIAMOND, A. and HAINMUELLER, J. (2015). Comparative Politics and the Synthetic Control Method. *American Journal of Political Science*, **59** (2), 495–510.
- and GARDEAZABAL, J. (2003). The Economic Costs of Conflict: A Case Study of the Basque Country. *American Economic Review*, **93** (1), 113–132.
- ACEMOGLU, D., HASSAN, T. A. and ROBINSON, J. A. (2011). Social Structure and Development: A Legacy of the Holocaust in Russia. *The Quarterly Journal of Economics*, 126 (2), 895–946.
- AGHION, P., JARAVEL, X., PERSSON, T. and ROUZET, D. (2018). Education and Military Rivalry. Journal of the European Economic Association, 17 (2), 376–412.
- AKBULUT-YUKSEL, M. and YUKSEL, M. (2015). The Long-Term Direct and External Effects of Jewish Expulsions in Nazi Germany. American Economic Journal: Economic Policy, 7 (3), 58–85.
- AKRESH, R., BHALOTRA, S., LEONE, M. and OSILI, U. O. (2012). War and Stature: Growing Up during the Nigerian Civil War. *American Economic Review*, **102** (3), 273–277.
- ALESINA, A. and FUCHS-SCHÜNDELN, N. (2007). Goodbye Lenin (or Not?): The Effect of Communism on People's Preferences. *American Economic Review*, **97** (4), 1507–1528.
- ALLEN, R. C. (1998). Capital Accumulation, the Soft Budget Constraint and Soviet Industrialisation. *European Review of Economic History*, **2** (1), 1–24.
- ANGRIST, J. and PISCHKE, J.-S. (2009). Mostly Harmless Econometrics: An Empiricist's Companion, Oxford and Princeton: Princeton University Press.
- ARBATLI, C. E. and GOKMEN, G. (2022). Human Capital Transfers and sub-national Development: Armenian and Greek Legacy in post-expulsion Turkey. *Journal of Economic Growth*, 28, 1–43.
- BARRO, R. J. and SALA-I MARTIN, X. (1991). Convergence across States and Regions. Brookings Papers on Economic Activity, **22** (1), 107–182.
- and (1992). Convergence. Journal of Political Economy, 100 (2), 223–251.
- BARSBAI, T., LICUANAN, V., STEINMAYR, A., TIONGSON, E. and YANG, D. (2020). Information and the Acquisition of Social Network Connections. Working Paper 27346, National Bureau of Economic Research.
- BAUER, M., BLATTMAN, C., CHYTILOVÁ, J., HENRICH, J., MIGUEL, E. and MITTS, T. (2016). Can War Foster Cooperation? *Journal of Economic Perspectives*, **30** (3), 249–274.
- BECKER, S. O. (2022). Forced Displacement in History: Some Recent Research. Australian Economic History Review, 62 (1), 2–25.
- —, BOECKH, K., HAINZ, C. and WOESSMANN, L. (2015). The Empire is Dead, Long Live the Empire! Long-Run Persistence of Trust and Corruption in the Bureaucracy. *The Economic Journal*, **126** (590), 40–74.
- and FERRARA, A. (2019). Consequences of Forced Migration: A Survey of Recent Findings. Labour Economics, 59, 1–16.

- BECKER, T., EICHENGREEN, B., GORODNICHENKO, Y., GURIEV, S., JOHNSON, S., MYLOVANOV, T., ROGOFF, K. and WEDER DI MAURO, B. (2022). A Blueprint for the Reconstruction of Ukraine. London: CEPR Press.
- BEDARD, K. and DESCHÊNES, O. (2006). The Long-Term Impact of Military Service on Health: Evidence from World War II and Korean War Veterans. American Economic Review, 96 (1), 176–194.
- BEER, S. (1991). Vertrauliche Berichte des Amerikanischen Geheimdienstes, Linz: Oberösterreichisches Landesarchiv, p. 177–225.
- (2004). Das sowjetische "Intermezzo": Die "Russenzeit" in der Steiermark. 8. Mai bis 23. Juli 1945, Graz: Selbstverlag der Historischen Landeskommission für Steiermark, pp. 35–59.
- BEKES, C., BORHI, L., RUGGENTHALER, P. and TRASCA, O. (2015). Soviet Occupation of Romania, Hungary, and Austria 1944/45-1948/49. Budapest-New York: Central European University.
- BELLOWS, J. and MIGUEL, E. (2009). War and Local Collective Action in Sierra Leone. Journal of Public Economics, 93 (11), 1144–1157.
- BESLEY, T. and REYNAL-QUEROL, M. (2014). The Legacy of Historical Conflict: Evidence from Africa. *American Political Science Review*, **108** (2), 319–336.
- BISCHOF, G. (1999). Austria in the First Cold War, 1945-55: The Leverage of the Weak. Basingstoke: Palgrave Macmillan.
- BLATTMAN, C. and MIGUEL, E. (2010). Civil War. Journal of Economic Literature, 48 (1), 3–57.
- BLEAKLEY, H. and LIN, J. (2012). Portage and Path Dependence. *The Quarterly Journal* of *Economics*, **127** (2), 587–644.
- BOLT, J. and VAN ZANDEN, J. L. (2014). The Maddison Project: Collaborative Research on Historical National Accounts. *The Economic History Review*, **67** (3), 627–651.
- BONDAR, M. and FUCHS-SCHÜNDELN, N. (2023). Good Bye Lenin Revisited: East-West Preferences Three Decades after German Reunification. *German Economic Review*, 24 (1), 97–119.
- BRAKMAN, S., GARRETSEN, H. and SCHRAMM, M. (2004). The Strategic Bombing of German Cities during World War II and its Impact on City Growth. *Journal of Economic Geography*, 4 (2), 201–218.
- BRAUN, S. and KVASNICKA, M. (2014). Immigration and Structural Change: Evidence from post-war Germany. *Journal of International Economics*, **93** (2), 253–269.
- and OMAR MAHMOUD, T. (2014). The Employment Effects of Immigration: Evidence from the Mass Arrival of German Expellees in Postwar Germany. The Journal of Economic History, 74 (1), 69–108.
- BRETTNER, F. (2013). Die letzten Kämpfe des II. Weltkrieges im Grenzgebiet Niederösterreich-Steiermark-Burgenland. Berndorf: KRAL-Verlag.
- BROSIG-KOCH, J., HELBACH, C., OCKENFELS, A. and WEIMANN, J. (2011). Still Different after all these Years: Solidarity Behavior in East and West Germany. *Journal* of Public Economics, 95 (11), 1373–1376.

- BUKOWSKI, P. (2019). How History Matters for Student Performance: Lessons from the Partitions of Poland. *Journal of Comparative Economics*, **47** (1), 136–175.
- CALONICO, S., CATTANEO, M. D., FARRELL, M. H. and TITIUNIK, R. (2017). rdrobust: Software for Regression Discontinuity Designs. *Stata Journal*, **17** (2), 372–404.
- —, and TITIUNIK, R. (2014a). Robust Data-Driven Inference in the Regression-Discontinuity Design. *The Stata Journal*, **14** (4), 909–946.
- —, and (2014b). Robust Nonparametric Confidence Intervals for Regression-Discontinuity Designs. *Econometrica*, **82** (6), 2295–2326.
- CATTANEO, M. D., TITIUNIK, R. and VAZQUEZ-BARE, G. (2020). Analysis of Regression-Discontinuity Designs with Multiple Cutoffs or Multiple Scores. *The Stata Journal*, 20 (4), 866–891.
- CHANEY, E. and HORNBECK, R. (2016). Economic Dynamics in the Malthusian Era: Evidence from the 1609 Spanish Expulsion of the Moriscos. *The Economic Journal*, **126** (594), 1404–1440.
- CHEREMUKHIN, A., GOLOSOV, M., GURIEV, S. and TSYVINSKI, A. (2013). Was Stalin Necessary for Russia's Economic Development? Working Paper 19425, National Bureau of Economic Research.
- —, —, and (2016). The Industrialization and Economic Development of Russia through the Lens of a Neoclassical Growth Model. *The Review of Economic Studies*, **84** (2), 613–649.
- COLELLA, F., LALIVE, R., SAKALLI, S. O. and THOENIG, M. (2019). *Inference with Arbitrary Clustering*. IZA Discussion Paper 12584, Bonn.
- CONLEY, T. G. (1999). GMM Estimation with Cross Sectional Dependence. *Journal of Econometrics*, **92** (1), 1–45.
- (2008). Spatial Econometrics, Basingstoke: Palgrave Macmillan.
- (2010). Spatial Econometrics, London: Palgrave Macmillan UK, pp. 303–313.
- DALGAARD, C.-J., KAARSEN, N., OLSSON, O. and SELAYA, P. (2022). Roman Roads to Prosperity: Persistence and non-persistence of Public Infrastructure. *Journal of Comparative Economics*, **50** (4), 896–916.
- DAVIS, D. R. and DINGEL, J. I. (2019). A Spatial Knowledge Economy. *American Economic Review*, **109** (1), 153–170.
- and WEINSTEIN, D. E. (2002). Bones, Bombs, and Break Points: The Geography of Economic Activity. *American Economic Review*, **92** (5), 1269–1289.
- and (2008). A Search For Multiple Equilibria in Urban Industrial Structure. Journal of Regional Science, 48, 29–65.
- DELL, M. (2010). The Persistent Effects of Peru's Mining Mita. *Econometrica*, **78** (6), 1863–1903.
- and QUERUBIN, P. (2017). Nation Building Through Foreign Intervention: Evidence from Discontinuities in Military Strategies. The Quarterly Journal of Economics, 133 (2), 701–764.

- DUBE, A., LESTER, T. W. and REICH, M. (2010). Minimum Wage Effects Across State Borders: Estimates Using Contiguous Counties. The Review of Economics and Statistics, 92 (4), 945–964.
- EBERHART, H. (1995). Wideraufbau und Nachkriegsalltag: Das Tagebuch von Anton Pirchegger, Graz: Selbstverlag der Historischen Landeskommission für Steiermark, pp. 361–387.
- ECONPOL-FORUM (2023). How to Reconstruct Ukraine? Challenges, Plans and the Role of the EU. Munich: CESifo GmbH.
- EDER, C. and HALLA, M. (2016). The Long-Lasting Shadow of the Allied Occupation of Austria on its Spatial Equilibrium. IZA Discussion Paper 10095, Institute for the Study of Labor, Bonn.
- —, and HILMBAUER-HOFMARCHER, P. (2023). On the Origin and Composition of the German East-West Population Gap. Working Paper No. 1817, Johannes Kepler University of Linz.
- EHRLICH, M. V. and SEIDEL, T. (2018). The Persistent Effects of Place-Based Policy: Evidence from the West-German Zonenrandgebiet. American Economic Journal: Economic Policy, 10 (4), 344–374.
- EICHENGREEN, B. (2023). Economic Recovery in Post-World War II West Germany and Ukraine Today. *EconPol-Forum*, **24** (2), 30–35.
- ELLISON, G. and GLAESER, E. L. (1999). The Geographic Concentration of Industry: Does Natural Advantage Explain Agglomeration? *American Economic Review*, **89** (2), 311–316.
- —, and KERR, W. R. (2010). What Causes Industry Agglomeration? Evidence from Coagglomeration Patterns. *American Economic Review*, **100** (3), 1195–1213.
- ERICKSON, E. L. (1950). The Zoning of Austria. The Annals of the American Academy of Political and Social Science, 267, 106–113.
- EUGSTER, B., LALIVE, R., STEINHAUER, A. and ZWEIMÜLLER, J. (2011). The Demand for Social Insurance: Does Culture Matter? *The Economic Journal*, **121** (556), F413– F448.
- FEIGENBAUM, J., LEE, J. and MEZZANOTTI, F. (2022). Capital Destruction and Economic Growth: The Effects of Sherman's March, 1850–1920. American Economic Journal: Applied Economics, 14 (4), 301–342.
- FUCHS-SCHÜNDELN, N. and MASELLA, P. (2016). Long-Lasting Effects of Socialist Education. *The Review of Economics and Statistics*, **98** (3), 428–441.
- GABAIX, X. (1999). Zipf's Law and the Growth of Cities. *American Economic Review*, **89** (2), 129–132.
- GALOFRÉ-VILÀ, G., MEISSNER, C. M., MCKEE, M. and STUCKLER, D. (2021). Austerity and the Rise of the Nazi Party. *The Journal of Economic History*, **81** (1), 81–113.
- GENNAIOLI, N., LA PORTA, R., DE SILANES, F. L. and SHLEIFER, A. (2014). Growth in Regions. *Journal of Economic Growth*, **19** (3), 259–309.
- —, —, LOPEZ-DE SILANES, F. and SHLEIFER, A. (2012). Human Capital and Regional Development. *The Quarterly Journal of Economics*, **128** (1), 105–164.

- GROSFELD, I. and ZHURAVSKAYA, E. (2015). Cultural vs. Economic Legacies of Empires: Evidence from the Partition of Poland. *Journal of Comparative Economics*, **43** (1), 55–75.
- GROSSMANN, J., JURAJDA, and ROESEL, F. (forthcoming). Forced Migration, Staying Minorities, and New Societies: Evidence from Postwar Czechoslovakia. *American Journal* of Political Science.
- HARRISON, M. (2003). Post-War Russian Economic Growth: Not a Riddle. Europe-Asia Studies, 55 (8), 1323–1329.
- (2011). Forging Success: Soviet Managers and Accounting Fraud, 1943–1962. Journal of Comparative Economics, 39 (1), 43–64.
- and KIM, B.-Y. (2006). Plans, Prices, and Corruption: The Soviet Firm Under Partial Centralization, 1930 to 1990. The Journal of Economic History, 66 (1), 1–41.
- HENDERSON, J. V. (1974). The Sizes and Types of Cities. The American Economic Review, 64 (4), 640–656.
- HOFBAUER, H. (1992). Westwärts: Osterreichs Wirtschaft im Wiederaufbau. Vienna: VG-Verlag für Gesellschaftskritik.
- HOLMES, T. (1998). The Effect of State Policies on the Location of Manufacturing: Evidence from State Borders. *Journal of Political Economy*, **106** (4), 667–705.
- HORNUNG, E. (2014). Immigration and the Diffusion of Technology: The Huguenot Diaspora in Prussia. American Economic Review, **104** (1), 84–122.
- HSIANG, S. M. (2010). Temperatures and Cyclones Strongly Associated with Economic Production in the Caribbean and Central America. *Proceedings of the National Academy* of Sciences, **107** (35), 15367–15372.
- HUBER, K., LINDENTHAL, V. and WALDINGER, F. (2021). Discrimination, Managers, and Firm Performance: Evidence from "Aryanizations" in Nazi Germany. *Journal of Political Economy*, **129** (9), 2455–2503.
- IBER, W. M., KNOLL, H., KARNER, S., PICKL, O., PRUEGGER, H., RUGGENTHALER, P., WONISCH, A. and STERN, S. (2008). Die Rote Armee in der Steiermark - Sowjetische Besatzung 1945. Graz: Leykam Buchverlag.
- IMBENS, G. W. and LEMIEUX, T. (2008). Regression Discontinuity Designs: A Guide to Practice. *Journal of Econometrics*, **142** (2), 615–635.
- KARNER, S. (2000). *Die Steiermark im 20. Jahrhundert*. Graz Vienna Cologne: Politik-Wirtschaft-Gesellschaft-Kultur.
- KEELE, L. J. and TITIUNIK, R. (2015). Geographic Boundaries as Regression Discontinuities. *Political Analysis*, 23 (1), 127–155.
- KELLY, M. (2019). The Standard Errors of Persistence. CEPR Discussion Paper 13783, CEPR, London.
- KLINE, P. and MORETTI, E. (2014). Local Economic Development, Agglomeration Economies, and the Big Push: 100 Years of Evidence from the Tennessee Valley Authority. *The Quarterly Journal of Economics*, **129** (1), 275–331.

- KNIRSCH, D. and NIEMANN, R. (2005). Die Abschaffung der österreichischen Gewerbesteuer als Vorbild für eine Reform der kommunalen Steuern in Deutschland? Working Paper. arque Discussion Papers in Quantitative Tax Research.
- KRUGMAN, P. (1991). Increasing Returns and Economic Geography. Journal of Political Economy, 99 (3), 483–499.
- KÖHLER, H. (1965). Economic Integration in the Soviet Bloc: With an East German Case Study. New York: Frederick A. Praeger.
- LEE, D. S. and LEMIEUX, T. (2010). Regression Discontinuity Designs in Economics. Journal of Economic Literature, 48 (2), 281–355.
- LEÓN, G. (2012). Civil Conflict and Human Capital Accumulation: The Long-term Effects of Political Violence in Perú. *The Journal of Human Resources*, **47** (4), 991–1022.
- LIBERMAN, P. (1996). Does Conquest Pay? The Exploitation of Occupied Industrial Societies. Princeton: Princeton University Press.
- LICHTER, A., LÖFFLER, M. and SIEGLOCH, S. (2020). The Long-Term Costs of Government Surveillance: Insights from Stasi Spying in East Germany. *Journal of the European Economic Association*, **19** (2), 741–789.
- LIST, J. A., SHAIKH, A. M. and XU, Y. (2019). Multiple Hypothesis Testing in Experimental Economics. *Experimental Economics*, **22** (4), 773–793.
- MADDISON-PROJECT (2013). http://www.ggdc.net/maddison/maddison-project/home.htm. Groningen: The Groningen Growth and Development Centre, University of Groningen.
- MARTINEZ, L. R., JESSEN, J. and XU, G. (2023). A Glimpse of Freedom: Allied Occupation and Political Resistance in East Germany. *American Economic Journal: Applied Economics*, **15** (1), 68–106.
- MASCHER-PICHLER, H. A. (2009). Baden bei Wien zur sowjetischen Besatzungszeit 1945–1955 mit besonderer Berücksichtigung der ersten beiden Besatzungsjahre und des Jahres 1955. Vienna: PhD Thesis at the University of Vienna.
- MIGUEL, E. and ROLAND, G. (2011). The long-run Impact of Bombing Vietnam. Journal of Development Economics, 96 (1), 1–15.
- MITTERER, K. and SEISENBACHER, M. (2020). Fact Sheets: Pflichtschule und Tagesbetreuung - Grundlagen und Finanzierung. Bericht, KDZ Zentrum für Verwaltungsforschung.
- MOSER, P., VOENA, A. and WALDINGER, F. (2014). German Jewish Émigrés and US Invention. *American Economic Review*, **104** (10), 3222–3255.
- MUNROE, E., NOSACH, A., PEDROZO, M., GUARNIERI, E., RIANO, J., TUR-PRATS, A. and VALENCIA CAICEDO, F. (2023). *The Legacies of War for Ukraine*. Discussion Paper No. 17841, CEPR Press.
- MURPHY, K. M., SHLEIFER, A. and VISHNY, R. W. (1989). Industrialization and the Big Push. *Journal of Political Economy*, **97** (5), 1003–1026.
- NAGY, D. K. (2022). Trade and Urbanization: Evidence from Hungary. American Economic Journal: Microeconomics, 14 (3), 733–790.
- OCHSNER, C. (2017). Dismantled once, Diverged forever? A quasi-natural Experiment of Red Army's Misdeeds in post-WWII Europe. Working Paper No. 240, ifo Working Paper.

- OCKENFELS, A. and WEIMANN, J. (1999). Types and Patterns: An Experimental East-West-German Comparison of Cooperation and Solidarity. *Journal of Public Economics*, 71 (2), 275–287.
- OFER, G. (1987). Soviet Economic Growth: 1928-1985. Journal of Economic Literature, 25 (4), 1767–1833.
- OTO-PERALÍAS, D. and ROMERO-ÁVILA, D. (2016). Historical Frontiers and the Rise of Inequality: The Case of the Frontier of Granada. *Journal of the European Economic Association*, **15** (1), 54–98.
- PASCALI, L. (2016). Banks and Development: Jewish Communities in the Italian Renaissance and Current Economic Performance. The Review of Economics and Statistics, 98 (1), 140–158.
- PETERS, M. (2022). Market Size and Spatial Growth—Evidence From Germany's Post-War Population Expulsions. *Econometrica*, **90** (5), 2357–2396.
- PICKL, O. (1995). Das Kriegsende 1945 und die frühe Besatzungszeit im mittleren Mürztal: Ein Beitrag zur Vergangenheitsbewältigung nach zeitgenössischen Berichten, Graz: Selbstverlag der Historischen Landeskommission für Steiermark, pp. 277–295.
- PIRCHEGGER, H. (1996). Geschichte der Steiermark. Graz: Leykam Verlag.
- RADSPIELER, T. (1955). The Ethnic German Refugee in Austria 1945 to 1955. The Hague: Martinus Nijhoff.
- RATHKOLB, O. (1985). Gesellschaft und Politik am Beginn der zweiten Republik. Vertrauliche Berichte der US-Militäradministration aus Österreich 1945 in englischer Originalfassung. Vienna: Hermann Böhlas Nachf.
- REDDING, S. J. and STURM, D. M. (2008). The Costs of Remoteness: Evidence from German Division and Reunification. *American Economic Review*, **98** (5), 1766–1797.
- —, and WOLF, N. (2011). History and Industry Location: Evidence from German Airports. *The Review of Economics and Statistics*, **93** (3), 814–831.
- RIHS, S. (2021). Die Bezirkshauptmannschaft. Grundlagen und Zukunftspotenzial einer unterschätzten Behörde im österreichischen Föderalismus. Verlag Österreich.
- ROHNER, D. and THOENIG, M. (2021). The Elusive Peace Dividend of Development Policy: From War Traps to Macro Complementarities. Annual Review of Economics, 13 (1), 111–131.
- —, and ZILIBOTTI, F. (2013a). Seeds of Distrust: Conflict in Uganda. Journal of Economic Growth, 18 (3), 217–252.
- —, and (2013b). War Signals: A Theory of Trade, Trust, and Conflict. *The Review of Economic Studies*, **80** (3), 1114–1147.
- ROSENSTEIN-RODAN, P. N. (1943). Problems of Industrialisation of Eastern and South-Eastern Europe. *The Economic Journal*, **53** (210/211), 202–211.
- SCHUMANN, A. (2014). Persistence of Population Shocks: Evidence from the Occupation of West Germany after World War II. American Economic Journal: Applied Economics, 6 (3), 189–205.

- SEMRAD, A. (2015). Immigration and Educational Spillovers: Evidence from Sudeten German Expellees in Post-War Bavaria. Discussion paper, University of Munich.
- SIMON, H. A. (1955). On a Class of Skew Distribution Functions. *Biometrika*, **42** (3/4), 425–440.
- SLAPNICKA, H. (1986). Oberösterreich-Zweigeteiltes Land: 1945-1955. Linz: Oberösterreichischer Landesverlag.
- SOLOW, R. M. (1956). A Contribution to the Theory of Economic Growth. *The Quarterly Journal of Economics*, **70** (1), 65–94.
- STATISTIKAUSTRIA (2014). Statistik der Einbürgerungen, 2014. Vienna.
- STELZL-MARX, B. (2012). Stalin Soldaten in Osterreich: Die Innensicht der sowjetischen Besatzung 1945-1955. Munich-Vienna: Böhlau Verlag & Oldenbourg Verlag.
- STIEBER, G. (1995). Volksdeutsche und Displaced Persons, Vienna: Dachs Verlag GmbH., pp. 140–156.
- STOLPER, W. F. (1960). The Structure of the East German Economy. Cambridge MA: Harvard University Press.
- TAYLOR, Z. (2008). The Dismantling and Removal of Railway Lines by Soviet Red Army Troops on Present-Day Polish Territory, 1944–1948. *Journal of Transport Geography*, 16, 217–228.
- TESTA, P. A. (2020). The Economic Legacy of Expulsion: Lessons from Post-War Czechoslovakia. *The Economic Journal*, **131** (637), 2233–2271.
- ULRICH, J. (1978). Der Luftkrieg über Österreich: Chronologischer Ablauf, Aktive und Passive Schutzmaßnahmen, Auswirkungen, Statistiken, 2nd ed. Vienna: Österreichischer Bundesverlag.
- VGR (2016). Bruttoinlandsprodukt, Bruttowertschöpfung in den Ländern der Bundesrepublik Deutschland 1991 bis 2015. Stuttgart: Arbeitskreis Volkswirtschaftliche Gesamtrechnungen der Länder.
- VONYÓ, T. (2017). War and Socialism: Why Eastern Europe Fell Behind between 1950 and 1989. *The Economic History Review*, **70** (1), 248–274.
- (2020). The Economic Consequences of the War West: Germany's Growth Miracle after 1945. Cambridge, UK: Cambridge Studies in Economic History.
- WAHL, F. (2017). Does European Development have Roman Roots? Evidence from the German Limes. *Journal of Economic Growth*, **22** (3), 313–349.
- WALDINGER, F. (2010). Quality Matters: The Expulsion of Professors and the Consequences for PhD Student Outcomes in Nazi Germany. *Journal of Political Economy*, 118 (4), 787–831.
- (2016). Bombs, Brains, and Science: The Role of Human and Physical Capital for the Creation of Scientific Knowledge. The Review of Economics and Statistics, 98 (5), 811–831.
- YAMADA, T. and YAMADA, H. (2021). The long-term Causal Effect of U.S. Bombing Missions on Economic Development: Evidence from the Ho Chi Minh Trail and Xieng Khouang Province in Lao P.D.R. Journal of Development Economics, 150, 102611.

Online Appendix

Hostility, population sorting, and backwardness

Quasi-experimental evidence from the Red Army after WWII

Christian Ochsner

CERGE-EI Prague

Contents

Α	Detailed historical background									
	A.1 Last combat operations in Styria	4								
	A.2 Local resistance in Styria	5								
	A.3 The line of contact	6								
в	Data sources and definition	10								
С	C Additional figures and tables									
	C.1 Additional figures	14								
	C.2 Additional tables	26								

List of figures in the Online Appendix

Figure A.1	The line of contact—Land battles, checkpoints, rivers and regions \ldots .	5
Figure A.2	Transcription of a Soviet report on checkpoints along the line of contact	7
Figure C.3	Occupation zone proposals, liberation and final occupation \ldots \ldots \ldots	14
Figure C.4	Population changes during and after WWII (District level)	15
Figure C.5	Population per occupation zone and Vienna, 1869 to 2011 \ldots	16
Figure C.6	Population growth from 1939 to 1946 in entire Styria (in %) \ldots .	17
Figure C.7	Coefficient plots of different RD estimates w.r.t. 1939, 1869 to 2011 $$. $$.	18
Figure C.8	RD plots on local firm tax revenues per employee $\ldots \ldots \ldots \ldots$	19
Figure C.9	Effects on the metropolitan area of Graz with alternative matching $\ . \ .$	20
Figure C.10	Illustration of RD samples for heterogeneous effects	21
Figure C.11	Population development from 1869 to 2011—Western Allies versus Red	
Army	$(1939=100) \ldots $	22
Figure C.12	Illustration of regional shocks and regional policies in Styria	23
Figure C.13	Illustration of RD samples along the line of contact	24
Figure C.14	Illustration of RD samples with pseudo-lines of contact	25

List of tables in the Online Appendix

Table	C.2	Geographic determinants of population growth from 1939 to 1946	26
Table	C.3	Summary statistics	27
Table	C.4	Functional forms of RD polynomials	28
Table	C.5	RD sample selection	29
Table	C.6	Unconditional continuity of covariates across the line of contact	30
Table	C.7	Population dynamics across the line of contact from census to census	31
Table	C.8	Measures of local labor productivity across the line of contact in 2011 $$.	32
Table	C.9	Measures of Labor productivity in other years	33
Table	C.10	Donor pool cities for the synthetic control method $\ldots \ldots \ldots \ldots \ldots$	34
Table	C.11	Heterogeneous effects and the entire UK zone	35
Table	C.12	Effects of regional shocks and regional policies on population dynamics .	36
Table	C.13	Effects of regional shocks and regional policies on local tax revenues	37
Table	C.14	Alternative RD polynomials	38
Table	C.15	Different bandwidths	39
Table	C.16	Optimal data-driven bandwidth selection procedure (rdrobust) \ldots .	40
Table	C.17	Standard errors with different spatial cut-offs (RD)	41
Table	C.18	P-Values adjusted for multiple hypothesis testing for variables in 2011 $$.	42
Table	C.19	Soviet proposal and pseudo lines of contact	43
Table	C.20	Differences-in-difference estimates with population dynamics \ldots .	44
Table	C.21	Births and deaths by year of census $\ldots \ldots \ldots$	45
Table	C.22	Municipal covariates in 2011	46
Table	C.23	Machinery in agriculture (contd.)	47

Appendix A: Detailed historical background

This Appendix provides further details on the historical background in Styria during and in the direct aftermath of World War Two (WWII). I discuss the situation in the last weeks of WWII (last combat operation and local resistance), and outline the formation of the line of contact between the Red Army and the Western Allies in the three subregions of Styria (Enns Valley, Mur Valley, Graz region).

The aims of this appendix are (i) to provide a more detailed picture of the period of study and (ii) to provide a detailed discussion of the quasi-random exposure to the Red Army for some regions of Styria but not for others. The outline of Appendix A is thus motivated to discuss the crucial assumptions of a regression discontinuity (RD) design to establish a causal relationship between Red-Army presence and subsequent economic backwardness.

A.1 Last combat operations in Styria

Most parts of Styria were under (Nazi-)German control until the German surrender on May 08, 1945. The later line of contact, i.e., the RD threshold of this study, was thus far away from any battleground and combat operation during WWII. In March 1945, the Red Army reached the eastern border of Austria. The main target at this late period of WWII was the Austrian capital city Vienna. In South Austria, the front line between the (Nazi-)German Wehrmacht and the Red Army was in the very east of Styria and thus around 75 to 100 kilometers east of the later line of contact (Iber *et al.*, 2008; Stelzl-Marx, 2012). Figure A.1 shows the Styrian regions in the very east exposed to combat operations from April until early May 1945 (red dashed areas). Almost all land battles in Styria took place in the very east of Styria along the border between the federal states of Styria and Burgenland. The front line during the last 1.5 months of war operations was more or less stable mainly because the Red Army forces were diverted for the fight in Vienna and did thus not have major achievements in South Austria (Stelzl-Marx, 2012).

The land battles in the very east of Styria started in late March/early April 1945 and continued until early May 1945. The German Wehrmacht had already began to withdraw from eastern Styria on May 7 following Karl Dönitz's (Hitler's successor) command to attempt to surrender to the Western Allies rather than the Soviets (Brettner, 2013). After the ceasefire agreement on May 8, 1945, the last German troops left the front line and also tried to escaped over the Mürz Valley and Mur Valley towards the West. The last defensive battle of some Wehrmacht troops took place in the night of May 8–9 in the region of "Schanz/Schanzsattel" (ca. 60 kilometers east of the later line of contact, see Figure A.1) to secure the withdrawal of (Nazi)German troops from eastern Styria towards the Mürz Valley in the west (Pickl, 1995). The Wehrmacht was only partially successful in fleeing from the Red Army. Nevertheless, what happened was an exodus of the German Wehrmacht from eastern Styria toward the US troops arriving in the northwest. All related literature (Karner, 2000; Beer, 2004; Iber *et al.*, 2008; Stelzl-Marx, 2012; Brettner, 2013) do not report any further combat operations among the escaping Wehrmacht and the Western Allies. With the exception of the last defensive battle at Schanzsattel, there was thus no strategic behavior of the German Wehrmacht that had any effect on the line of contact. To sum up, land battles in Styria took place in the very east (75 to 100 kilometers away from the later line of contact), and also the last defensive operation at Schanzsattel was around 60 kilometers east of the later line of contact. This setting thus ensures that postwar economic differences across the line of contact are not caused by any (land) battles.

Figure A.1: The line of contact—Land battles, checkpoints, rivers and regions



Notes: The map shows the regions of Styria exposed to land battles between the German Wehrmacht and the Red Army from late March/early April 1945 until May 7, 1945, the location of the last and westernmost land battle in Styria, and the points where the Red Army met the Western Allies (US and UK troops). The map also indicates the main geographical regions and mountain ranges in Styria. Sources: See text. Own illustration.

A.2 Local resistance in Styria

Evidence of local resistance that might have influenced the line of contact is also very limited. Generally, the people of Styria favored liberation by the Western Allies rather than the Red Army but there was no active resistance against the advancing Red Army. There is, however, an anecdote from the city of Murau (around 30 kilometers west of the

line of contact, and thus outside the main RD sample). The citizens wanted to avoid capture by the Red Army. They flagged their houses with the Union Jack in the hope that, if the Red Army arrived before the UK troops, they would believe that the city has already been liberated by the UK troops (Stelzl-Marx, 2012). However, Murau was actually captured by UK troops first anyway. Thus, this anecdote of local resistance did not influence the line of contact. The related literature is otherwise silent about examples of local resistance, indicating that the line of contact was not determined by endogenous behavior of Styrian people. The line of contact is, given the absence of any evidence, not a result of any local resistance.

A.3 The line of contact

The capture of Styria did not follow any liberation plan and was not coordinated by the Allies (Karner, 2000; Iber *et al.*, 2008; Stelzl-Marx, 2012). The liberation of Styria can be described as rather chaotic. For example, the city of Leibnitz (35 kilometers south of Graz, and thus outside the RD sample) was triple occupied by the "Eastern Allies" on the same day by troops from the Red Army and Bulgaria and Tito partisans (all of them somewhat under the coordination of the Red Army). This indicates that the liberation did not follow any military strategy and military coordination, not even among the "Eastern Allies" (Karner, 2000).

Confusion and a lack of Allied coordination also dominated in the rest of Styria where the Western Allies and the Red Army tried to capture as much land as possible (Karner, 2000; Stelzl-Marx, 2012). The situation on May 8 and 9, 1945 was somewhat different in the three main regions of Styria where the Allies met (Graz region, Mur Valley, Enns Valley; see Figure A.1). I describe below the random points of contact on May 9 by the Red Army and UK troops (Graz region, Mur Valley) and between the Red Army and the US troops (Enns Valley) respectively. The initial points of contact by the Allies in various parts of Styria was quasi-random, i.e., there were no agreements about the time and place of the initial contact, but some arbitrary decisions were made about how to divide Styria outside these initial points of contact. I will discuss them below.

The demarcation line between the Red Army and the Western Allies was in place until the start of the withdrawal of the Red Army from Styria on July 22, 1945. Figure A.2 shows a transcription of a Red Army report on the location of checkpoints (Iber *et al.*, 2008, p. 298). This report states the exact location of Red Army check points in Styria and along the Styrian-Carinthian state border as of June 1, 1945. I mainly use Iber *et al.* (2008) to assign municipalities to their respective liberation force. I cross-checked the unclear liberation treatment of Kleinlobming, Lassing, Modriach, Salla and St. Georgen ob Judenburg with the respective municipality chronicle.⁴⁹ Nevertheless, the line of contact did not necessarily follow (2011) municipality borders. Some farms in the municipalities of Edelschrott and Kleinlobming and the hamlet of Piber in the city of Köflach were divided from their main municipality. I assign the liberation treatment in these cases according to liberation of the respective main locality.

Figure A.2: Transcription of a Soviet report on checkpoints along the line of contact

Die Fühlungnahme zwischen den Truppen der Roten Armee und jenen der Armeen der [West-]Alliierten verlief mit 1. Juni 1945 entlang folgender Linie: Mauthausen (4060), Lauf der Enns bis Döllach (6040), Kote 2116⁴ (5040), Kote 1939 (4040), Mitterhuber⁵, Pusterwald (4050), Kote 1504, St. Georgen⁶ (2060), Lauf der Mur, Lind, Lobming⁷ (2080), Kote 1671, Kote 1555 (2090), Krenhof (1000), Piber, Oberdorf⁸, Kowald⁹ (1000), Edelschrott¹⁰ (0000), Kote 1026 (8000), Kote (9010), Deutschlandsberg (8010), Kote 2141¹¹ (8090), Kote 1521¹² (6000), Lauf der Drau, Dravograd.¹³

Notes: The figure shows the locations of Red Army control and check points based on a transcription of a Soviet report (Iber *et al.*, 2008, p. 298). I use these locations of checkpoints to define the line of contact from May 9 to July 22 in Styria (with cross-checking municipality chronicles).

I describe separately for the three main regions in Styria the formation of the intra-Styrian line of contact that was in place for 74 days. However, along most of the line of contact, places were randomly captured by one army first just before the other army arrived from another direction. Whether a certain municipality was just the last unlucky one to be captured by the Red Army, or the last one to be captured by Western troops was neither foreseeable nor determined by any economic or—except for some sections of rivers and connection roads (see discussion below)—geographical characteristics.

Graz region (Red Army and UK troops)

At the end of WWII, the frontline between the German Wehrmacht and the Red Army was in the very east of Styria and thus around 75 to 100 kilometers east of the later line of contact. Starting in the night of May 7 to May 8, 1945 (i.e., the night before the Nazi-German surrender), the Red Army started to advance toward the west after the German Wehrmacht began to escape in the same direction. The Red Army captured the Styrian capital city of Graz in the night from May 08 to May 09 (Karner, 2000; Stelzl-Marx, 2012). Some southern Styrian regions were captured by Bulgarian and Yugoslavian (Tito partisans) troops first before the Red Army took control there, too (Iber *et al.*, 2008; Stelzl-Marx, 2012).

British troops were long held back in northern Italy. After the ceasefire agreement, they expanded toward the Austrian state of Carinthia and over the Packsattel toward Styria.

 $^{^{49}\}mathrm{I}$ would like to thank Dr. Ernst Reinhold Lasnik for the information about the liberation treatment of the municipality of Salla.

The UK troops reached the city of Voitsberg (Graz region) on May 9, 1945. In the evening of May 9, forerunner troops of the 8th British Army met forerunner troops of the 57th Red Army (Ukrainian Front) near Köflach, whereby "there was a handshake between the generals" (Stelzl-Marx, 2012, p. 144). This place of contact near the municipality of Rosenthal an der Kainach was fully random and not planned. This location then also became the location of a check point among the UK and Red Army troops until July 1945 (see Figure A.2). However, not all localities along the line of contact were direct and initial meeting points of the respective Allies. Thus, the assignment of municipalities to the liberation force outside of these random places of contact was often determined by connection roads (i.e., starting from the random points of contact, the respective Army tried to figure out how to connect these points): The UK troops entered the Graz region via Packsattel and thus captured the localities from Packsattel to Köflach while those places not connected to this road were captured by the Red Army. In the north of Rosenthal an der Kainach, the road over the Reisstrasse that connects the UK liberated part in the Graz region with the UK liberated part in the Mur Valley defines the line of contact. All in all, the point of contact near Rosenthal and er Kainach was fully random. Regions captured by the respective troops on the way to Rosenthal remained under the control of the respective troops. Stelzl-Marx (2012), however, reports some smaller adjustments of the line of contact in the Graz region in the days after May 9. These adjustments did not affect the assignment of the main localities of a municipality, but may have affected some farms, as discussed in one example in the municipality of Edelschrott.

Mur Valley (Red Army and UK troops)

The Mur Valley is the central region along the line of contact (see Figure A.1). The Red Army (parts of the 26th Army of the Ukrainian Front) approached from the east via Loeben towards Knittelfeld and Judenburg. The UK troops reached the Mur Valley over the Obdacher Sattel from Carinthia and later over the Reissstrasse from the Graz Region (see the liberation of the Graz region). Judenburg, where the respective Ally met, was then divided along the River Mur (Karner, 2000; Iber *et al.*, 2008; Stelzl-Marx, 2012). Places in the southwest of Judenburg came under British control. Towards the east, the localities east of the Reisstrasse (the connection road towards the Graz region) was captured by the Red Army. Once again, the main point of contact in the Mur Valley (Judenburg) was neither foreseeable nor planned (Stelzl-Marx, 2012)). The line of contact partially followed the River Mur and was determined by the advances of the UK troops in other parts, i.e., along the Reisstrasse or via the Obdacher Sattel from the south.

Enns Valley (Red Army and US troops)

The northwest of Styria consists of the Enns Valley (see Figure A.1). There, the Red Army approached the River Enns in the municipality of Landl on May 9, 1945, via Bruck an der Mur, Leoben and Eisenerz. Landl was later divided between the US troops and the Red Army. In Landl, the Red Army continued its expansion and followed the River Enns to the West where they met the approaching US troops in the south of Liezen (Stelzl-Marx, 2012). US troops arrived in Styria from the northwest, mainly from Salzburg and Upper Austria, through the Ausseerland and Pyhrnpass. The bridge over the River Enns in Liezen was the point of contact and became an official check point until the withdrawal of the Red Army. The demarcation line in the Enns Valley region thus followed the River Enns from the Upper Austrian border to the municipality of Liezen. Regions in the north of the River Enns and in the west of Liezen were liberated by US troops while the Red Army captured places south of the River Enns and east of the point of contact in the south of Liezen. Stelzl-Marx (2012) reports that some forerunner troops of the Red Army managed to reach the region close to Schladming in the southwest of Liezen, but Schladming was never under Red Army control in Styria—it seems to be the case that the US troops and the Red Army decide about the line of contact at the point where their regular armies met. To sum up, there were some arbitrary decision in the Enns Valley, i.e., by the Red Army whether to follow the River Enns once they reached Landl, or whether to cross the River Enns. I thus control for potential effects of rivers as the demarcation line in the robustness section. However, the main place of contact, the municipality of Liezen, seems to be orthogonal to any military decisions.

Appendix B: Data sources and definition

This section lists the data sources of the paper. Archival data are digitized from hard copies. Self-compiled data are transformed to the 2011 territorial status of Styrian and Carinthian municipalities. I traced mergers of municipalities since 1934 based on "Auflösungen bzw. Vereinigungen von Gemeinden ab 1945" and "Historisches Ortslexikon: Statistische Dokumentation zur Bevölkerungs- und Siedlungsgeschichte der Steiermark" (Part 1 and Part 2). Mergers often combined smaller territorial entities to larger ones. In some exceptions, some smaller localities or single farms were portioned and became part of other territorial units than the former main locality of a municipality. This results in minor differences of the merged archival data with the official population data by Statistik Austria. However, I only face small deviations of less than 1% on average for around 2,5% of all municipalities (mainly in the census of 1939). To avoid any calculation errors, I always compute self-referential socio-economic and socio demographic shares in the censuses of 1934, 1939, 1951 and 1961, i.e., the share of female is the ratio of female/(male + female).

Population data

Population data for 1869 to 1939 and for 1951 to 2011 at the 2011 territorial status of municipalities are retrieved from the statistical database STATcube by *Statistik Austria* (https://statcube.at/statistik.at/). My data contains data for the federal states of Styria and Carinthia as well as for all Austrian municipalities for the SCM analysis.

Population measures for 1946 (at the level of municipalities) and population figures for 1940–1945 (at the level of districts in the German Reich) use the following archival sources (hard copies):

- Die Zivilbevölkerung des Deutschen Reiches 1940—1945: Ergebnisse der Verbrauchergruppen-Statistik. Statistische Berichte Arb.Nr. Vlll/19/1. Wiesbaden: Statistisches Bundesamt. 1953.⁵⁰
- Gemeindeverzeichnis von Österreich: Auf Grund einer besonderen Erhebung aus dem Jahre 1946. Vienna: Österreichischen Statistischen Zentralamt. 1948.

The archival data comprises the number of local residents eligible for food vouchers during (until January 1945) or in the direct aftermath of WWII (1946). These food voucher data provide the best figures for local population for 1940 to 1946.

⁵⁰Data are based on food voucher data from Nazi Germany and use the territorial status of Austria according to the (Nazi) German Statistical Office. The number of districts is smaller than in the Austrian censuses due to some district mergers in the state of Burgenland and the inclusion of some district free cities to their surrounding area.

Population data separated for West and East Germany base on the following study:

 Roesel, Felix (2019): Die Wucht der deutschen Teilung wird völlig unterschätzt, ifo Dresden berichtet 3 (2019), pp. 23-25.

I am grateful to Felix Roesel to provide the underlaying data for my study.

Socio-economic and socio-demographic variables

Socio-economic and socio-demographic variables of municipality populations for 1934, 1939, 1951 and 1961 are based on the following archival sources (hard copies):

- Die Ergebnisse der österreichischen Volkszählung vom 22. März 1934. Issue 5.
 Vienna: Österreichisches Bundesamt für Statistik. 1935.
- Ergebnisse der Volks-, Berufs- und landwirtschaftlichen Betriebszählung vom 17. Mai 1939: Alpen- und Donau-Reichsgaue. Volume 559, Issue 13, Berlin/Vienna: Statistik des Deutschen Reichs. 1943.
- Ergebnisse der Volkszählung vom 1. Juni 1951 nach Gemeinden. Beiträge zur österreichischen Statistik. Issue 11, Vienna: Österreichischen Statistischen Zentralamt. 1953.
- Ergebnisse der Volkszählung vom 21. März 1961. Beiträge zur österreichischen Statistik. Various Issues (Styria: Issue 9), Vienna: Österreichischen Statistischen Zentralamt. 1964.

Data for 1971 to 2001 are based on decennial censuses, and for 2011 on register data. Data for 1971 and thereafter are retrieved from the statistical database STATcube by *Statistik Austria* (https://statcube.at/statistik.at/).

Socio-economic variables comprise information on residents' occupation, occupation status, unemployment and self-employment. Data until 2001 mainly comprise counts according to the head of the family. Data for 2011 contains individual measures. Socio-demographic variables comprise information on gender, age cohorts, number of housing units, commuters, and highest achieved education level (some data only available for specific years).

Natural population growth

Data on births, deaths and marriages at the level of municipalities for 1951 to 1971 use the following archival sources (hard copies):

 Die natürliche Bevölkerungsbewegung im Jahre 1951 nach Gemeinden. Beiträge zur österreichischen Statistik. Issue 9, Vienna: Österreichischen Statistischen Zentralamt. 1953.

- Die natürliche Bevölkerungsbewegung im Jahre 1960. Beiträge zur österreichischen Statistik. Issue 65, Vienna: Österreichischen Statistischen Zentralamt. 1961.⁵¹
- Die natürliche Bevölkerungsbewegung im Jahre 1971. Beiträge zur österreichischen Statistik. Issue 300, Vienna: Österreichischen Statistischen Zentralamt. 1972.

Data on births and deaths at the municipal level for 1981 to 2011 are based on a special evaluation by *Statistik Austria*.

Communal tax data

Communal tax data for 2010 to 2012 are retrieved from the statistical database STATcube by *Statistik Austria* (https://statcube.at/statistik.at/). Local tax measures for 1987 use the following archival source (hard copies):

 Gebarungsübersichten 1987 (Gebietskörperschaften und sonstige öffentlich-rechtliche Körperschaften). Vienna: Österreichischen Statistischen Zentralamt in collaboration with the Federal Ministry of Finance. 1989.

Occupation and firm data

Data on occupation per sector at the municipality level and the number of firms per sector per municipality in 2011 are retrieved from the statistical database STATcube by *Statistik Austria* (https://statcube.at/statistik.at/).

Machinery in agriculture

Machinery in agriculture for 1953 and 1962 are available at the court-district level (*Gerichts-bezirk*) and are based on the following archival sources (hard copies):

- Ergebnisse der Erhebung des Bestandes an landwirtschaftlichen Maschinen und Geräten im Jahre 1953. Beiträge zur Österreichischen Statistik. Issue 13, Vienna: Österreichischen Statistischen Zentralamt. 1954.
- Ergebnisse der Erhebung des Bestandes an landwirtschaftlichen Maschinen und Geräten im Jahre 1962. Beiträge zur Österreichischen Statistik. Issue 13, Vienna: Österreichischen Statistischen Zentralamt. 1964.

 $^{^{51}}$ Data on the natural population movement for 1961 do not exist at the municipal level. I thus use the 1960 birth and death census for the birth and death rates for 1961.

The number of farms (incl. information of farm size intervals) are based on the following archival source (hard copies):

 Ergebnisse der Volks-, Berufs- und landwirtschaftlichen Betriebszählung vom 17. Mai 1939: Alpen- und Donau-Reichsgaue. Volume 559, Issue 13, Berlin/Vienna: Statistik des Deutschen Reichs. 1943.

Data on regional policies

Local eligibility for the European Regional Development Fund (ERDF) for 2000 to 2006 stem from the European Commission (https://ec.europa.eu/regional_policy/en/funding/erdf/).

Data on Red Army dismantling stem from:

Iber et. al (2008). Die Rote Armee in der Steiermark—Sowjetische Besatzung 1945.
 Graz: Leykam Buchverlag.

Geographic data

Municipal centroids in longitude and latitude and local land condition (i.e., total area, settlement area, usable land area) are retrieved from *Statistik Austria*. The location of highway slip roads are from google maps. Distance measures use Euclidean distances from municipal centroids. Measures on elevation and roughness are retrieved from the geographic information system QGIS (Version 2.18 and older).

Appendix C: Additional figures and tables

C.1 Additional figures

Figure C.3: Occupation zone proposals, liberation and final occupation

Proposal #1: August 22, 1944 by the UK

Proposal #2: November 13, 1944 by the Soviet Union



Notes: The maps show different occupation zone proposals by the Allies (figures at the top and in the middle), the liberation treatment as of May 9, 1945 (bottom left) and the final occupation zones (bottom right). Proposals #1 to #3 were official claims by the respective Ally (Erickson, 1950). Proposal #4 was an informal claim by the Soviet Union aiming to control southern Austria to exert influence on the Balkans but was never discussed by the other Allies (Stelzl-Marx, 2012). The date that corresponds to each proposal indicates the day on which the occupation proposal was shared (for the most part in secret) by the respective Ally. The dates for the liberation and occupation treatments show the time period when the respective troops controlled their liberated or occupied regions. Vienna was divided into four occupation zones for the respective Ally, and the Vienna city center (Innerer Berzirk) was a joint zone with rotating responsibility. Bold lines within Austria depict state borders, thin lines show the district borders. Sources: Erickson (1950) and Stelzl-Marx (2012).



Figure C.4: Population changes during and after WWII (District level)

Notes: The figures show the population at the level of 84 Austrian districts. The upper figures show the shift in district population from 1939 to March 1943 (left hand figure) and from 1939 to January 1945 (right hand figure). The lower figures show the shift in population over WWII until 1946 and until 2011 respectively. The occupation zones (Soviet Union and Western Allies) refer to the permanent Austrian occupation zones from July 1945 to 1955. The correlation of district population for 1939 and 1943 is 0.993 (for Styrian districts: 0.998) and the correlation for 1939 and 1945 is 0.983 (Styria: 0.989). The city district of Vienna is not shown in the figures (outlier at 14.4) but is included in the correlation based on food vouchers. Resident population is the entire population less the number of community catered people (mainly foreigners; in 1945 already German refugees). Sources: Die Zivilbevölkerung des Deutschen Reiches, 1940–1945; see Appendix B.



Figure C.5: Population per occupation zone and Vienna, 1869 to 2011

Notes: The figure shows the long-term evolution of population (in 100,000) according to the four post-WWII occupation zones in Austria and separately for quadripartite Vienna. The respective occupation zones were in place from July 1945 to October 1955. Styria, the region of this study, belongs to the UK occupation zone. Dashed vertical lines indicate the period of WWI; solid vertical lines the period of WWI.



Figure C.6: Population growth from 1939 to 1946 in entire Styria (in %)

Notes: The figures show population growth at the municipal level from 1939 (last census before WWII) to 1946 (first municipal population figures after WWII based on food vouchers) in percentage points for entire Styria. The map on the left-hand side shows the population growth at the level of 542 municipalities. The black bold line shows the line of contact between the Red Army and the Western Allies (along the municipality borders, if feasible). The right-hand figure depicts the average municipality population shift according to the liberation treatment (Red-Army-liberated parts vs. Western-Allies-liberated regions).



Figure C.7: Coefficient plots of different RD estimates w.r.t. 1939, 1869 to 2011

Notes: The figures show coefficient plots of spatial discontinuities in municipality population growth (in percentage points) w.r.t. the year 1939 across the intra-Styrian line of contact. The left-hand figures use distance of the nearest line-of-contact municipality as a single-dimensional forcing variable, the right-hand figures use longitude and latitude as multi-dimensional forcing variables. All RD estimates employ the same set of segment and regional fixed effects as in the baseline specification in Table 2. The sample is restricted to municipalities within 20km of the nearest line-of-contact municipality. The dashed vertical lines show the period of WWII, the grey solid line shows the period of the Red Army's presence in Styria. Coefficients are shown with 95% confidence bands based on spatially corrected standard errors.



Figure C.8: RD plots on local firm tax revenues per employee

Notes: The figures show RD plots on local tax revenue per employee in 2011 (in Euro) across the line of contact. Plots are based on a linear fit and include segment and geography fixed effects. The left-hand graph shows local tax figures without tax base controls, the right-hand graph includes the full set of tax base controls (see Columns (1) and (6) in Table 3). The bins represent local averages of municipalities within a 5-kilometer interval. Negative (positive) distances in all RD plots represent municipalities with respect to the nearest line-of-contact municipality that were liberated by the Western Allies (Red Army). The vertical dashed lines represent the line of contact. Thin grey lines represent 95% confidence interval.



Figure C.9: Effects on the metropolitan area of Graz with alternative matching

City of Graz versus synthetic control units

Notes: The figures compare population growth in the Graz region to population growth in its synthetic control unit (SCU). The upper figures show city level outcomes, and the figures in the middle show metropolitan area outcomes of municipalities within 25 kilometers of the respective city. The donor pool of the upper and middle figures consists of all Austrian cities (and their respective surrounding municipalities) with more than 10,000 inhabitants in 1939 that were liberated and occupied by the Western Allies after WWII. The left-hand figures compare the city of Graz (metropolitan area of Graz) with its SCU; the right-hand figures show the relative change in population of all major cities (all major metropolitan areas) compared to their respective SCU. SCUs are matched over population scores in 1880, 1910 and 1934 in respect to 1939, socio-demographic variables in 1934 (occupational share in industry and public sector, share of female) and population density of settlement area in 1934. The SCU of Graz consists of Bregenz (60.3%) and Bad Ischl (39.7%). The SCU of the metropolitan area of Graz consists of Dornbirn (80.1%), Villach (19.7%) and Wolfsberg (0.1%). The bottom figures consist of state capital cities that were liberated and occupied by the Western Allies (Graz, Klagenfurt, Linz, Salzburg, Innsbruck, Bregenz) and are matched over population scores. The state capital of Bregenz forms the SCU of Graz.



Figure C.10: Illustration of RD samples for heterogeneous effects

Notes: The maps show different regional subsamples of RD samples with respect to the distance of the nearest line of contact municipality (LoCM). The upper figures divide Styria according to the liberation treatment (US vs. Red-Army liberation in the Enns Valley; UK vs. Red-Army liberation in the Mur Valley and Graz region). The bottom figures extend the line of contact along the Styrian-Carinthian border where the federal state border coincides with the line of contact. The corresponding estimates with samples based on different regional subsamples are shown in Table C.11 in this Online Appendix.



Contiguous line of contact municipalities

Figure C.11: Population development from 1869 to 2011—Western Allies versus Red Army (1939=100)

Notes: The figures plot the population dynamics across the line of contact with respect to 1939 (=100). The upper figures consist of contiguous line-of-contact municipalities, the lower figures include municipalities within 10km of the nearest line of contact municipality. The left-hand figures compare population dynamics across the entire line of contact within Styria. The figures in the middle compare population figures across Red Army and US Army liberated places (Enns Valley). The right-hand figures compare population figures across Red Army and British Army liberated places (Mur Valley and Graz region). Dashed vertical lines represent the period of WWII, the grey solid line shows the period of the Red Army's presence in Styria.





Notes: The maps depict the exposure to regional shocks and regional policies during and after WWII. The upper left-hand map shows the targets of aerial bombing in Styria during the last years of WWII (and their respective surrounding municipalities) based on Ulrich (1978). The upper right-hand map shows the municipalities where the Red-Army officially dismantled plants in Styria (so-called German assets) according to Iber *et al.* (2008). The bottom maps depict regional policies after WWII. These are regions near to highway construction (left-hand map) and regions eligible to different ERDF funding scheme according to the EU funding period of 2000 to 2006 (for the funding periods before and thereafter, regions just changed status but Styrian internal variation in EU funding eligibility remains largely unchanged from 1994/5 to 2012). The corresponding estimates that control for potential effects of these regional shocks and policies are shown in Table C.12 (for population measures) and in Table C.13 (for local tax figures per employee) in this Online Appendix.




Notes: The maps show different RD samples with respect to the distance of the nearest line of contact municipality. The upper left-hand figure shows the sample of the contiguous line-of-contact municipalities (employed in the pairwise regressions). The right-hand map in the middle shows the main RD sample of the study with municipalities within 20km of the nearest line of contact municipality. The corresponding estimates that are based on samples with different bandwidths are shown in Table C.15 in this Online Appendix.

Figure C.14: Illustration of RD samples with pseudo-lines of contact



Notes: The maps show different pseudo-RD samples in Styria. The upper left-hand map depicts the pseudo-occupation treatment according to an informal Soviet occupation proposal along district borders in Styria according to (Stelzl-Marx, 2012) (see occupation proposals in Figure C.3 in this Online Appendix). The upper right-hand map shows the pseudo occupation treatment following the River Mur in the Graz region. The bottom maps show pseudo demarcation lines 20 kilometers and 40 kilometers east of the realized line of contact. Distance measures are always with respect to the nearest pseudo-line of contact municipality. The corresponding estimates that are based on samples with different pseudo-lines of contact are shown in Table C.19 in this Online Appendix.

C.2 Additional tables

						Dependent u	variable: Mu	nicipality pop	ulation growt	th 1939-1946					
		Entire Styria						RDD Sample			Line-of-contact municipalities (LoCM)				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)
Red Army	-7.96^{***} (2.34)	-10.39*** (2.39)	-9.76^{***} (2.09)	-4.49^{*} (2.68)	$^{-5.16**}_{(2.25)}$	-6.58^{**} (2.70)	-9.72^{***} (2.54)	-8.57^{***} (2.19)	-7.28*** (2.79)	-6.23*** (2.21)	-11.84^{*} (6.75)	-15.91** (7.20)	-14.16^{**} (6.10)	-10.87^{*} (5.67)	-13.76** (5.97)
Share of usable area		0.09^{***} (0.03)			-0.01 (0.07)		0.20^{***} (0.04)			$0.11 \\ (0.10)$		0.45^{***} (0.07)			0.67^{***} (0.15)
Elevation range			-0.03^{***} (0.01)		-0.03^{***} (0.01)			-0.04^{***} (0.01)		-0.03^{***} (0.01)			-0.07^{***} (0.02)		-0.03* (0.02)
Roughness			0.13^{***} (0.05)		0.11^{**} (0.05)			0.14^{**} (0.06)		0.12^{**} (0.06)			0.26^{**} (0.10)		0.22^{**} (0.08)
Distance to Graz				-0.35^{***} (0.08)	-0.26^{***} (0.08)				-0.46^{***} (0.12)	-0.19 (0.13)				$^{-1.34}_{(0.87)}$	-0.57 (0.43)
Dist. to Graz squared				0.00^{***} (0.00)	0.00^{***} (0.00)				0.00^{***} (0.00)	0.00^{**} (0.00)				$0.01 \\ (0.01)$	$0.00 \\ (0.00)$
Constant	114.74^{***} (2.24)	112.47^{***} (2.49)	$ \begin{array}{c} 118.91^{***} \\ (2.45) \end{array} $	$ \begin{array}{c} 119.34^{***} \\ (3.25) \end{array} $	120.87^{***} (5.62)	$ \begin{array}{c} 113.25^{***} \\ (2.46) \end{array} $	107.65^{***} (2.99)	121.53^{***} (2.81)	124.80^{***} (4.14)	$ \begin{array}{c} 115.36^{***} \\ (8.01) \end{array} $	113.81^{***} (6.49)	103.36^{***} (7.44)	128.62^{***} (10.25)	149.66^{***} (19.71)	96.05^{***} (13.85)
No. of obs.	539	539	539	539	539	191	191	191	191	191	48	48	48	48	48

Table C.2: Geographic determinants of population growth from 1939 to 1946

Notes: The dependent variable is municipality population growth from 1939 (last municipality census before WWII) to the first municipality population figures after WWII in 1946 (food vouchers data) in percentage points. The explanatory variables are time-invariant geographic controls. Divided municipalities along the line of contact are excluded from the regressions. Columns (1) to (5) show population growth for all of Styria; Columns (6) to (10) show population growth for the RD sample (municipalities within 20 kilometers of the nearest line of contact municipality); Columns (11) to (15) show population growth for the contiguous line-of-contact municipalities (pairwise sample). Spatial dependent standard errors and standard errors clustered at the pair level in Columns (11) to (15) respectively are in parentheses. Significance levels: *** 0.01, ** 0.05, * 0.10.

Table	C.3:	Summary	statistics
-------	------	---------	------------

		Full san	ple (entire	e Styria)		RDD Sai	mple (means)		-of-contact alities (means
	Obs.	Mean	Std. Dev.	Min.	Max.	Red Army (n=116)	Western Allies (n=75)	Red Army (n=24)	Western Allies (n=24)
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Liberation of Styria									
Red Army (n=446)	542	0.82	0.38	0	1	1.00	0	1.00	0
Western Allies (n=93)	542	0.17	0.38	0	1	0	1.00	0	1.00
US troops (n=37)	542	0.07	0.25	0	1	0	0.33	0	0.42
British troops (n=56)	542	0.10	0.30	0	1	0	0.67	0	0.58
Partitioned municipalities $(n=3)$	542	0.01	0.07	0	1	-	-	-	-
Population figures									
Population	8,672	1901.04	9062.03	99	261,726	1722.13	1138.37	2123.57	1444.07
Population per km ² of total area	8,672	96.25	178.30	1.46	3229.34	101.43	56.31	89.01	50.95
Population per km^2 of usable area	8,672	179.35	212.815	35.099	3231.42	196.924	152.81	207.84	156.45
Population in 1900 (1939=100)	542	95.39	16.69	32.92	158.26	93.74	94.26	90.84	93.35
Census 1939									
Population share female	542	50.26	2.60	27.65	59.67	48.83	49.64	47.48	48.25
Population share 20–65 years	542	65.37	4.08	52.99	83.74	66.18	64.10	65.83	65.12
Share unemployed ^a	542	10.60	5.34	2.19	42.05	12.19	11.60	12.72	11.65
Share agriculture	542	66.07	24.21	1.83	98.93	59.96	61.82	51.43	56.46
Share industry	542	22.51	16.59	1.02	85.19	25.89	23.83	31.46	28.01
Share self-employed	542	32.20	11.77	3.38	53.67	26.52	29.66	21.86	25.32
Share medium farms (10–19ha)	542	16.51	7.17	0.00	42.22	14.85	14.58	12.82	13.59
Census 1951									
Population share female	542	51.51	2.00	44.15	59.20	50.82	50.76	50.40	49.74
Population share 20 – 65 years	542	58.53	3.14	47.56	67.62	58.69	56.76	58.30	57.26
Population share foreigners	542	3.44	4.25	0.00	46.09	3.58	2.57	2.96	2.85
Share agriculture	542	58.11	24.63	1.59	95.98	51.93	53.25	46.36	49.42
Share industry	542	30.27	17.85	0.57	89.07	33.99	32.06	37.36	36.99
Share self-employed	542	21.92	6.65	4.09	35.54	19.39	18.26	16.60	16.25
Census 2011									
Population share female	542	50.28	1.72	37.75	55.71	50.37	50.45	50.45	50.53
Population share $20 - 65$ years	542	61.20	2.35	51.18	67.66	61.08	60.21	60.21	60.56
Population share foreigners	542	3.34	2.79	0.00	34.93	3.42	3.39	3.57	2.77
Share unemployed	542	3.88	1.58	0.65	11.08	3.81	3.92	4.14	4.50
Share agriculture	542	9.77	7.11	0.17	46.43	9.57	12.68	11.29	12.65
Share industry	542	34.97	7.99	10.39	58.18	33.58	32.68	34.38	30.40
Share self-employed	542	12.63	4.14	4.21	31.18	12.30	13.58	12.14	13.75
Share workplaces in industry	542	24.69	16.70	0.00	87.72	24.73	22.08	24.37	19.46
Share workplaces in services	542	51.07	18.04	9.46	97.17	52.00	50.54	52.41	56.59
Workplaces per firm	542	3.81	2.56	1.25	26.84	3.96	3.52	4.57	3.35
Industrial workplaces per ind. firm	542	9.10	9.55	0.00	75.63	9.77	7.67	11.59	6.88
Share medium farms (10–19ha)	542	18.59	7.40	0.00	45.45	16.84	16.46	14.36	15.42
Municipal tax revenue / employee $({\mathfrak C})^b$	540	400.23	241.75	13.43	1323.27	427.10	383.54	407.30	471.81
Geography									
Area $(in km^2)$	542	30.84	32.31	1.08	285.30	36.08	44.26	50.08	43.02
Share of usable area	542	48.82	24.44	2.19	100.13	43.04	27.58	23.13	32.14
Distance to highway slip road (in km)	542	12.11	8.80	0.22	39.08	9.01	19.38	11.69	9.73
Distance to Graz (in km)	542	47.43	28.87	0	141.85	41.05	83.35	71.47	57.69

Notes: The table shows the summary statistics for all 542 municipalities in Styria (territorial status in 2011). The shares are shown in percentage points. The descriptive statistics for the overall sample are shown in Columns (1) to (5). Columns (6) and (7) divide the RD sample with municipalities within 20 kilometers of the nearest line-of-contact municipality according to the liberation treatment (Red Army vs. Western Allies); Columns (8) and (9) divide line-of-contact municipalities according to the liberation treatment (Red Army vs. Western Allies). Population data cover 15 censuses from 1869 to 2011 and food vouchers data for 1946. Census data on demographic and occupation characteristics (according to the head of the family) in 1934/1939 and 1951 are self-compiled and merged to the municipal territorial status as of 2011. Recent data are retrieved from STATcube from *Statistik Austria*. See data sources in the Online Appendix B. a) Share of population without occupation (only available for 1934). b) Three-year averages (2010–2012).

Description	Functional form RDD polynomial			
(1)	(2)			
Panel A: Single-dimension	al RDD polynomials			
Linear	km			
Quadratic	$km + km^2$			
Cubic	$km + km^2 + km^3$			
Quartic	$km + km^2 + km^3 + km^4$			
Panel B: Multi-dimension	al RDD polynomials			
Linear	lon + lat			
Quadratic	lon + lon2 + lat + lat2 + (lon × lat)			
Cubic	$\boxed{\operatorname{lon} + \operatorname{lon}^2 + \operatorname{lon}^3 + \operatorname{lat} + \operatorname{lat}^2 + \operatorname{lat}^3 + (\operatorname{lon} \times \operatorname{lat}) + (\operatorname{lon}^2 \times \operatorname{lat}) + (\operatorname{lon} \times \operatorname{lat}^2)}$			
Quartic $lon + lon^2 + lon^3 + lon^4 + lat + lat^2 + lat^3 + lat^4 + (lon \times lat) + (lon \times lat^2) + (lon^2 \times lat^2) + (lon^2 \times lat^2) + (lon^2 \times lat^3) + (lon^3 \times lat) + (lon^3 \times lat^2)$				

Table C.4: Functional forms of RD polynomials

Notes: The table provides all functional forms of the RD polynomials that are employed in the paper (either in the main text or in the Online Appendix). The single-dimensional RD uses kilometers (km) as the forcing variable, defined as the municipality distance of the nearest line-of-contact municipality. Distances in the Red-Army-liberated areas (Western-Allies liberated areas) are positive (negative). The multi-dimensional RD employs longitude (lon) and latitude (lat) as forcing variables (standardized to Styrian-Carinthian means).

		Optimal bandwidth (BW) selection in kilometers								
	Population	n growth, 19	939 to 2011	Tax revenues	s per local e	mployee 2011	Average			
	BW estimator	BW bias	rho (est./bias)	BW estimator	BW bias	rho (est./bias)	BW estimator			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)			
Linear RD regression	11.76	18.37	0.64	17.59	22.43	0.78	14.68			
Quadratic RD regression	21.03	24.49	0.86	19.36	20.03	0.97	20.20			
Cubic RD regression	19.97	17.90	1.12	24.35	21.80	1.12	22.16			
Quartic RD regression	21.43	19.61	1.09	26.22	25.37	1.03	23.83			
Average bandwidth (columns)	18.55	20.09	-	21.88	22.41	-	20.22			
Segment & Geography FE No. of obs.	Yes 539	Yes 539	Yes 539	Yes 539	Yes 539	Yes 539	-			

Table C.5: RD sample selection

Notes: The table reports the optimal bandwidth selection for single-dimensional RD estimates based on Stata's package *rdrobust* (Calonico *et al.*, 2017). The estimates are based on the entire sample of Styrian municipalities (without divided municipalities). The estimates use linear, quadratic, cubic and quartic polynomial fits respectively and a uniform kernel function to construct the local-polynomial estimator(s). Selection of the bandwidth for the main "RD Sample" are based on averages of the BW estimators (20.22 kilometers ≈ 20 kilometers). A cubic polynomial fit for population figures and a quadratic polynomial fit for tax revenues per employee have bandwidths closest to the overall averages. The numbers in bold represent those bandwidths that are close to 20 kilometers. Rho (BW estimator divided by BW bias) is reasonably close to one for these bandwidth selections, too. The main estimates will thus use a bandwidth of 20 kilometers, a cubic RD polynomial fit for population estimates, and a quadratic RD polynomial fit for tax revenues per local employee.

							Depende	nt variabl	e: Municipalit	y covariates						
		Ge	eography			Census in	1934				Census in 1939					
	$\frac{\rm Area}{\rm (km^2)}$	Share use- able area	Average sea level	Distance to Graz (km)	Agriculture	Industry	Unemp- loyed	Female	Agriculture	Industry	${ m Farms} < 10 { m ha}(\%)$	Farms per useable area	Self emp- loyed	Pop per household	Age below 18	Age above 65
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)
Panel A: OLS	dummy s	specification														
Red Army	$0.48 \\ (0.57)$	15.46^{***} (5.49)	-409.55^{***} (83.66)	-42.30^{***} (7.73)	-0.81 (4.15)	$1.75 \\ (2.97)$	$0.60 \\ (1.04)$	-0.28 (0.34)	$^{-1.87}_{(4.25)}$	2.07 (2.95)	$ \begin{array}{c} 18.33^{***} \\ (4.63) \end{array} $	4.20^{***} (1.29)	-1.27 (4.01)	-0.62^{***} (0.20)	-2.85^{***} (1.00)	0.76^{*} (0.40)
No. of obs. Segment FE Geography FE	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No
Panel B: Pairw	vise regre	ession with a	contiguous d	lemarcation li	ine municipa	lities										
Red Army	$^{-1.12}_{(1.14)}$	-0.79 (4.50)	-23.88 (64.89)	-2.80 (6.46)	-2.31 (5.55)	-0.15 (4.53)	$0.42 \\ (1.01)$	-0.01 (0.32)	$^{-1.14}_{(5.23)}$	-0.93 (4.29)	1.24 (4.47)	1.27 (1.02)	-2.19 (3.79)	$\begin{array}{c} 0.27 \\ (0.28) \end{array}$	$\begin{array}{c} 0.46 \\ (0.83) \end{array}$	$\begin{array}{c} 0.01 \\ (0.23) \end{array}$
No. of obs. No. municipality Pair FE	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No	90 48 No
Panel C: Single	e-dimens	ional RDD	in distance t	o demarcatio	n line munic	ipalities										
Red Army	$0.19 \\ (1.05)$	$9.45 \\ (7.07)$	-185.22** (82.24)	-15.63 (9.70)	$^{-1.80}_{(7.95)}$	$\begin{array}{c} 0.05 \\ (5.94) \end{array}$	$0.76 \\ (1.64)$	-0.17 (0.48)	-0.65 (8.10)	-0.51 (5.39)	$6.00 \\ (6.00)$	$ \begin{array}{c} 0.37 \\ (2.20) \end{array} $	-3.45 (6.35)	-0.14 (0.26)	-0.63 (1.30)	$0.13 \\ (0.43)$
No. of obs. Segment FE Geography FE	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No
Panel D: Mult	i-dimensi	ional RDD i	n longitude	and latitude												
Red Army	$0.24 \\ (0.84)$	$3.27 \\ (6.96)$	-23.04 (76.01)	-0.41 (0.46)	4.41 (8.71)	-4.54 (7.15)	-0.50 (1.74)	$\begin{array}{c} 0.13 \\ (0.52) \end{array}$	3.83 (8.32)	-5.34 (6.49)	$5.00 \\ (6.38)$	$0.50 \\ (1.73)$	$0.65 \\ (6.01)$	-0.22 (0.27)	-0.96 (1.39)	$\begin{array}{c} 0.32 \\ (0.37) \end{array}$
No. of obs. Segment FE Geography FE	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No	191 No No

Table C.6: Unconditional continuity of covariates across the line of contact

Notes: The table tests for spatial discontinuities across line-of-contact municipalities in various time-invariant (Columns (1) to (4)) and pretreatment covariates (Columns (5) to (16)). Coefficients represent shares (in %) except for area (in km^2), average sea level (in m), Distance to Graz (in km), farms per km² and population per household. The shares of the industrial sectors, of unemployed and of self-employed are according to the head of the family. Panels A, C and D consist of municipalities within 20 kilometers of the nearest line-of-contact municipality (RD sample). Panel B consists of contiguous line-of-contact municipalities. Panel A employs an OLS dummy specification (Dummy=1 for Red-Army-liberated municipalities, 0 otherwise). RD estimates in Panel C and D employ a cubic polynomial fit. Estimates that include segment and geography fixed effects are shown in Table 1 of the paper. Standard errors in parentheses in Panel B are clustered at the contiguous pair level and corrected for spatial dependence in Panel A, C and D. Significance levels: *** 0.01, ** 0.05, * 0.10.

						Dep	pendent vari	able: Munici	pal populatio	on growth (ir	n %)					
		Pre-WWII period (census to census)						Pre-WW post-W				Post-WWII	period (cens	us to census)		
	1969-1880	1880-1890	1890-1900	1900-1910	1910-1923	1923-1934	1934-1939	1939-1946	1939-1951	1946-1951	1951-1961	1961-1971	1971-1981	1981-1991	1991-2001	2001-2011
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)	(16)
Panel A: Pairw	ise regressi	on with cor	ntiguous der	marcation l	ine municip	alities										
Red Army	$\begin{array}{c} 0.37 \\ (4.19) \end{array}$	$ \begin{array}{c} 1.05 \\ (2.37) \end{array} $	$ \begin{array}{r} 1.90 \\ (2.79) \end{array} $	$1.44 \\ (1.43)$	$ \begin{array}{r} 1.41 \\ (2.26) \end{array} $	$0.06 \\ (1.56)$	8.52^{**} (3.99)	-17.60^{***} (4.65)	-13.80*** (3.05)	-0.05 (3.99)	$2.26 \\ (1.96)$	$ \begin{array}{c} 0.05 \\ (2.07) \end{array} $	-2.04 (1.40)	-1.51 (1.47)	-5.34*** (1.54)	-4.21^{***} (1.13)
No. of obs. No. municipality Pair FE	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes
Panel B: Single	-dimension	al RDD in	distance to	demarcatio	on line mun	icipalities										
Red Army	$2.83 \\ (4.91)$	$4.18 \\ (3.00)$	$2.36 \\ (3.48)$	2.47^{*} (1.29)	$ \begin{array}{c} 1.08 \\ (2.72) \end{array} $	-0.85 (2.35)	$1.08 \\ (4.37)$	-11.18^{***} (4.10)	-11.60^{***} (2.44)	-4.46 (3.80)	$ \begin{array}{r} 1.39 \\ (2.36) \end{array} $	-0.30 (2.56)	-2.14 (1.68)	-2.60 (1.73)	-5.93^{***} (1.76)	-2.27^{*} (1.21)
No. of obs. Segment FE Geography FE	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes
Panel C: Multi-	dimension	al RDD in l	ongitude ar	nd latitude												
Red Army	-0.69 (4.32)	4.09 (3.23)	$ \begin{array}{c} 1.83 \\ (3.32) \end{array} $	3.49^{**} (1.49)	-0.68 (2.44)	-0.14 (2.45)	2.59 (3.55)	-6.43^{*} (3.47)	-9.74^{***} (2.53)	-7.16^{**} (3.20)	$2.68 \\ (2.25)$	$^{-1.02}_{(2.46)}$	-2.08 (1.71)	-2.43 (1.72)	-4.19^{***} (1.57)	$^{-1.15}_{(1.23)}$
No. of obs. Segment FE Geography FE	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes	191 Yes Yes

Table C.7: Population dynamics across the line of contact from census to census

Notes: The table tests for spatial discontinuities in municipality population growth (in percentage points) across the intra-Styrian line of contact from one census to the next census (incl. the population growth from the census in 1939 to the food voucher data in 1946 in column (8)). Panel A consists of contiguous line-of-contact municipalities. Panels B and C consist of municipalities within 20 kilometers of the nearest line-of-contact municipality (RD sample). The estimates in Panel A include pair fixed effects; estimates in Panels B and C include segment and geography fixed effects. RD estimates employ a cubic polynomial fit. Standard errors in parentheses in Panel A are clustered at the contiguous pair level and corrected for spatial dependence in Panels B and C. Significance levels: *** 0.01, ** 0.05, * 0.10.

			Communal tax rev	-	1 5 1 (- /
			Tax b	ase controls		
	Baseline	Sector	Firm size	Occupation	$\operatorname{Non-tax}$ occupation	All controls
	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: Pairwi	se regression					
Red Army	-146.12^{***} (48.16)	-116.57^{**} (49.52)	-196.95^{***} (40.80)	-121.65^{**} (47.13)	-122.21^{**} (47.23)	-139.94^{***} (47.81)
No. of obs.	90	90	90	90	90	90
No. of munc.	48	48	48	48	48	48
Pair FE	Yes	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2 adjusted	0.60	0.71	0.72	0.67	0.69	0.77
Panel B: Single-	dimensional R	DD				
Red Army	-105.01*	-94.59*	-125.83***	-85.49*	-96.84**	-98.76**
	(62.28)	(50.96)	(48.15)	(48.16)	(46.77)	(41.68)
No. of obs.	191	191	191	191	191	191
Seg. & Geo. FE	Yes	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2 adjusted ^a	0.21	0.48	0.45	0.46	0.45	0.54
Panel C: Multi-	dimensional RI	DD				
Red Army	-78.79	-58.20	-92.94*	-51.02	-65.49	-61.58
-	(56.76)	(52.37)	(49.51)	(46.01)	(46.61)	(46.46)
No. of obs.	191	191	191	191	191	191
Seg. & Geo. FE	Yes	Yes	Yes	Yes	Yes	Yes
R ² adjusted ^a	0.16	0.45	0.41	0.45	0.43	0.52

Notes: The table tests for spatial discontinuities in municipal tax revenues per local employee in 2011 across the intra-Styrian line of contact. Panel A consists of contiguous line-of-contact municipalities. Panels B and C consist of municipalities within 20 kilometers of the nearest line-of-contact municipality (RD sample). RD estimates employ a quadratic polynomial fit. Column (1) shows the baseline specification. Columns (2) to (5) include economic variables that determine the taxable base (Industrial sector controls: share of workplaces in industry, share of workplaces in services (agriculture as residual); Firm size controls: workplaces divided by the number of firms, industrial workplaces divided by the number of industrial firms; Work occupation controls: share of blue collar workers; Non-taxable occupations controls: share of self-employed, share of workplaces in public administration). Column (6) gives the combined view. Spatial discontinuities of variables that affect the taxable base are shown in Table C.22 in this Online Appendix. The estimates include segment and geography fixed effects. a) Adj. R2 measures stem from Stata's reg command with robust standard errors. Standard errors in parentheses in Panel A are clustered at the contiguous pair level and corrected for spatial dependence in Panels B and C. Significance levels: *** 0.01, ** 0.05, * 0.10.

Panel A		Dependen	t variable: Wage	sum tax in 1987 (in Austrian Schilling	5)
	Per po	pulation	Per working	g age cohort	Per taxable occ	cupation of residents
	(1)	(2)	(3)	(4)	(5)	(6)
Red Army	-66.29 (89.53)	-79.83 (80.51)	-101.21 (144.08)	-124.11 (129.23)	-96.00 (131.04)	-112.22 (117.19)
No. of obs. No. of municipalities Pair FE Lagged controls R ² adj.	90 48 Yes No 0.53	90 48 Yes Yes 0.69	90 48 Yes No 0.53	90 48 Yes Yes 0.69	90 48 Yes No 0.53	90 48 Yes Yes 0.69
Panel B		Dependent va	<i>uriable:</i> Business t	ax revenues in 19	87 (in Austrian Schil	lling)
	Per total	population	Per working	g age cohort	Per taxable occ	cupation of residents
	(1)	(2)	(3)	(4)	(5)	(6)
Red Army	-38.69 (91.65)	-74.37 (101.15)	-51.56 (149.30)	-106.89 (164.74)	-36.41 (137.36)	-79.98 (153.28)
No. of obs. No. of municipalities Pair FE Lagged controls R ² adj.	79 48 Yes No 0.49	79 48 Yes Yes 0.52	79 48 Yes No 0.49	79 48 Yes Yes 0.52	79 48 Yes No 0.48	79 48 Yes Yes 0.51
Panel C		Depend	dent variable: Co	mmunal tax reven	ues in 2011 (in €)	
	Per total	population	Per working	g age cohort	Per taxable occ	cupation of residents
	(1)	(2)	(3)	(4)	(5)	(6)
Red Army	-31.71 (28.77)	-35.99 (27.14)	-47.81 (49.41)	-54.65 (46.73)	-56.08 (54.66)	-61.54 (51.69)
No. of obs. No. of municipalities Pair FE Controls R ² adj.	90 48 Yes No 0.49	90 48 Yes Yes 0.60	90 48 Yes No 0.49	90 48 Yes Yes 0.59	90 48 Yes No 0.47	90 48 Yes Yes 0.58

Table C.9:	Measures o	f Labor	productivity	in	other years
			r		

Notes: The table tests for spatial discontinuities across contiguous line-of-contact municipalities for various tax figures in the year 1987 (Panels A and B) and compares the respective approach to the tax figures in 2011 (Panel C). Columns (1) and (2) divides municipality tax revenues by local population; Columns (3) and (4) by the local number of the working age cohort; and Column (5) and (6) by the local number of residents (according to the head of the family) that work in taxable occupation (i.e., without residents employed in agriculture or public sector). Lagged controls are the tax controls according to Table 3 for the year 1981 in Panels A and B. Standard errors in parentheses are clustered at the contiguous pair level. Significance levels: *** 0.01, ** 0.05, * 0.10.

Donor pool West Austria	Donor pool East Austria
Description of Sample: Liberated by Western Allies in May 1945; occupied by Western Allies (USA; UK; France) from July 1945 to October 1955	Description of Sample: Liberated by the Red Army in Spring 1945; occupied by the Soviet Union from July 1945 to October 1955
Linz	St. Pölten
Innsbruck	Wiener Neustadt
Salzburg	Klosterneuburg
Klagenfurt am Wörthersee	Baden
Villach	Krems an der Donau
Steyr	Amstetten
Wels	Mödling
Wolfsberg	Schwechat
Dornbirn	Stockerau
Bregenz	Ternitz
Feldkirch	Zwettl
Gmunden	Hollabrunn
Bad Ischl	Neunkirchen
	Waidhofen an der Ybbs
	Berndorf
	Perchtoldsdorf
	Mistelbach

Table C.10: Donor pool cities for the synthetic control method

Notes: The table lists the donor pools for the synthetic control unit of the city of Graz (of the metropolitan area of Graz). Donor pool cities are all Austrian cities above 10,000 inhabitants in 1939 that were either liberated and occupied by Western Allies or liberated by the Red Army and occupied by the Soviet Union after WWII. Cities are sorted by size in 1939 in the respective donor pools. Cities in bold are state capital cities. Quadripartite Vienna is excluded from all donor pools.

				De	pendent varial	ble: Population gr	owth (in %) an	d tax revenues	(in €)			
			Heterogeneous eff	fects within Sty	ria				UK Zone (Styri	a and Carinthi	a)	
		US vs Red Ar (Enns Valley		(Mur	UK vs Red Ar Valley and Gra		(incl. S	Entire UK zo tyrian-Carinth			order only North)	
	Populatio	on growth	Tax revenues	Populati	on growth	Tax revenues	Populatio	on growth	Tax revenues	Populati	on growth	Tax revenues
	1939-1951	1939 - 2011	per employee	1939-1951	1939 - 2011	per employee	1939 - 1951	1939 - 2011	per employee	1939-1951	1939 - 2011	per employee
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Panel A: Pairwise regre	ession											
Red Army	-23.20^{***} (5.32)	-56.94^{***} (14.47)	-280.10^{**} (116.71)	-9.07** (3.60)	-5.92 (7.53)	-99.45*** (36.16)	-14.10^{***} (2.63)	-27.87^{***} (6.44)	-132.61^{***} (39.92)	-15.44^{***} (4.78)	-46.57^{***} (5.48)	411.00 (421.85)
No. of obs. No. of municipalities Pair FE Tax base controls	30 17 Yes No	30 17 Yes No	30 17 Yes Yes	58 31 Yes No	58 31 Yes No	58 31 Yes Yes	110 59 Yes No	110 59 Yes No	110 59 Yes Yes	20 12 Yes No	20 12 Yes No	20 12 Yes Yes
Panel B: Single-dimens	ional RDD											
Red Army	-15.54^{***} (4.14)	-33.17^{*} (19.09)	$^{-170.68**}_{(69.44)}$	-8.31^{***} (2.55)	$^{-18.04*}_{(9.96)}$	-74.10^{***} (22.08)	-10.93^{***} (2.01)	-23.21*** (8.71)	-91.04^{**} (35.11)	-8.28^{**} (4.05)	-23.53^{***} (5.22)	-72.02 (81.39)
No. of obs. Segment & Geography FE Tax base controls	42 Yes No	42 Yes No	42 Yes Yes	146 Yes No	146 Yes No	146 Yes Yes	218 Yes No	218 Yes No	218 Yes Yes	56 Yes No	56 Yes No	56 Yes Yes
Panel C: Multi-dimensi	onal RDD											
Red Army	$^{-16.48**}_{(7.09)}$	-43.32^{**} (16.68)	-123.06 (78.68)	-8.27** (3.64)	-20.67^{*} (11.82)	-68.37^{***} (24.17)	-8.07^{***} (2.56)	-14.04^{*} (7.75)	-52.95 (45.26)	-36.12^{**} (14.38)	-36.78 (23.55)	-228.32* (134.13)
No. of obs. Segment & Geography FE Tax base controls	42 Yes No	42 Yes No	42 Yes Yes	146 Yes No	146 Yes No	146 Yes Yes	218 Yes No	218 Yes No	218 Yes Yes	56 Yes No	56 Yes No	56 Yes Yes

Table C.11: Heterogeneous effects and the entire UK zone

Notes: The table tests for spatial discontinuities across line-of-contact municipalities in municipality population growth (in percentage points) and municipal tax revenues per local employee for different subsamples along the line of contact. Columns (1) to (6) look at within-Styria heterogeneity; Columns (7) to (9) expand the line of contact along the Styrian-Carinthian border; Columns (10) to (12) look at the Styrian-Carintian state border sample separately. A graphical representation of the respective regional subsamples is shown in Figure C.10 in this Online Appendix. Panel A consists of contiguous line-of-contact municipalities and Panels B and C consist of municipalities within 20 kilometers of the nearest line-of-contact municipality in the respective regional subsample. RD estimates on population growth employ a cubic polynomial fit; RD estimates on tax revenues per local employee employ a quadratic polynomial fit. All estimates include segment and geography fixed effects and estimates on tax revenues per employee also control for economic variables that determine the taxable base (see Table 3). Standard errors in parentheses in Panel A are clustered at the contiguous pair level and corrected for spatial dependence in Panels B and C. Significance levels: *** 0.01, ** 0.05, * 0.10.

		Dependent variable: Municipality population growth 1939-2011 (in %)													
		Pai	rwise regres	sion		Single-dimensional RDD					Multi-dimensional RDD				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)
Red Army	-23.55^{***} (7.72)	-26.09^{***} (7.49)	-24.72^{**} (9.53)	-22.92^{***} (7.78)	-26.28^{***} (9.70)	-26.89^{***} (9.95)	-24.67^{***} (9.39)	-25.83^{***} (9.59)	-22.42^{**} (9.45)	-22.10** (9.94)	-22.89*** (7.90)	-21.96^{***} (7.97)	-23.13*** (8.03)	-21.50** (8.38)	-22.77^{***} (8.56)
Aerial bombing WWII	3.61 (53.24)				22.53 (48.60)	13.43 (16.22)				$18.49 \\ (20.46)$	10.69 (12.58)				12.87 (15.98)
Dismantled plants in 1945		$35.58 \\ (46.17)$			$43.29 \\ (57.74)$		-8.06 (13.11)			-15.43 (17.19)		-2.89 (13.39)			-8.07 (16.32)
Highway access in 2011			4.55 (13.82)		$6.11 \\ (14.16)$			$3.08 \\ (6.37)$		$1.15 \\ (6.01)$			$6.34 \\ (6.37)$		5.83 (6.23)
ERDF Objective-2				$8.91 \\ (12.22)$	7.56 (12.63)				-24.03 (18.55)	-22.80 (18.76)				$^{-3.89}_{(12.89)}$	$^{-1.40}_{(14.52)}$
ERDF Phasing-out ^{a}									-29.37* (17.67)	-29.20 (17.67)				-5.89 (11.92)	$^{-3.96}_{(13.25)}$
No. of obs.	90	90	90	90	90	191	191	191	191	191	191	191	191	191	191
No. of municipalities Pair FE	48 Yes	48 Yes	48 Yes	48 Yes	48 Yes	191 No	191 No	191 No	191 No	191 No	191 No	191 No	191 No	191 No	191 No
Segment & Geo. FE	No	No	No	No	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

Table C.12: Effects of regional shocks and regional policies on population dynamics

Notes: The table tests for spatial discontinuities across line-of-contact municipalities in municipality population growth (in percentage points) and controls for regional shocks and regional policies during and after WWII. Columns (1) to (5) consist of contiguous line-of-contact municipalities. Columns (6) to (15) consist of municipalities within 20 kilometers of the nearest line-of-contact municipality (RD sample). RD estimates employ a cubic polynomial fit and include segment and geography fixed effects. Proxies for regional shocks and regional policies are: Aerial bombing during WWII (dummy that equals 1 for targeted municipalities and their direct neighbors); Dismantled plants (dummy that equals 1 for municipalities within 5 kilometers of the nearest municipality where the Red Army officially dismantled plants); Highway access (dummy that equals 1 if the nearest slip road is within 10 km of a municipality); ERDF Objective-2 (Dummy that equals 1 for municipalities that were eligible for Objective-2 EU funds in the 2000–2006period); ERDF Phasing-out (Dummy that equals 1 for municipalities that were eligible for Phasing-out EU funds in the 2000–2006 period). a) ERDF Phasing-out" serves as the reference category in the pairwise regression because areas that are not eligible for EU funds are entirely missing at the line of contact. Graphical representations of the respective regional shocks during WWII and regional policies after WWII are shown in Figure C.12 in this Online Appendix. Standard errors in parentheses Columns (1) to (5) are clustered at the contiguous pair level and corrected for spatial dependence in Columns (6) to (15). Significance levels: *** 0.01, ** 0.05, * 0.10.

		Dependent variable: Municipality tax revenues per employee (in \mathfrak{C})													
		Pa	irwise regress	sion			Sing	le-dimension	al RDD		Multi-dimensional RDD				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	(13)	(14)	(15)
Red Army	-153.30^{***} (42.94)	-169.14^{***} (45.08)	-114.91^{**} (45.48)	-150.21^{***} (40.11)	-134.36^{***} (46.57)	-103.60*** (36.38)	-97.16** (38.23)	-96.83^{***} (36.31)	-104.34^{***} (36.67)	-100.60*** (35.34)	-66.87 (44.04)	-57.15 (46.26)	-53.17 (41.83)	-69.05^{*} (39.36)	-67.68^{*} (39.91)
Aerial bombing WWII	-139.58 (96.07)				-37.60 (79.10)	64.62^{**} (32.31)				76.75^{**} (37.53)	86.19^{**} (34.51)				83.38^{**} (35.36)
Dismantled plants in 1945		247.20^{**} (117.75)			$199.64 \\ (120.73)$		-26.00 (39.85)			-50.78 (42.54)		$^{-1.48}_{(39.38)}$			-31.09 (41.57)
Highway access in 2011			$^{-164.80**}_{(82.01)}$		$^{-149.71*}_{(85.41)}$			-30.91 (36.60)		-28.86 (38.15)			-42.24 (35.93)		-43.90 (36.55)
ERDF Objective-2				$^{-13.04}_{(94.36)}$	$11.32 \\ (90.05)$				-41.37 (31.13)	-40.23 (28.90)				$^{-104.85^{***}}_{(37.10)}$	-101.74^{***} (35.58)
ERDF Phasing-out ^{a}									$19.82 \\ (47.20)$	$13.65 \\ (48.12)$				-35.76 (42.72)	-41.48 (44.72)
No. of obs. No. of municipalities Pair FE	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	191 191 No	191 191 No	191 191 No	191 191 No	191 191 No	191 191 No	191 191 No	191 191 No	191 191 No	191 191 No
Fair FE Segment & Geo. FE Tax base controls	Yes No Yes	Yes No Yes	Yes No Yes	Yes No Yes	Yes No Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes	Yes Yes

Table C.13: Effects of regional shocks and regional policies on local tax revenues

Notes: The table tests for spatial discontinuities across line-of-contact municipalities in local tax revenue per employee and controls for regional shocks and regional policies during and after WWII. Columns (1) to (5) consist of contiguous line-of-contact municipalities. Columns (6) to (15) consist of municipalities within 20 kilometers of the nearest line-of-contact municipality (RD sample). RD estimates employ a quadratic polynomial fit and include segment and geography fixed effects. All estimates control for economic variables that determine the taxable base (see Table 3). Proxies for regional shocks and regional policies are: Aerial bombing during WWII (dummy that equals 1 for targeted municipalities and their direct neighbors); Dismantled plants (dummy that equals 1 for municipalities within 5 kilometers of the nearest municipality where the Red Army officially dismantled plants); Highway access (dummy that equals 1 if the nearest slip road is within 10 km of a municipality); ERDF Objective-2 (Dummy that equals 1 for municipalities that were eligible for Objective-2 EU funds in the 2000–2006 period); ERDF Phasing-out (Dummy that equals 1 for municipalities that were eligible for Phasing-out EU funds in the 2000–2006 period). a) ERDF Phasing-out" serves as the reference category in the pairwise regression because areas that are not eligible for EU funds are entirely missing at the line of contact. Graphical representations of the respective regional shocks during WWII and regional policies after WWII are shown in Figure C.12 in this Online Appendix. Standard errors in parentheses in Columns (1) to (5) are clustered at the contiguous pair level and corrected for spatial dependence in Columns (6) to (15). Significance levels: *** 0.01, ** 0.05, * 0.10.

	Dependent variable: Population growth (in %) and tax revenues (in \mathfrak{C})										
	Si	ingle-dimensional	RDD	N	Iulti-dimensional	RDD					
	Populatio	on growth	Tax revenues	Populatio	on growth	Tax revenues					
	1939–1951	1939 - 2011	per employee	1939–1951	1939 - 2011	per employee					
	(1)	(2)	(3)	(4)	(5)	(6)					
Panel A: Linear polyno	mial										
Red Army	-10.50*** (2.22)	-27.13^{***} (9.57)	-100.93*** (38.50)	-8.19*** (2.37)	-12.35 (7.83)	-38.41 (46.24)					
Panel B: Quadratic pol	ynomial										
Red Army	-10.42*** (2.26)	-27.03^{***} (9.49)	-99.61*** (37.50)	-6.97^{***} (2.44)	-11.05 (8.78)	-57.35 (43.72)					
Panel C: Cubic polynoi	mial										
Red Army	-11.60^{***} (2.44)	-25.70^{***} (9.52)	-78.59* (41.37)	-9.74^{***} (2.53)	-22.14^{***} (7.91)	-48.32 (48.03)					
Panel D: Quartic polyn	omial										
Red Army	-11.65^{***} (2.45)	-25.36^{***} (9.53)	-82.88* (42.35)	-11.90*** (3.25)	-31.85^{***} (11.59)	-69.77 (47.71)					
No. of obs. in all samples Segment & Geography FE Tax base controls	191 Yes No	191 Yes No	191 Yes Yes	191 Yes No	191 Yes No	191 Yes Yes					

Table C.14: Alternative RD polynomials

Notes: The table tests for spatial discontinuities across line-of-contact municipalities in municipality population growth (in percentage points) in Columns (1), (2), (4) and (5) and for spatial discontinuities in local tax revenues per local employee in Columns (3) and (6). Panels A to D employ different functional forms of the RD polynomial for both the single-dimensional RD and the multi-dimensional RD. The definition of the respective functional forms of the polynomial are shown in Table C.4 in this Online Appendix. All estimates include segment and geography fixed effects and Columns (3) and (6) also control for economic variables that determine the taxable base (see Table 3). Standard errors in parentheses are corrected for spatial dependence. Significance levels: *** 0.01, ** 0.05, * 0.10.

		Dependent variable: Population growth (in %) and tax revenues (in \mathfrak{C})												
]	Bandwidth ± 10)km	:	Bandwidth ± 15	5km	Bandwidth $\pm 25 \text{km}$			Bandwidth ± 30 km				
	Populatio	on growth	Tax revenues	Populatio	on growth	Tax revenues	Populatio	on growth	Tax revenues	Populatio	on growth	Tax revenues		
	1939-1951	1939 - 2011	per employee	1939 - 1951	1939-2011	per employee	1939-1951	1939 - 2011	per employee	1939-1951	1939 - 2011	per employee		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)		
Panel A: Single-dimens	ional RDD 4													
Red Army	-10.22*** (2.77)	-13.54 (9.98)	-108.49*** (35.76)	-11.24*** (2.36)	-18.01* (9.45)	-85.70** (39.68)	-10.56^{***} (2.48)	-25.27^{***} (9.52)	-98.04^{**} (37.96)	-12.55*** (2.33)	-31.57*** (10.82)	-82.61^{**} (37.45)		
No. of obs. Segment & Geography FE Tax base controls	111 Yes No	111 Yes No	111 Yes Yes	158 Yes No	158 Yes No	158 Yes Yes	232 Yes No	232 Yes No	232 Yes Yes	268 Yes No	268 Yes No	268 Yes Yes		
Panel B: Multi-dimensi	onal RDD4													
Red Army	-9.21^{***} (3.05)	-11.33 (11.68)	-73.84^{*} (37.58)	$^{-11.75^{***}}_{(2.77)}$	-24.42^{**} (10.01)	-56.17 (35.21)	-11.22^{***} (2.97)	-19.40** (8.88)	-60.32 (42.58)	-10.08^{***} (2.90)	-22.62^{**} (8.84)	-56.00 (43.41)		
No. of obs. Segment & Geography FE Tax base controls	111 Yes No	111 Yes No	111 Yes Yes	158 Yes No	158 Yes No	158 Yes Yes	232 Yes No	232 Yes No	232 Yes Yes	268 Yes No	268 Yes No	268 Yes Yes		

Table C.15: Different bandwidths

Notes: The table tests for spatial discontinuities across line-of-contact municipalities in municipality population growth (in percentage points) and municipal tax revenues per local employee. The estimates employ different bandwidths of 10, 15, 25 and 30 kilometers of the nearest line-of-contact municipality. Panel A uses single-dimensional RD; Panel B uses multi-dimensional RD. RD estimates on population growth employ a cubic polynomial fit; RD estimates on tax revenues per local employee employ a quadratic polynomial fit. A graphical representation of the respective regional samples is shown in Figure C.13 in this Online Appendix. All estimates include segment and geography fixed effects and estimates on tax revenues per employee also control for economic variables that determine the taxable base (see Table 3). Standard errors in parentheses are corrected for spatial dependence. Significance levels: *** 0.01, ** 0.05, * 0.10.

		Depende	nt variable:	Population gr	owth (in %) a	and tax revenu	ıes (in €)	
	Eq	ual bandwidt	h (MSE-crite	ria)	Unequ	al bandwidth	(MSE-two-c	riteria)
	Populatio	on growth	Tax r	evenues	Populatio	on growth	Tax r	evenues
	1939–1951	1939–2011	w/o tax controls	Incl. tax controls	1939–1951	1939–2011	w/o tax controls	Incl. tax controls
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Panel A: Linear local polyno	mial for poin	nt estimator						
Red Army	-11.77^{***} (3.73)	-22.43** (9.11)	-107.47^{**} (50.78)	-85.22** (40.96)	-11.53^{***} (3.56)	-23.91*** (8.41)	-125.47^{**} (51.05)	-84.03^{**} (40.60)
BW selection in km (left/right) Effective No. of obs. (left/right)	$12.6/12.6 \\ 56/83$	${11.8/11.8 \atop 54/77}$	$17.6/17.6 \\ 69/108$	$15.1/15.1 \\ 63/95$	${12.2/19.7 \atop 54/116}$	$17.4/14.8 \\ 68/95$	$rac{15.6}{18.3} \ 65/109$	$rac{15.9/15.2}{66/95}$
Panel B: Quadratic local pol	ynomial for	point estima	tor					
Red Army	-12.04^{***} (4.16)	-24.24^{**} (9.57)	-68.41 (57.05)	-98.98** (45.76)	-12.61^{***} (4.02)	-23.89^{**} (9.89)	-117.07^{**} (55.61)	-147.56^{***} (41.74)
BW selection in km (left/right) Effective No. of obs. (left/right)	${19.2/19.2 \atop 72/112}$	$21.0/21.0 \\ 76/123$	$19.4/19.4 \\ 72/115$	${13.1/13.1} \atop {57/85}$	$20.1/36.1 \\ 76/222$	$20.1/25.5 \\ 76/150$	$21.0/23.2 \\ 76/141$	$14.4/35.8 \\ 61/220$
Panel C: Cubic local polynor	nial for poin	t estimator						
Red Army	-11.40^{**} (4.48)	-17.17^{*} (9.99)	-70.85 (58.66)	-106.72^{**} (46.22)	-12.75^{***} (4.23)	-25.94^{***} (9.77)	-99.44^{*} (58.10)	-138.70^{***} (42.89)
BW selection in km (left/right) Effective No. of obs. (left/right)	$28.0/28.0 \\ 85/168$	$20.0/20.0 \\ 75/116$	$24.3/24.3 \\ 82/143$	$15.8/15.8 \\ 66/96$	$27/57.2 \\ 85/365$	$19.0/39.0 \\ 72/248$	$21.5/34.5 \\ 76/212$	$rac{16.4/37.2}{67/228}$
Panel D: Quartic local polyn	omial for po	int estimato	r					
Red Army	-10.95^{**} (4.49)	-17.81^{*} (10.06)	-77.80 (59.71)	-115.77^{***} (43.63)	-11.62^{***} (4.49)	-22.02^{**} (10.06)	-108.86^{*} (58.98)	-131.69^{***} (43.50)
BW selection in km (left/right) Effective No. of obs. (left/right)	$23.5/23.5 \\ 81/143$	$21.4/21.4 \\ 76/127$	$rac{26.2/26.2}{84/156}$	$21.8/21.8 \\ 78/131$	$23.9/57.0 \\ 82/365$	$19.4/46.4 \\ 73/292$	$25.1/47.0 \\ 84/298$	$22.6/58.6 \\ 79/377$
No. of obs. in all samples Segment and Geography FE Tax base controls	539 Yes No	539 Yes No	539 Yes No	539 Yes Yes	539 Yes No	539 Yes No	539 Yes No	539 Yes Yes

Table C.16:	Optimal	data-driven	bandwidth	selection	procedure	(rdrobust))
-------------	---------	-------------	-----------	-----------	-----------	------------	---

Notes: The table employs the optimal data driven bandwidth selection procedure rdrobust (Calonico et al., 2014a,b, 2017) to test for spatial discontinuities in municipality population growth (in percentage points) and municipal tax revenues per local employee (three-year averages for 2010–2012 in Euro) across the intra-Styria line of contact. Bandwidth selection is based on uniform weighting of observations and ignores the presence of mass points (i.e., line-of-contact municipalities). Columns (1) to (4) employ equal bandwidth selection procedures on both sides of the line of contact; Columns (5) to (8) allow for different bandwidth selections on both sides. "BW selection in km" reports the optimal bandwidth to the left (Western Allies) and to the right (Red Army) of the line of contact. The "Effective number of observations" reports the respective numbers of municipalities in the subsamples. All estimates include segment and geography fixed effects and Columns (4) and (8) also control for economic variables that determine the taxable base (see Table 3). Standard errors are based on rdrobust and are in parentheses. Significance levels: *** 0.01, ** 0.05, * 0.10.

	Dependent variable: Population growth (in %) and tax revenues (in \mathfrak{C})										
		Single-dim	ensional RDD			Multi-dim	ensional RDD				
	Populatio	on growth	Tax rev	enues	Populatio	on growth	Tax rev	enues			
	1939-1951	1939–2011	w/o tax base controls	Tax base controls	1939–1951	1939–2011	w/o tax base controls	Tax base controls			
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)			
Red Army	-11.60	-25.70	-106.71	-99.61	-9.74	-22.14	-74.22	-57.35			
Spatial correlated standa	rd errors										
Spatial cutoff 0.0 km	$(3.94)^{***}$	(9.27)***	(51.50)**	(37.48)***	$(3.54)^{***}$	(8.28)***	$(52.04)^{\#}$	(45.92)			
Spatial cutoff 2.5 km	(3.93)***	(8.83)***	(52.21)**	(37.88)***	$(3.55)^{***}$	(8.19)***	$(53.12)^{\#}$	(46.43)			
Spatial cutoff 5.0 km	(3.39)***	(9.18)***	(53.29)**	(36.46)***	(3.37)***	(8.15)***	$(55.01)^{\#}$	(46.34)			
Spatial cutoff 7.5 km	$(2.44)^{***}$	(9.52)***	(59.30)*	(37.50)***	(2.53)***	(7.91)***	$(55.60)^{\#}$	$(43.72)^{\#}$			
Spatial cutoff 10.0 km	$(2.50)^{***}$	(10.21)**	(58.08)*	(38.69)**	$(1.62)^{***}$	(8.11)***	$(56.55)^{\#}$	$(40.13)^{\#}$			
Spatial cutoff 12.5 km	$(1.25)^{***}$	(9.92)**	(58.40)*	(40.32)**	$(1.01)^{***}$	(8.26)***	$(53.99)^{\#}$	(38.43)##			
Spatial cutoff 15.0 km	$(2.67)^{***}$	(10.85)**	(57.76)*	(44.04)**	$(1.81)^{***}$	(8.86)**	(60.58)	(45.60)			
Spatial cutoff 17.5 km	$(3.82)^{***}$	$(9.69)^{***}$	(53.92)**	$(47.29)^{**}$	$(2.42)^{***}$	$(6.90)^{***}$	$(55.43)^{\#}$	(50.04)			
Spatial cutoff 20.0 km	(3.90)***	$(9.51)^{***}$	$(34.85)^{***}$	(31.24)***	$(2.67)^{***}$	$(6.85)^{***}$	$(42.63)^*$	$(36.26)^{\#\#}$			
Spatial cutoff 25.0 km^a	$(2.94)^{***}$	$(10.76)^{**}$	$(24.29)^{***}$	$(13.47)^{***}$	$(1.26)^{***}$	$(7.46)^{***}$	(29.37)**	$(10.48)^{***}$			
Spatial cutoff 30.0 km	$(1.85)^{***}$	(9.93)**	(38.22)***	$(16.64)^{***}$	$(1.97)^{***}$	$(7.26)^{***}$	$(48.95)^{\#\#}$	(11.21)***			
Spatial cutoff 35.0 km	$(2.92)^{***}$	(9.12)***	(29.63)***	(19.08)***	$(2.51)^{***}$	$(3.76)^{***}$	$(48.10)^{\#\#}$	(13.00)***			
Spatial cutoff 40.0 km	$(2.19)^{***}$	(7.37)***	(28.01)***	(29.09)***	$(1.86)^{***}$	$(3.27)^{***}$	$(47.42)^{\#\#}$	$(37.24)^{\#\#}$			
Spatial cutoff 45.0 km	$(2.18)^{***}$	(6.23)***	(14.95)***	(34.30)***	(2.49)***	(1.02)***	(32.49)**	(13.30)***			
Spatial cutoff 50.0 ${\rm km}^b$	$(1.72)^{***}$	$(4.91)^{***}$	$(16.00)^{***}$	(29.95)***	$(2.97)^{***}$	$(2.03)^{***}$	$(14.92)^{***}$	(19.04)***			
"Conventional" clustered	standard e	rrors									
No correction Clustered at municipality	$(3.57)^{***}$ $(4.06)^{***}$	$(11.31)^{**}$ $(9.58)^{***}$	$(55.39)^*$ $(53.05)^{**}$	$(42.75)^{**}$ $(39.38)^{**}$	$(3.47)^{***}$ $(3.71)^{***}$	$(10.66)^{**}$ $(8.67)^{**}$	$(55.33)^{\#}$ $(54.06)^{\#}$	$(42.32)^{\#}$ (48.68)			
No. of obs. Segment and Geography FE Tax base controls	191 Yes No	191 Yes No	191 Yes No	191 Yes Yes	191 Yes No	191 Yes No	191 Yes No	191 Yes Yes			

Table C.17: Standard errors with different spatial cut-offs (RD)

Notes: The table shows different standard errors based on different spatial cutoffs and "conventional" standard errors for the main results in Table 2 and Table 3. The samples consist of municipalities within 20 kilometers of the nearest line-of-contact municipality (RD sample). Population figures employ a cubic polynomial fit, local tax revenues per employee employ a quadratic polynomial fit. Segment and geography fixed effects, and tax base controls equal the ones in the main tables. Robust standard errors with various spatial cutoffs (Conley, 1999, 2008; Colella *et al.*, 2019) and "conventional" standard errors are in parentheses. a) Spatial cutoffs in Columns (4) and (8) differ slightly from the reported ones to achieve feasible standard errors. b) Spatial cutoff in Column (8) slightly differs from the reported one to achieve feasible standard errors. Significance levels: *** 0.01, ** 0.05, * 0.10, ## 0.15, # 0.20.

	Dependent variable: Population growth (in %) and tax revenues (in $\textcircled{\mbox{\ \ e}})$										
	Pairwise regres	sion outcomes	MHT	adjusted p-values	(based on mhtre	eg)					
	Coefficient (Red Army)	Naïve p-values	Unadjusted	Thm. 3.1 (LSX-2019)	Bonferroni	Holm					
	(1)	(2)	(3)	(4)	(5)	(6)					
Main outcome variables:											
Population growth, 1939-2011 (in %)	-23.714	0.003***	0.003***	0.037**	0.051*	0.048**					
Local tax revenue per employee (in \mathfrak{C})	-153.288	0.001***	0.001***	0.016**	0.020**	0.020**					
Further variables of interest (amo	ng them tax-base	e controls and	variables for ch	annel discussio	n):						
Workplaces in agriculture (in %)	7.680	0.024**	0.025**	0.208	0.422	0.322					
Workplaces in industry (in %)	1.262	0.662	0.635	0.953	1.000	1.000					
Workplaces in services (in %)	-8.943	0.020**	0.019**	0.175	0.320	0.263					
Workplaces in public services (in %)	0.408	0.772	0.765	0.947	1.000	1.000					
Workplaces per resident	-0.008	0.815	0.791	0.791	1.000	0.791					
Workplaces per firm	0.437	0.316	0.287	0.866	1.000	1.000					
Workplaces per farm	-0.044	0.449	0.419	0.925	1.000	1.000					
Workplaces per industrial firm	3.166	0.061*	0.053^{*}	0.364	0.901	0.583					
Workplaces per service firm	-0.267	0.540	0.528	0.937	1.000	1.000					
Blue collar (in %)	-2.843	0.225	0.205	0.784	1.000	1.000					
Self-employed (in %)	3.824	0.309	0.275	0.864	1.000	1.000					
Compulsory education (in %)	1.962	0.009***	0.007***	0.074^{*}	0.112	0.099*					
Tertiary education (in %)	-0.311	0.390	0.364	0.922	1.000	1.000					
Unemployed (in %)	-0.589	0.121	0.106	0.575	1.000	1.000					
Out-commuters (in %)	-3.318	0.062*	0.049**	0.353	0.833	0.588					

Notes: The table reports p-values adjusted for multiple hypothesis testing (MHT) for different outcomes based on all variables of potential interest in 2011 (main outcomes and the variables reported in the channel discussion in Table C.22 in the Online Appendix). Columns (1) and (2) show the treatment coefficients and the respective p-values from the pairwise regression with line-of-contact municipalities. Columns (3) to (6) show different p-values adjusted for multiple hypothesis testing. The reported p-values control the familywise error rates (FWER) on the treatment coefficient for each outcome and stem from the Stata command "*mhtreg*" introduced by Barsbai *et al.* (2020). *mhtreg* is an adaption to List *et al.* (2019) for regressions (LSX-2019). All reported pairwise regressions are jointly estimated with pair fixed effects, clustered standard errors at the pair level and 5000 replications. Bootstrapped resampling is also clustered. ***/**/* indicate that the corresponding (FWER adjusted) p-values are less than 1%, 5% and 10% respectively.

	Dependent variable: Population growth (in %) and tax revenues (in $$)													
	Sovi	et occupation p	proposal	А	long the Mur l	River		20 km eastwa	rd		40 km eastwa	rd		
	Populatio	on growth	Tax revenues	Populatio	on growth	Tax revenues	Populatio	on growth	Tax revenues	Populati	on growth	Tax revenues		
	1939-1951	1939 - 2011	per employee	1939 - 1951	1939 - 2011	per employee	1939 - 1951	1939 - 2011	per employee	1939-1951	1939 - 2011	per employee		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)		
Panel A: Pairwise regre	ssion													
Red Army	-5.25 (5.60)	2.31 (16.08)	-73.76 (108.15)	-7.85 (4.90)	-11.77 (15.42)	41.41 (92.00)	$ \begin{array}{c} 0.69 \\ (2.45) \end{array} $	-3.15 (10.03)	-71.61^{*} (39.90)	-3.50 (2.12)	3.79 (7.22)	$^{-1.62}_{(33.34)}$		
No. of obs. No. of municipalities Pair FE Tax base controls	48 25 Yes No	48 25 Yes No	48 25 Yes Yes	50 30 Yes No	50 30 Yes No	50 30 Yes Yes	110 55 Yes No	110 55 Yes No	110 55 Yes Yes	108 56 Yes No	108 56 Yes No	108 56 Yes Yes		
Panel B: Single-dimens	ional RDD													
Red Army	-4.43 (8.06)	5.89 (21.40)	-9.11 (39.57)	-6.53 (5.12)	1.62 (23.34)	-16.63 (18.31)	$ \begin{array}{r} 1.81 \\ (4.07) \end{array} $	7.40 (16.63)	4.61 (46.54)	-2.73 (2.65)	$13.68 \\ (14.50)$	30.27 (27.71)		
No. of obs. Segment & Geography FE Tax base controls	128 Yes No	128 Yes No	128 Yes Yes	191 Yes No	191 Yes No	191 Yes Yes	272 Yes No	272 Yes No	272 Yes Yes	279 Yes No	279 Yes No	279 Yes Yes		
Panel C: Multi-dimensi	onal RDD													
Red Army	-3.15 (5.56)	7.08 (23.98)	-2.28 (32.96)	-6.87 (5.07)	4.19 (23.89)	-28.24 (19.22)	5.04 (5.61)	22.07 (20.43)	4.59 (50.13)	$ \begin{array}{c} 0.32 \\ (2.22) \end{array} $	31.67^{*} (16.53)	23.10 (30.86)		
No. of obs. Segment & Geography FE Tax base controls	128 Yes No	128 Yes No	128 Yes Yes	191 Yes No	191 Yes No	191 Yes Yes	272 Yes No	$\begin{array}{c} 272 \\ \mathrm{Yes} \\ \mathrm{No} \end{array}$	272 Yes Yes	279 Yes No	279 Yes No	279 Yes Yes		

Table C.19: Soviet proposal and pseudo lines of contact

Notes: The table tests for spatial discontinuities across pseudo-line-of-contact municipalities in municipality population growth (in percentage points) and municipal tax revenues per local employee. Columns (1) to (3) test for potential discontinuities along an occupation zone proposal by the Soviet Union (see Figure C.3 in this Online Appendix). Columns (4) to (6) test for potential discontinuities along the River Mur in the Graz region. Columns (7) to (12) test for potential discontinuities along pseudo-line-of-contact municipalities that are located 20 and 40 kilometers east of the realized line of contact respectively. Panel A consists of contiguous line-of-contact municipalities and Panels B and C consist of municipalities within 20 kilometers of the nearest line-of-contact municipality in the respective pseudo-sample. RD estimates on population growth employ a cubic polynomial fit; RD estimates on tax revenues per local employee employ a quadratic polynomial fit. A graphical representation of the respective pseudo-samples is shown in Figure C.14 in this Online Appendix. All estimates include segment and geography fixed effects and estimates on tax revenues per employee also control for economic variables that determine the taxable base (in accordance with Table 3). Standard errors in parentheses in Panel A are clustered at the contiguous pair level and corrected for spatial dependence in Panels B and C. Significance levels: *** 0.01, ** 0.05, * 0.10.

Pa Overall (1) -0.65*** (0.24)	irwise Per period (2)	L 	oCM Per period (4)	Overall	to LoCM Per period	Overall	to LoCM Per period		n to LoCM
(1) -0.65***		(3)			Per period	Overall	Per period		
-0.65***	(2)		(4)		Overall Per period		rei period	Overall	Per period
		-0.46**		(5)	(6)	(7)	(8)	(9)	(10)
		(0.20)		-0.39** (0.17)		-0.24* (0.13)		-0.11 (0.10)	
	-1.48^{***} (0.41)		-1.12^{***} (0.32)		-1.11^{***} (0.22)		-0.73^{***} (0.17)		-0.52^{***} (0.16)
	-0.10 (0.29)		$0.03 \\ (0.25)$		-0.10 (0.26)		-0.16 (0.23)		-0.13 (0.18)
	-0.32 (0.33)		-0.06 (0.37)		-0.17 (0.28)		-0.12 (0.22)		-0.07 (0.17)
	-0.53^{*} (0.30)		-0.44 (0.28)		-0.36^{*} (0.22)		-0.25 (0.17)		-0.20 (0.16)
	-0.48 (0.37)		-0.48 (0.32)		-0.31 (0.22)		-0.18 (0.17)		-0.05 (0.15)
	-0.86^{**} (0.36)		-0.62^{**} (0.30)		-0.52^{*} (0.27)		-0.31 (0.21)		-0.04 (0.17)
	-0.75^{***} (0.28)		-0.49^{*} (0.25)		-0.14 (0.22)		$0.08 \\ (0.19)$		0.25 (0.16)
1260 48	1260	672	672 48	938 67	938 67	1554	1554	2674	$2674 \\ 191$
Yes No	Yes No	No Yes	No Yes	No Yes	No Yes	No Yes	No Yes	No Yes	No Yes Yes
	48 Yes	-0.10 (0.29) -0.32 (0.33) -0.53* (0.30) -0.48 (0.37) -0.86** (0.36) -0.75*** (0.36) -0.75*** (0.28) 1260 1260 48 48 48 Yes Yes No No	-0.10 (0.29) -0.32 (0.33) -0.53* (0.30) -0.48 (0.37) -0.86** (0.36) -0.75*** (0.28) 1260 1260 48 48 48 48 48 48 48 Yes No No No Yes	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$				

Table C.20: Differences-in-difference estimates with population dynamics

Notes: The dependent variable is the annualized municipality population growth (in percentage points) from 1869 to 2011 at the level of municipalities across the line of contact. Columns (1) and (2) show pairwise difference-in-differences estimates with year and pair fixed effects. Columns (3) to (10) use year and municipality fixed effects and extend the sample from contiguous line-of-contact municipalities to larger regional samples of municipalities within 5, 10 and 20 kilometers of the nearest line-of-contact municipality. Standard errors in parentheses are corrected for spatial dependence. Significance levels: *** 0.01, ** 0.05, * 0.10.

	Dependent variable: Measures of natural population growth (in %, per year)											
	1951	1961^{a}	1971	1981	1991	2001	2011	Pooled $(1951-2011)$				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)				
Panel A: I	Natural popu	lation growth	(Panel B - F	anel C) per	capita							
Red Army	-0.18 (0.14)	-0.41^{***} (0.14)	0.17 (0.26)	$0.12 \\ (0.11)$	-0.51^{***} (0.13)	-0.28^{**} (0.12)	-0.22 (0.15)	-0.19^{***} (0.06)				
No. of obs. Pair FE R^2 adj.	90 Yes 0.58	90 Yes 0.62	90 Yes 0.52	90 Yes 0.69	90 Yes 0.58	90 Yes 0.52	90 Yes 0.52	630 Yes 0.48				
Panel B: I	Births per ca	pita										
Red Army	-0.13 (0.17)	-0.14 (0.10)	-0.03 (0.11)	-0.00 (0.07)	-0.04 (0.09)	-0.12^{**} (0.06)	0.17^{**} (0.08)	-0.04 (0.03)				
No. of obs. Pair FE R ² adj.	90 Yes 0.54	90 Yes 0.70	90 Yes 0.42	90 Yes 0.75	90 Yes 0.48	90 Yes 0.59	90 Yes 0.58	630 Yes 0.69				
Panel C: I	Deaths per c	apita										
Red Army	$0.04 \\ (0.11)$	0.27^{**} (0.11)	-0.19 (0.26)	-0.13 (0.08)	0.46^{***} (0.09)	$0.16 \\ (0.10)$	0.38^{***} (0.12)	0.14^{***} (0.05)				
No. of obs. Pair FE R ² adj.	90 Yes 0.40	90 Yes 0.47	90 Yes 0.47	90 Yes 0.53	90 Yes 0.64	90 Yes 0.53	90 Yes 0.56	630 Yes 0.16				

Table C.21: Births and deaths by year of census

Notes: The table tests for spatial discontinuities across contiguous line-of-contact municipalities in measures of natural population growth (as percentage points) based on pairwise regressions. Panels B and C test for differences in births per capita and deaths per capita, respectively, while Panel A shows the combined view (births minus deaths). Columns (1) to (7) show estimates by census year, column (8) shows the pooled estimates based on all decennial census since WWII. Pairwise estimates include pair fixed effects. a.) Data for 1961 are based on the birth and death census in 1960. Standard errors in parentheses are clustered at the contiguous pair level. Significance levels: *** 0.01, ** 0.05, * 0.10.

	Dependent variable: Municipality covariates														
	Employees per sector (in %)			Workplaces	Em	Employees per production units			Type of occupation (in %)		Municipal residents' characteristics (in $\%$)				
	Agriculture In (1)	Industry Services (2) (3)	Services	Public (4)	per resident (5)	All firms (6)	Farms (7)	Industrial firms (8)	Service firms (9)	Blue collar (10)	Self-em- ployed (11)	Compul- sory educ. (12)	Tertiary educ. (13)	Unem- ployed (14)	Out-com- muters (15)
			(3)												
Panel A: Pair	wise regression	n													
Red Army	7.68^{**} (3.35)	$1.26 \\ (2.88)$	-8.94^{**} (3.76)	$0.41 \\ (1.41)$	-0.01 (0.03)	$0.44 \\ (0.43)$	-0.04 (0.06)	3.17^{*} (1.67)	-0.27 (0.43)	-2.84 (2.33)	$3.82 \\ (3.74)$	1.96^{***} (0.74)	-0.31 (0.36)	-0.59 (0.38)	-3.32^{*} (1.76)
No. of obs. No. of munc Pair FE	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes	90 48 Yes
Panel B: Sing	le-dimensional	RDD								-					
Red Army	4.81 (4.84)	1.50 (4.43)	-6.31 (5.67)	$0.01 \\ (1.78)$	$0.01 \\ (0.05)$	$0.75 \\ (0.65)$	$0.00 \\ (0.10)$	2.35 (2.46)	$0.20 \\ (0.71)$	-0.72 (4.24)	$0.06 \\ (6.05)$	1.83^{*} (1.01)	-0.55 (0.54)	-0.15 (0.43)	-3.28 (2.13)
No. of obs. Seg. & Geo FE	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes
Panel C: Mult	ti-dimensional	RDD													
Red Army	$5.61 \\ (4.78)$	$0.26 \\ (4.49)$	-5.87 (5.95)	-0.17 (1.44)	$0.01 \\ (0.05)$	0.41 (0.62)	-0.02 (0.10)	1.97 (2.48)	-0.14 (0.63)	-1.09 (4.29)	$3.30 \\ (5.88)$	$0.80 \\ (1.01)$	$0.13 \\ (0.51)$	-0.22 (0.42)	-4.52^{**} (1.81)
No. of obs. Seg. & Geo FE	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes	191 Yes

Table C.22: Municipal covariates in 2011

Notes: The table tests for spatial discontinuities across line-of-contact municipalities in various covariates for the year 2011. Columns (1) to (4) divide the number of local workplaces per sector by the overall number of workplaces in the municipality. Column (5) divides the number of local workplaces by local population. Columns (6) to (9) divide the number of sector-specific local workplaces by the number of local firms per sector. Columns (10) and (11) divide the number of the respective local type of workplace by the total number of local workplaces. Columns (12) to (15) divide the respective characteristics of local residents by the overall municipality population. Panel A consists of contiguous line-of-contact municipalities and includes pair fixed effects. Panels B and C consist of municipalities within 20 kilometers of the nearest line-of-contact municipality (RD sample) and include segment and geography fixed effects. RD estimates employ a cubic polynomial fit. Standard errors in parentheses in Panel A are clustered at the contiguous pair level and corrected for spatial dependence in Panels B and C. Significance levels: *** 0.01, ** 0.05, * 0.10.

	Dependent variable: Machinery per 1,000 workers in agriculture									
	Propensity score matching									
	OLS di	fferences	Match over: 1939	Farm structure in	Match over: Farm structure and Occupation shares in 1939					
	1953	1962	1953	1962	1953	1962				
Machinery x Red Army:	(1)	(2)	(3)	(4)	(5)	(6)				
Tractors all	-9.98**	-34.70^{**}	-9.13**	-40.98^{**}	-7.12^{*}	-31.11^{**}				
	(3.99)	(15.07)	(4.33)	(16.08)	(3.56)	(14.68)				
Tractor weak (below 16PS)	-4.11^{*}	-8.82	-3.66^{*}	-16.48	-1.58	-8.46				
	(2.04)	(9.73)	(2.06)	(11.47)	(1.50)	(11.20)				
Tractor strong (above 16PS)	-5.87^{**}	-25.88^{***}	-5.47^{**}	-24.50***	-5.55^{**}	-22.65^{***}				
	(2.34)	(7.42)	(2.64)	(7.51)	(2.19)	(6.94)				
Engine all (electro, gasoline, diesel)	-26.19	-57.84	-39.16*	-84.03*	-22.20	-57.85				
	(19.03)	(34.84)	(21.78)	(45.69)	(19.65)	(53.62)				
Stationary diesel and gasoline engines	-33.56^{***}	-30.43***	-32.60**	-22.83**	-34.04***	-19.84^{*}				
	(8.38)	(10.23)	(12.01)	(10.99)	(11.93)	(10.31)				
Electro engine all	7.37 (20.71)	-27.41 (37.24)	-6.56 (20.16)	-61.20 (44.30)		-38.01 (49.89)				
Trailer all (for tractors and horses)	-14.43^{**}	-41.41*	-15.87**	-60.46^{**}	-18.17**	-54.77*				
	(5.98)	(22.19)	(6.82)	(24.80)	(7.96)	(28.26)				
Trailers for tractors	-11.21** (4.15)		-10.27** (4.45)		-10.65** (4.81)					
Trailers for horses	-3.23 (2.53)		-5.60 (3.62)		-7.52 (5.23)					
Crop machines all $(w/o harvesters)^a$	-42.11^{***}	-120.37***	-44.24^{***}	-143.07***	-40.23***	-140.84***				
	(13.28)	(32.46)	(13.37)	(34.26)	(11.13)	(38.68)				
Tillage machine (for tractors and horses)	-4.82^{*}	-11.54**	-6.12^{**}	-14.58***	-4.33*	-11.84^{**}				
	(2.60)	(4.85)	(2.58)	(4.57)	(2.51)	(5.18)				
Tillage for tractors	-0.48	-7.85^{***}	-0.81^{*}	-8.95***	-0.82*	-9.03***				
	(0.44)	(2.78)	(0.41)	(2.71)	(0.48)	(3.18)				
Tillage for horses	-4.33^{*}	-3.69	-5.31^{**}	-5.64^{*}	-3.51	-2.81				
	(2.29)	(2.97)	(2.33)	(2.99)	(2.33)	(3.20)				
Harvesters (crops and potatos) ^{b}	-18.96^{***} (5.93)	-54.13*** (12.82)	-23.26^{***} (5.89)	-68.74^{***} (13.13)	-23.44^{***} (4.14)	-66.55^{***} (11.97)				
Combined crop harvester	-3.66^{**}	-13.41***	-3.27	-14.48^{**}	-2.80**	-13.35**				
	(1.60)	(4.87)	(1.99)	(5.42)	(1.13)	(4.82)				
Potato harvester	-15.30^{***} (4.79)	-40.72^{***} (9.42)	-19.99^{***} (4.38)	-54.25^{***} (9.51)	-20.64^{***} (3.68)	-53.20^{***} (8.97)				
Thresing machine all	-52.19***	-65.48***	-63.03***	-75.00**	-64.62***	-80.47**				
	(11.32)	(20.41)	(16.29)	(29.51)	(17.23)	(33.74)				
Thresing machine incl. cleaning	-8.58^{*}	-17.68^{*}	-5.49	-11.85	-7.30	-13.87				
	(4.68)	(10.20)	(5.09)	(12.31)	(4.74)	(13.09)				
Thresing machine without cleaning	-43.62^{***} (10.23)	(10.20) -47.80*** (15.10)	-57.54*** (13.90)	-63.15^{***} (21.63)	(1117) -57.31*** (15.17)	-66.59^{**} (24.04)				
Haymachines ^c	-33.40^{*}	-114.90***	-37.92*	-142.40^{***}	-29.04	-118.81***				
	(17.31)	(31.70)	(19.92)	(30.74)	(23.77)	(37.29)				
Milking machine (electirc)	-2.97^{**}	-24.70^{*}	-3.36***	-26.85^{*}	-2.84^{**}	-20.77				
	(1.12)	(12.31)	(1.19)	(14.39)	(1.22)	(14.37)				
No of units	38	38	34	34	27	27				

Table C.23: Machinery in agriculture (contd.)

Notes: The table shows differences in agriculture machinery per 1,000 employees in agriculture among Red-Army- and Western-Allies-liberated areas in South Austria. The sample consists of court districts along the within-Styrian and Styrian-Corinthian line of contact (court districts along the line of contact and the respective neighboring court districts). Census data from 1939, 1951 and 1961 are merged to the agricultural census in 1953 and 1962. Matching variables for Columns (3) and (4) include average agricultural residents per farm in 1939, matching variables for Columns (5) and (6) include average agricultural residents per farm in 1939, share of agriculture and share of industry in 1939. a) Crop cultivation machines include: Plows, harrows, sugar beet hoes, tillage machines (for horses and tractors), network harrows, sewing machines and fertilizer spreaders. b) Harvesters include: Sheaf-binding harvesters, combine harvesters and potato harvesters. c) Hay machines include: Hay tedders, hay rakes (various types), grass and hay loaders. Significance levels (robust standard errors): *** 0.01, ** 0.05, * 0.10.

Abstrakt

Způsobuje krátká epizoda konfliktu nebo vystavení nepřátelským jednotkám regionální ekonomickou zaostalost, a pokud ano, proč a jak přetrvává? Na tyto otázky odpovídám využíváním ekonomických rozdílů napříč idiosynkratickou a krátkodobou linií kontaktů mezi Rudou armádou a západními spojenci v jižním Rakousku na konci druhé světové války. Regresní odhady využívající prostorové diskontinuity ukazují, že nepřátelská přítomnost Rudé armády po dobu 74 dnů způsobila okamžitý relativní pokles populace o přibližně 12 %, dnes zesílený na 25 %. Věkově specifické migrační trendy a následné rozdíly v plodnosti vysvětlují multiplikační efekty. Rozvoj odvětví a měřítka místní produktivity práce v roce 2011 také zaostávají v regionech, kterých se nakrátko zmocnila Rudá armáda, což je pravděpodobně způsobeno migrací odborné pracovní síly a zbržděnými investičními trendy po druhé světové válce. Zjištění poskytují nový pohled na dlouhodobé dopady válek a konfliktů a poukazují na izolovanou roli nepřátelských akcí Rudé armády po druhé světové válce k pochopení evropského hospodářského rozdělení mezi Východem a Západem.

Working Paper Series ISSN 2788-0443

Individual researchers, as well as the on-line version of the CERGE-EI Working Papers (including their dissemination) were supported from institutional support RVO 67985998 from Economics Institute of the CAS, v. v. i.

Specific research support and/or other grants the researchers/publications benefited from are acknowledged at the beginning of the Paper.

(c) Christian Ochsner, 2023

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means, electronic, mechanical or photocopying, recording, or otherwise without the prior permission of the publisher.

Published by Charles University, Center for Economic Research and Graduate Education (CERGE) and Economics Institute of the CAS, v. v. i. (EI) CERGE-EI, Politických vězňů 7, 111 21 Prague 1, tel.: +420 224 005 153, Czech Republic. Phone: + 420 224 005 153 Email: office@cerge-ei.cz Web: https://www.cerge-ei.cz/

Editor: Byeongju Jeong

The paper is available online at https://www.cerge-ei.cz/working-papers/.

ISBN 978-80-7343-575-2 (Univerzita Karlova, Centrum pro ekonomický výzkum a doktorské studium) ISBN 978-80-7344-701-4 (Národohospodářský ústav AV ČR, v. v. i.)